

Examining the Challenges in Mainstreaming Climate Change Adaptation into Local Land-use Planning: The Case of Albay, Philippines

Sining C. Cuevas Master of Arts in Economics Bachelor of Science in Economics

A thesis submitted for the degree of Doctor of Philosophy at The University of Queensland in 2015 School of Geography, Planning, and Environmental Management

Abstract

Mainstreaming climate change adaptation (CCA) is an approach that links adaptation and sustainable development goals by integrating climate change information, concerns, and considerations into existing development planning, and policy- and decision-making processes. However, a gap exists in the operationalisation of mainstreaming, mainly because the tools and methodologies in mainstreaming neglect the institutional reforms needed in the approach. This thesis focused on mainstreaming CCA into local land-use planning, and asked "How can mainstreaming of CCA into local land-use planning be understood?" and "How can the challenges in the operationalisation of mainstreaming be overcome?" To answer these questions, a four-stage mixed methodology was devised and successfully applied in examining the challenges in mainstreaming CCA into local land-use planning in Albay, Philippines. Local land-use planning in Albay is a "critical case" because it presents evidence of institutional capacity for long-term adaptation to climate change, with indication of transformational opportunities for mainstreaming. The methodology applied the mixed method, case study, and scorecard approaches, and it involved triangulation by data technique (i.e., document review, interview, survey, and key informant consultations). By using this methodology, the research generated 20 quantitative "mainstreaming indicators" and produced qualitative assessments of the state-of-play and the challenges in local mainstreaming of CCA. Analyses revealed that (1) mainstreaming challenges exist within a spectrum, with barriers and opportunities for adaptation representing the extreme ends of this spectrum; (2) barriers can be overcome and can transcend into opportunities for mainstreaming CCA; (3) barriers can be classified according to varying levels of severity; and (4) barriers themselves are interconnected, but to differing degrees. Also, the research showed that mainstreaming operationalisation involves a network of interacting institutions and institutional arrangements that transcend across governance scales. Likewise, the challenges in mainstreaming CCA encompass a chain of interactions or interplays within the network (of institutions). Accordingly, overcoming mainstreaming challenges necessitates broad institutional reforms that go beyond the institutional setting where CCA is to be integrated. A deep understanding of these concerns can help scholars, practitioners, planners, and decision-makers anticipate the types of challenges to be encountered during the mainstreaming process; determine the severity of the impacts of these challenges; and formulate strategies that will overcome the challenges.

Declaration by author

This thesis is composed of my original work, and contains no material previously published or written by another person except where due reference has been made in the text. I have clearly stated the contribution by others to jointly-authored works that I have included in my thesis.

I have clearly stated the contribution of others to my thesis as a whole, including statistical assistance, survey design, data analysis, significant technical procedures, professional editorial advice, and any other original research work used or reported in my thesis. The content of my thesis is the result of work I have carried out since the commencement of my research higher degree candidature and does not include a substantial part of the work that has been submitted to qualify for the award of any other degree or diploma in any university or other tertiary institution. I have clearly stated which parts of my thesis, if any, have been submitted to qualify for another award.

I acknowledge that an electronic copy of my thesis must be lodged with the University Library and, subject to the policy and procedures of The University of Queensland, the thesis be made available for research and study in accordance with the Copyright Act 1968 unless a period of embargo has been approved by the Dean of the Graduate School.

I acknowledge that copyright of all material contained in my thesis resides with the copyright holder(s) of that material. Where appropriate I have obtained copyright permission from the copyright holder to reproduce material in this thesis.

Publications during candidature

Peer-reviewed papers:

Cuevas, S.C. (Accepted). The Interconnected Nature of Challenges in Mainstreaming Climate Change Adaptation: Evidence from Local Land-use Planning. *Climatic Change*.

Cuevas, S.C., Peterson, A., Morrison, T.H., & Robinson, C. (In press). Methodology for Examining the Challenges in Mainstreaming Climate Change Adaptation. *International Journal of Climate Change Strategies and Management*.

Cuevas, S.C., Peterson A., Robinson, C. & Morrison, T.H. (2015). Institutional Capacity for Long-Term Climate Change Adaptation: Evidence from Land-use Planning in Albay, Philippines. *Regional Environmental Change*. DOI 10.1007/s10113-015-0909-8

Cuevas, S.C., Peterson, A., Robinson, C. & Morrison, T. H. (2015). Challenges in Mainstreaming Climate Change Adaptation into Local Land-use Planning: Evidence from Albay, Philippines. *The International Journal of Climate Change: Impacts and Responses*, 7(3):45–65.

Cuevas, S.C. (2012). Examining Climate Change Adaptation Measures: An Early Warning System in the Philippines. *International Journal of Climate Change Strategies and Management*, 4(4), 358-385.

Book chapter:

Cuevas, S.C., Peterson, A., & Morrison, T. (2014). An Analytical Framework for Investigating Complex Institutions in Climate Change Adaptation: The Institutional Environment Matrix. In W. Leal Filho (Ed.), *Handbook of Climate Change Adaptation* (pp. 1-22). Berlin, Germany: Springer Berlin Heidelberg.

Conference Proceedings:

Cuevas, S.C. (2015). Linkages among the Challenges in Mainstreaming Climate Change Adaptation into Local Land-Use Planning. In (Eds.), *The Sustainable City X* (pp. 263-274). Boston: WITPress.

Conference abstracts:

Cuevas, S.C., Peterson, A., Morrison, T., & Robinson, C. (2015). Challenges in Mainstreaming Climate Change Adaptation into Local Land Use Planning: Evidence from Albay, Philippines. Conference on Climate Change: Impacts and Responses, 10-11 April 2015, Vancouver, Canada.

Cuevas, S.C., Peterson, A., Morrison, T., & Robinson, C. (2014). Challenges in Mainstreaming Climate Change Adaptation: The Case of Albay, Philippines. Climate Adaptation Conference, 30 September – 2 October 2014, Gold Coast, Queensland, Australia.

Publications included in this thesis

Cuevas, S.C., Peterson, A., Morrison, T.H., & Robinson, C. (In press). Methodology for Examining the Challenges in Mainstreaming Climate Change Adaptation. *International Journal of Climate Change Strategies and Management* – incorporated as Chapter 3

Contributor	Statement of contribution
Sining C. Cuevas (Candidate)	Analysis of data (100%)
	Wrote the paper (100%)
	Editing (70%)
Ann Peterson	Commented on and edited the paper (15%)
Tiffany Morrison	Commented on and edited the paper (8%)
Catherine Robinson	Commented on and edited the paper (7%)

Cuevas, S.C. (Accepted). The Interconnected Nature of Challenges in Mainstreaming Climate Change Adaptation: Evidence from Local Land-use Planning. *Climatic Change* – substantial discussion included in Chapter 4.

Contributor	Statement of contribution
Sining C. Cuevas (Candidate)	Analysis of data (100%)
	Wrote the paper (100%)

Cuevas, S.C., Peterson, A., Robinson, C. & Morrison, T.H. (2015). Challenges in Mainstreaming Climate Change Adaptation into Local Land-use Planning: Evidence from Albay, Philippines. *The International Journal of Climate Change: Impacts and Responses*, 7(3):45–65. – substantial discussion is included in Chapter 5.

Contributor	Statement of contribution
Sining C. Cuevas (Candidate)	Analysis of data (100%)
	Wrote the paper (100%)
	Editing (70%)
Ann Peterson	Commented on and edited the paper (16%)
Catherine Robinson	Commented on and edited the paper (7%)
Tiffany Morrison	Commented on and edited the paper (7%)

Cuevas, S.C., Peterson, A., Robinson, C. & Morrison, T.H. (2015). Institutional Capacity for Long-Term Climate Change Adaptation: Evidence from Albay, Philippines. *Regional Environmental Change* – incorporated as Chapter 6.

Contributor	Statement of contribution
Sining C. Cuevas (Candidate)	Analysis of data (100%)
	Wrote the paper (100%)
	Editing (70%)
Ann Peterson	Commented on and edited the paper (10%)
Catherine Robinson	Commented on and edited the paper (10%)
Tiffany Morrison	Commented on and edited the paper (10%)

Cuevas, S., Peterson, A., & Morrison, T. (2014). An Analytical Framework for Investigating Complex Institutions in Climate Change Adaptation: The Institutional Environment Matrix. In W. Leal Filho (Ed.), *Handbook of Climate Change Adaptation* (pp. 1-22). Berlin, Germany: Springer Berlin Heidelberg. – incorporated as Appendix A4.

Contributor	Statement of contribution
Sining C. Cuevas (Candidate)	Analysis of data (100%)
	Wrote the paper (100%)
	Editing (80%)
Ann Peterson	Commented on and edited the paper (10%)
Tiffany Morrison	Commented on and edited the paper (10%)

Contributions by others to the thesis

This thesis benefitted from significant inputs from Dr. Ann Peterson, Dr. Tiffany Morrison, and Dr. Catherine Robinson, who helped me set the foundation, concept, and design of the research.

Statement of parts of the thesis submitted to qualify for the award of another degree

None.

Acknowledgements

I wrote this thesis as a full-time student at the School of Geography, Planning, and Environmental Management, from 2012 to 2015, with financial support from The University of Queensland through the International Postgraduate Research Scholarship and Centennial Awards. I am also very grateful to the Commonwealth Scientific and Industrial Research Organisation and the University of Queensland Integrated Natural Resource Management PhD Program (CSIRO-UQ INRM PhD Program) for financial support in conducting the research activities.

I am eternally grateful to my supervisors, Dr. Ann Peterson, Dr. Tiffany Morrison, and Dr. Catherine Robinson, for sharing their knowledge and expertise in completing this research, and in providing comments to improve this thesis.

I also would like to thank all the people who participated in this research—survey respondents, interviewees, and key informants-especially the representatives from the Provincial Government of Albay, Albay Provincial Planning and Development Office, Albay Public Safety and Emergency Management Office, the Albay Provincial Agriculturist, Department of Interior and Local Government (Region V and Albay Provincial Office), Department of Environment and Natural Resources Region V, Department of Agrarian Reform (Region V and Legazpi City Office), Environmental Management Bureau Region V, Mines and Geosciences Bureau Region V, Housing and Land Use Regulatory Board (Central Office and Region V), City Government of Legazpi, City Planning Development Office (Legazpi), City Disaster Risk Reduction Management Council (Legazpi), Municipal Government of Camalig, Municipal Planning Development Office (Camalig), Climate Change Commission, National Economic and Development Authority (Central Office and Region IV), United Nations Human Settlement Programme (Philippines), University of the Philippines Los Banos, University of the Philippines Diliman, Philippine Institute of Environmental Planners, Department of Budget and Management, and the Civil Service Commission.

I would like to acknowledge the valuable assistance of Dr. Arturo Martinez, Jr., my go to person in statistical matters.

Most especially, I thank my family who continuously encouraged me throughout this journey; at last the thesis is here. Finally, I thank God for His blessings over the years.

Keywords

climate change adaptation, mainstreaming challenges, mainstreaming indicators, institutional capacity, resource capacity, information capacity, modified IAD, mixed methodology, barriers, opportunities

Australian and New Zealand Standard Research Classifications (ANZSRC)

ANZSRC code: 120504, Land-use and Environmental Planning, 30% ANZSRC code: 120599, Urban and Regional Planning not elsewhere classified, 30% ANZSRC code: 160510, Public Policy, 40%

Fields of Research (FoR) Classification

FoR code: 1205, Urban and Regional Planning, 60% FoR code: 1605, Policy and Administration, 40%

Table of Contents

Abstract	ii
Acknowledgements	ix
List of Figures	XV
List of Tables	xvi
List of Abbreviations used in the thesis	xvii
CHAPTER 1: INTRODUCTION	1
1.1 Background	1
1.2 What is mainstreaming climate change adaptation?	2
1.2.1 Mainstreaming climate change adaptation into local land-use planning	4
1.2.2 Issues in mainstreaming climate change adaptation	5
1.3 Problem statement	8
1.4 Research questions and objectives	8
1.5 Structure of thesis	10
CHAPTER 2: MAINSTREAMING CLIMATE CHANGE ADAPTATION, A	
REVIEW OF THEORY AND APPLICATION	12
2.1 Introduction	12
2.2 Mainstreaming climate change adaptation – the theory	13
2.2.1 Definitions and concepts	13
2.2.2 Mainstreaming climate change adaptation and institutional linkages	15
2.2.3 Sectoral mainstreaming: land-use planning	
2.4 Developments in mainstreaming climate change adaptation	21
2.4.1 Mainstreaming in policy-making	21
2.4.2 Mainstreaming in planning	24
2.4.3 Operationalising mainstreaming of climate change adaptation	26
2.6 Summary and conclusions	29
CHAPTER 3: METHODOLOGY FOR EXAMINING THE CHALLENGES IN	
MAINSTREAMING CLIMATE CHANGE ADAPTATION	31
3.1 Introduction	31
3.2 Mixed methodology in climate change adaptation research	
3.3 Analytical framework for examining the challenges in mainstreaming climate	
change adaptation	

3.3.1 The IAD-CCA framework	40
3.3.2 Quantitative aspect of the mixed methodology	42
3.4 The case study approach	46
3.4.1 Case study area selection	47
3.4.2 Selection of survey and in-depth interview respondents	54
3.4.3 Survey, interview, and consultation set-up	57
3.5 Data collection scheme	58
3.6 Methodology issues and limitations	59
3.7 Summary	61
3.8 Conclusion	61
CHAPTER 4: THE INTERCONNECTED NATURE OF THE CHALLENGES	
IN MAINSTREAMING CLIMATE CHANGE ADAPTATION: EVIDENCE	
FROM LOCAL LAND-USE PLANNING	63
4.1 Introduction	63
4.2 Identifying the challenges in climate change adaptation	64
4.2.1 Institutional challenges	65
4.2.2 Information challenges	68
4.2.3 Resource challenges	70
4.2.4 Linkages among the challenges in mainstreaming	71
4.3 Nature of the challenges in mainstreaming climate change adaptation into local	
land-use planning	72
4.3.1 Interconnections among mainstreaming challenges	73
4.3.2 Tripartite relationship among leadership, local government prioritisation, and commit	mont
to CCA	1110111 77
4.2.2 Palationship batwaan knowladge and awaraness and institutional incentive	
4.5.5 Kelauonsinp between knowledge and awareness and institutional incentive	
4.4 Summary and conclusions	81
CHAPTER 5: CHALLENGES IN MAINSTREAMING CLIMATE CHANGE	
ADAPTATION INTO LOCAL LAND-USE PLANNING: EVIDENCE FROM	
ALBAY, PHILIPPINES	83
5.1 Introduction	83
5.2 Generating the mainstreaming indicators	84
5.3 Challenges mainstreaming CCA-DRR into land-use planning	85
5.3.1 Survey results	85

5.3.2 Nature of mainstreaming challenges	
5.4 Summary and conclusions	94
CHAPTER 6: INSTITUTIONAL CAPACITY FOR LONG-TERM CLIMATE	
CHANGE ADAPTATION: EVIDENCE FROM LAND-USE PLANNING IN	
ALBAY, PHILIPPINES	96
6.1 Introduction	96
6.2 Significance of developing institutional capacity for mainstreaming CCA	97
6.2.1 Institutional issues	98
6.2.2 Leadership: Climate change champion	102
6.3 Summary and conclusion	106
CHAPTER 7: BARRIERS AND OPPORTUNITIES IN MAINSTREAMING	
ADAPTATION: A CRITICAL ASSESSMENT	108
7.1 Introduction	108
7.2 Institutional dimension and the barriers in mainstreaming climate change adaptation	109
7.3 Influence of institutional nestedness on the mainstreaming process	111
7.3.1 Primary barriers: Institutional issues	111
7.3.2 Second level barriers: Availability and access to information	112
7.3. 3 Opportunity: Credibility and reliability of information	115
7.4 Impacts of local initiatives on the barriers and opportunities for mainstreaming	
CCA-DRR into local land-use plans in Albay	116
7.4.1 Third level barrier: Organisational cohesion	116
7.4.2 Opportunity: Local leadership	116
7.4.3 Opportunity: Knowledge and awareness	118
7.4.4 Opportunity: Local government prioritisation	119
7.4.5 Opportunity: Community support and institutional incentive	120
7.5 Institutional settings and institutional networks in climate change adaptation	120
7.6 Summary and conclusion	123
CHAPTER 8: CONCLUSION, POLICY IMPLICATIONS, AND FUTURE	
RESEARCH POSSIBILITIES	126
8.1 Introduction	126
8.2 Gap between theory and application of mainstreaming climate change adaptation	127
8.3 Methodology for examining the challenges in mainstreaming climate change	
adaptation	130

8.4 Nature of challenges in mainstreaming climate change adaptation: Assessing	
linkages and state-of-play	
8.5 Institutional dimension of mainstreaming climate change adaptation	
8.6 Final conclusions	143
8.7 Future research possibilities	
REFERENCES	150
APPENDIX	179
A1: Table A Mainstreaming indicator scores by scale	
A2: Survey on the challenges of mainstreaming climate change adaptation into land-use	
planning	
A3: Final interview schedule as revised in the field	
A4: An Analytical Framework for Investigating Complex Institutions in Climate	
Change Adaptation: The Institutional Environment Matrix	

List of Figures

Figure 1 Visibility of "mainstreaming" in IPCC Assessment Reports (2001-2014)22
Figure 2 The research methodology
Figure 3 Basic components of the IAD
Figure 4 Modified Institutional Analysis and Development framework as applied in mainstreaming climate change adaptation: IAD-CCA framework
Figure 5 Philippines, with the Albay case study area encircled
Figure 6 Survey and interview respondents
Figure 7 Interconnections among the mainstreaming challenges, provincial data75
Figure 8 Interconnections among the mainstreaming challenges, city/municipal data76
Figure 9 Mainstreaming indicators linked to local leadership: Albay, Philippines
Figure 10 Mainstreaming indicator scores and levels, by scale
Figure 11 Radar chart of the indicators used to assess mainstreaming of CCA-DRR in Albay, Philippines: Barriers or opportunities?
Figure 12 Network of selected institutions, institutional settings, and institutional arrangements for mainstreaming climate change adaptation: Case of local land-use planning

			 	and provide since	 01 100001 100	P2	\$
in Albay,	, Philipp	ines	 		 	 1	21

List of Tables

Table 1 Examples of mainstreaming by sector
Table 2 Comparison of institutional frameworks
Table 3 Components of the IAD-CCA evaluation criteria 43
Table 4 Survey scorecard: Selected indicators for mainstreaming climate change adaptation into local land-use planning
Table 5 List of the top 10 most vulnerable or at risk countries to climate change
Table 6 Local scale area selection process 54
Table 7 Data collection scheme 1: Method, scale and coverage
Table 8 Data collection scheme 2: Requirements and methods 59
Table 9 Mainstreaming indicators with correlation \geq 0.70, provincial scale74
Table 10 Mainstreaming indicators with correlation \geq 0.70, city/municipal scale75
Table 11 Matrix of challenges in mainstreaming climate change adaptation and disaster riskreduction into the local land-use plans: Albay, Philippines
Table 12 Primary local laws, programs and activities in Albay related to climate change adaptation
Table 13 Primary barrier and substantial opportunity for mainstreaming climate changeadaptation into local land use plans in Albay, Philippines
Table 14 Selected local government unit mandated plans and their legal basis
Table 15 List of local policies in Albay that helped improve organisational cohesion117

List of Abbreviations used in the thesis

A2C2	Albay in Action on Climate Change
AR	Assessment Report
CCA	Climate Change Adaptation
CCA-DRR	Climate Change Adaptation-Disaster Risk Reduction
CIRCA	Center for Initiatives and Research for Climate Adaptation
CRISTAL	Community-Based Risk Screening Tool-Adaptation and Livelihoods
CVCA	Climate Vulnerability and Capacity Analysis
DENR	Department of Environment and Natural Resources
DRR	Disaster Risk Reduction
DRRM	Disaster Risk Reduction and Management
HLURB	Housing and Land-use Regulatory Board
IAD	Institutional Analysis and Development framework
IAD-CCA	Institutional Analysis and Development framework for Climate Change
	Adaptation
IPCC	Intergovernmental Panel on Climate Change
KRA	Key Result Areas
LDRRM	Local Disaster Risk Reduction and Management
LDRRMO	Local Disaster Risk Reduction and Management Officer
LGU	Local Government Unit
MoEF	Ministry of Environment and Forests
NAP	National Adaptation Plan
NAPA	National Adaptation Programs of Action
OECD	Organisation for Economic Co-operation and Development
PAGASA	Philippine Atmospheric, Geophysical, and Astronomical Services Administration
PGA	Provincial Government of Albay
PLUC	Provincial Land-Use Committee
RA	Republic Act
SEA	Strategic Environmental Assessment
SPREP	Secretariat of the Pacific Regional Environment Programme
UNDP	United Nations Development Programme
UNEP	United Nations Environment Programme
USAID	United States Agency for International Development

CHAPTER 1: INTRODUCTION

1.1 Background

Climate change exists and presents new and diverse risks. Although humanity is fundamentally adaptive, the increasing exposure and losses from the effects of climate change pose critical adaptation difficulties (Hedger et al. 2008; Preston et al. 2009). In response, the climate change debate has shifted from whether there is a *need* to adapt to *how* to adapt to the predicted impacts of climate change (Biesbroek et al. 2013; Noble et al. 2014). Decision-makers, practitioners, and scholars are now prioritising climate change adaptation (CCA) in their agendas, resulting in a rapid rise in the number of adaptation-related publications in the last decade (Inderberg & Eikeland 2009; Nicholson-Cole & O'Riordan 2009; IPCC 2014a, 2014b). Consequently, CCA is now an important research field (Gupta et al. 2010; Rodima-Taylor et al. 2012).

The Intergovernmental Panel on Climate Change (IPCC) Fourth Assessment Report (AR4) has been instrumental in shaping the direction of current adaptation research (IPCC 2007) from the biophysical to the social, economic and institutional aspects of adaptation (Dovers & Hezri 2010; Noble et al. 2014). This change has prompted researchers to focus on the linkages between CCA and sustainable development:

Sustainable development can reduce vulnerability to climate change, and climate change could impede the nations' abilities to achieve sustainable development pathways (IPCC 2007, p. 20).

Hence, development and adaptation are acknowledged as mutually dependent strategies, and efforts to streamline climate-related concerns into the development-planning and decision-making processes are emerging (Ayers et al. 2014; Noble et al. 2014). Along with this is the change in the focus of CCA planning from impact, vulnerability, or risk assessments to adjusting the direction in development planning (Olhoff & Schaer 2010; Schipper et al. 2010). This approach in adaptation is called "mainstreaming."

Mainstreaming CCA integrates climate change and adaptation concerns into a broader set of actions within the current development-planning, policy-making, and decision-making processes (OECD 2009; UNDP-UNEP 2011). It is a proactive, long-term adaptation planning measure that addresses the origins of vulnerability, deals with issues of adaptive capacity, and most importantly, is a primary prerequisite for sustainability (Agrawala 2006; Parry et al. 2007). As such, developing countries, which typically have low adaptive capacity and are the most vulnerable to the effects of climate change, are encouraged to apply CCA mainstreaming (Mertz et al. 2009; Ayers et al. 2014; IPCC 2014a). Although research interest in mainstreaming CCA is growing, there is limited literature on how to operationalise the approach, especially at the local scale (Huxtable & Yen 2009; Olhoff & Schaer 2010; Mimura et al. 2014). Consequently, information is lacking on how to transcend from planning to the implementation of mainstreaming, identify the challenges in operationalising the approach, and assess the progress and setbacks of mainstreaming efforts (Persson & Klein 2008; Measham et al. 2011; Ayers et al. 2014).

Institutional issues are often neglected in mainstreaming CCA. The tools and methodologies applied in mainstreaming are typically concerned with climate change-related issues (vulnerability assessments, climate risk screening, impact models, climate change scenario building, etc.), and seldom address the institutional changes created by the approaches (Olhoff & Schaer 2010; Schipper et al. 2010; Lebel et al. 2012; SPREP & UNDP 2013; Hamin & Gurran 2014). This oversight also is observed in the general adaptation literature. Although scholars now are realising the significance of institutions in CCA research (Adger et al. 2005; Inderberg & Eikeland 2009; Rodima-Taylor 2012), the institutional dimension of CCA remains the least understood aspect of the issue (Evans & Stevens 2009; Pradhan et al. 2012; Rodima-Taylor 2012; Rodima-Taylor et al. 2012).

This chapter explores these concerns and establishes the foundations of this research. First, it introduces the concept of mainstreaming CCA, particularly as it relates to land-use planning. Second, it explores existing literature in order to identify knowledge gaps in this field, and probes how these gaps relate to mainstreaming CCA into land-use planning. Third, these discussions are framed to identify the research problem, research questions, aims, and objectives. The chapter concludes by explaining the general structure of the thesis.

1.2 What is mainstreaming climate change adaptation?

In general, the term "mainstreaming" is defined as follows (1) to incorporate, bring, or place something into (Stevenson 2004); (2) to cause (someone or something) to be included in (Merriam-Webster [Online] n.d.); or (3) to integrate something into the mainstream (Collins Dictionaries 2012). The "mainstream" term, on the other hand, is the shared opinion, ideas, attitudes, or activities that are regarded by most people as normal or conventional (Stevenson 2004). As a concept, mainstreaming refers to "integrating an issue into existing (usually development) institutions and decision-making" (Ayers et al. 2014, p. 295). The mainstreaming concept is not new and has been used within the contexts of environment, gender issues, disaster risk reduction, poverty, and lately, in CCA (Klein et al 2005; Olhoff & Schaer 2010; Uittenbroek et al 2013).

As in other fields, mainstreaming is chiefly associated with the term "integrating"; thus, mainstreaming CCA is defined as integrating CCA issues, concerns, information, policies, and activities into all or any aspect of development-planning and decision-making processes and practices (Agrawala 2006; Ayers & Dodman 2010; Olhoff & Schaer 2010; Uittenbroek et al. 2013; Ayers et al. 2014). As the mainstreaming approach offers extensive benefits, it has gathered a growing number of advocates from international funding agencies such as the Asian Development Bank, the World Bank, United Nations Development Programme, and the likes. Similarly, it has emerged as a popular climate change response in developing countries such as Bangladesh, the Philippines, and Vietnam, among others (UNDP-UNEP 2011; UNDP 2012).

International funding organisations encourage developing countries to mainstream CCA as it is an adaptation approach facilitated through existing schemes. Hence, the strategy uses limited resources more efficiently, as compared to designing and creating new or separate institutions for managing CCA (Klein et al. 2005, 2007; Lebel et al. 2012). Moreover, the approach combines several policy goals (i.e., reduce risk and vulnerability to climate change, build resilience against climate change, and plan for sustainable development); thus, it stimulates policy coherence (Risbey et al. 2006; Rauken et al. 2015). Likewise, integrating adaptation concerns into development activities will help new policies, strategies, or plans to avoid worsening system vulnerability (i.e., maladaptation) (Olhoff & Schaer 2010). Mainstreaming promotes synergy in planning as it accounts for the following information (1) how climate will change in the future; (2) the uncertainty of the climate information; and (3) the vulnerability of systems to this uncertain future (O'brien et al. 2012). Essentially, mainstreaming adaptation minimises the trade-offs between climate change and sustainable development objectives, and maximises the opportunities that a harmonised planning and policy-making environment offers (Kok & de Coninck 2007; UNDP-UNEP 2011; Uittenbroek et al. 2013; Rauken et al. 2015).

Mainstreaming can exist at varying scales (i.e., international, regional, national, and local) and in different sectors (i.e., infrastructure, water, agriculture, poverty reduction, and education) (Kok & de Coninck 2007; OECD 2009; Tang et al. 2009; Dovers & Hezri 2010; UNDP-UNEP 2011). Whereas mainstreaming can be applied in various ways, mainstreaming CCA into land-use planning is paramount in achieving successful adaptation and sustainable development (Enemark 2012; Revi et al. 2014).

1.2.1 Mainstreaming climate change adaptation into local land-use planning

Local land-use planning often is referred to as the "constitution for future development" (Tang et al. 2009, p. 368) since it encompasses most of the locality's planning area, affects significant development concerns, reflects the community's development goals, and represents the future direction of public policies. Thus, integrating CCA into land-use planning is a key strategy to ensure sustainable development and efficient use of limited resources amid climate change conditions. The approach enhances the capability of communities to address the present and expected climate change risks, and to respond to and recover from climate change impacts (Klein et al. 2005, 2007; IPCC 2014a).

Likewise, mainstreaming CCA into land-use planning is an important task because of the complex relationship between climate change and land-use—climate change affects both the "demand and supply for space" (Koomen et al. 2008, p. 262). By altering the present and future use of that space, climate change influences the use, productivity, and access to land (Koomen et al. 2008; Robichaud & Wade 2011). Consequently, land issues and policies are vital concerns in adaptation planning (Quan & Dyer 2008).

Mainstreaming CCA into land-use planning can be through (1) expanding planning horizons to incorporate longer climate predictions; (2) strategising development in floodprone and other high-risk areas; (3) considering the medium- to long-term risks posed by climate change in vulnerable areas; (4) revising land-use regulations and standards that reflect climate variability; and (5) incorporating climate change risk assessments into land-information systems, among others (Govind 2011; UNDP-UNEP 2011; IPCC 2014a, 2014b). In the last decade, literature on mainstreaming CCA has included conceptual frameworks, guidelines, and handbooks on how to apply the approach. However, these documents only provide generic guidance for mainstreaming at the national, sectoral, programme and project levels. They do not offer detailed, operational instructions on how to implement the approach in practice (ADB 2005; Collins et al. 2005; Huxtable & Yen 2009; Olhoff & Schaer 2010). Essentially, "research is needed to establish the conditions under which the process of mainstreaming can be most effective" (Klein et al. 2005, p. 579).

Recently, several scholars have answered this call by investigating the mainstreaming process in specific cases. For example, researchers examined mainstreaming in development assistance in Mozambique (Sietz et al. 2011), urban planning in The Netherlands (Uittenbroek et al. 2013), municipal planning in South Africa (Pasquini et al. 2013), development planning in Bangladesh (Ayers et al. 2014), and environmental assessment in

Indonesia (Hamdani et al. 2014). Although these studies enrich the literature in mainstreaming CCA, significant knowledge gaps remain.

1.2.2 Issues in mainstreaming climate change adaptation

This section presents two of the important concerns in mainstreaming CCA that require immediate attention—the barriers that impede the effective operationalisation of the approach, and the methodology for investigating CCA mainstreaming. Although the general idea of mainstreaming is progressing, information is still lacking on how to (1) identify the challenges of mainstreaming; (2) transcend mainstreaming from planning to implementation; and (3) evaluate mainstreaming efforts, especially at the local scale (Huq & Ayer 2008; Persson & Klein 2008; Measham et al. 2011). Thus, there are knowledge gaps in terms of the practical application, operationalisation, monitoring, and assessment of mainstreaming initiatives (Mangoyana et al. 2012; Uittenbroek et al. 2013; Mimura et al. 2014).

1.2.2.1 Barriers in mainstreaming climate change adaptation

In practice, the operationalisation of mainstreaming has been slow and in some cases, it has not been implemented effectively (Uittenbroek et al. 2013; Revi et al. 2014; Lehmann et al. 2015). This is because information is lacking on how mainstreaming CCA can be applied on-ground, which consequently highlights a knowledge deficit regarding the mainstreaming process (Lal et al. 2012; Ayers et al. 2014; Picketts et al. 2014). This condition is aggravated by the lack of understanding about the barriers or challenges in the operationalisation of mainstreaming CCA at the local scale (Huq & Ayer 2008; OECD 2009; Mangoyana et al. 2012; Mimura et al. 2014).

Literature has referred to the barriers to adaptation in a number of ways, including limitations, constraints, obstacles, or challenges, that hinder, impede, restrain, limit, or delay planning, implementation, and the general advancement of CCA measures and approaches (Pervin et al. 2013; Ayers et al. 2014; Dang et al. 2014; Eisenack et al. 2014; Picketts et al. 2014; Waters et al. 2014). Early studies identified a wide array of barriers and grouped them under varying categories. For instance, Waters et al. (2014) identified 50 barriers spread across five classifications, namely governance, policy, psychosocial, resources, and information. Meanwhile, Dang et al. (2014) characterised the barriers to include socio-economic (e.g., lack of access to credit, lack of access to financial resources) and psychological (e.g., strength of belief to the existence of climate change, personal or community perception on climate risk) factors. Still, Ekstrom and Moser (2014) related the

barriers to institutional or governance issues; attitudes, values, and motivations of the actors involved; and leadership, among others. Clearly, these studies are answering the queries on (1) What are the barriers to adaptation? and (2) How are the barriers categorised?

However, on-ground mainstreaming conditions need answers to questions beyond identifying what the barriers to adaptation are. For example, some studies have shown that the refusal of local leaders to acknowledge climate change significantly hinders the ability of some local governments to plan for effective mainstreaming of CCA (Roberts 2008; Oberlack & Eisenack 2014). In the context of mainstreaming, this lack of leadership for CCA can be an effect of the lack of knowledge and awareness of climate change issues. In turn, this leadership dilemma can be the motivational impediment causing the community's unwillingness to act on climate change or the obstacle keeping the local government from prioritising CCA in the political agenda (Burch 2010; Gardner et al. 2010; Biesbroek et al. 2011; Oberlack & Eisenack 2014). In general, adaptation "barriers do not exist in isolation but are produced through inter-related processes" (Kuruppu & Willie 2015, p. 77). Interdependencies among the barriers to adaptation exist and since mainstreaming involves integrating development and adaptation processes and involves interactions among varying sectors, understanding these interdependencies (i.e., strength of associations and causal linkages) is crucial. Similarly, in-depth analysis of the circumstances surrounding the barriers is needed. These sets of information contribute to the assessment of how the barriers arise and continue to exist. Most importantly, they clarify how these barriers can be overcome and can be transformed into opportunities for adaptation (Burch 2010; Eisenack et al. 2014; Hamin et al. 2014; Klein et al. 2014). Essentially, more comprehensive knowledge concerning the nature of the barriers will help to more effectively mainstream CCA.

1.2.2.2 Methodology for assessing CCA mainstreaming

The ability of planners, policy-makers, and other decision-makers to measure and examine effectively the complex processes involved in mainstreaming CCA, including the barriers to operationalisation, is constrained by the lack of methodologies to investigate how mainstreaming is applied in practice (Tang et al. 2009; Ayers et al. 2014). At times, the operationalisation of mainstreaming is delayed or hindered because of limited or no institutional preparation. To illustrate, practitioners in urban planning in the Rockhampton region in Australia believed:

...in the ability of existing urban planning practices and principles to accommodate and respond to climate change, but indicated that reforms in the governance of spatial modelling (i.e., the centralised generation and provision of data such as climate model information, together with user support for local councils) and a handbook for integration within risk management frameworks were required for mainstreaming (Fry & Williams 2013, p. 1).

This situation demonstrates that mainstreaming CCA goes beyond climate change issues. This is so because "[i]n practice, the pathway to mainstreaming is not linear. It is made up of a patchwork of processes, stakeholders and approaches that converge or co-exist" (Ayers et al. 2014, p. 302).

At present, handbooks and guidelines for the operationalisation of mainstreaming lack this perspective. For example, Daze et al. (2009), OECD (2009), USAID (2009) UNEP-UNDP (2011) and SPREP and UNDP (2013) all advocate for "applying the climate lens" to the on-ground application of mainstreaming. A climate lens is applied as the "first step in national, sector, local and project policy and programme planning" (Frankel-Reed et al. 2011, p. 1). Applying the climate lens entails examining each stage of policy, planning, or strategy formulation from a climate-risk perspective to ensure that climate change and sustainable development goals will be attained (Lebel et al. 2012; SPREP & UNDP 2013). Specifically, applying a climate lens involves analysing the degree to which (1) the policy, strategy, or plan may be vulnerable to climate change; (2) the climate change risks have been accounted for in formulating the policy, strategy, or plan; and (3) the policy, strategy or plan can exacerbate the impacts of climate change, thus resulting in maladaptation (i.e., increased vulnerability to climate change) (OECD 2009). As the term "climate lens" suggests, this step is focused on climate change concerns. However, mainstreaming CCA is a system of institutional changes that needs to consider the institutional transformations that mainstreaming CCA will need or create—a perspective that the current mainstreaming processes lack.

In addition, while most planners are aware and do acknowledge the need to act on climate change, they are uncertain about how to proceed (Hamin et al. 2014). Given that onground implementation of mainstreaming is not straightforward (Ayers et al. 2014); planners need a systematic methodology for analysing the institutional settings where CCA will be integrated. This is an important point especially since CCA is an "abstract concept" (Persson & Klein 2008, p. 13) that needs to be translated into a language that planners can understand. A possible course of action is to develop indicators to help planners assess mainstreaming situations, thus providing a solid basis for action in planning (Oates 2011). Some scholars believe that metrics would help determine the state-of-play of the adaptation effort, evaluate and assess adaptation outcomes, and essentially measure adaptation progress and its effectiveness (Berrang-Ford et al. 2011; Mimura et al. 2014; Noble et al. 2014). Having mainstreaming indicators that can trace the "the extent to which targeted outcomes are occurring" would be most useful in both planning and policy-learning (Noble et al. 2014, p. 837).

Essentially, planning for the operationalisation of mainstreaming needs a methodology with an institutional perspective that can accommodate the non-linear nature of the approach. Thus, it calls for a methodology that can help planners to (1) investigate the existing institutional settings where CCA will be integrated; (2) monitor and assess the state-of-play of the mainstreaming process through quantitative measures; and (3) incorporate the barriers or challenges in mainstreaming CCA into the planning process.

1.3 Problem statement

Developing countries such as the Philippines are encouraged to mainstream CCA into their land-use plans as a long-term strategy for addressing the effects of climate change. Although interest in mainstreaming CCA is growing, planners and decision-makers have little understanding of the practical aspects of the approach. There are knowledge gaps in terms of the workable application, operationalisation, monitoring, and assessment of mainstreaming initiatives, especially at the local scale. Consequently, there is a lack of information on the challenges encountered in operationalising the mainstreaming endeavour and on how to overcome these challenges.

1.4 Research questions and objectives

Given the knowledge gaps outlined in the earlier discussion, this research addresses two questions:

- (1) How can mainstreaming climate change adaptation into local land-use planning be understood?
- (2) How can the challenges in the operationalisation of mainstreaming be overcome?

To answer these questions, this research analysed the challenges in integrating or "mainstreaming" CCA into local land-use planning in Albay, Philippines. It also determined how to identify, characterise, categorise, and assess these challenges in order to aid planners and decision-makers to overcome them effectively.

Accordingly, this research poses four objectives, namely:

Objective 1: To explore the process of mainstreaming CCA, from its theoretical foundations to its operationalisation, with special interest in local land-use planning.

This objective establishes (1) the need for research on mainstreaming CCA that specialises on local land-use planning; (2) why the research focuses on the challenges in operationalising the approach; and (3) what specific aspects of the mainstreaming challenges should be investigated intensively. This objective allows for a better understanding of the mainstreaming process, thereby setting the foundations for the significance of the research in the field of CCA.

Objective 2: To determine the analytical framework and methodology that can (1) examine effectively the challenges in mainstreaming CCA into land-use planning; and (2) generate metrics that can be used by planners and decision-makers in addressing these challenges.

This objective answers the call for a methodology and metrics for assessing the process of mainstreaming CCA, thereby indicating both scholarly and practical significance. This objective aims to improve the tool sets available for CCA planning, and decision- and policy-making.

Objective 3: To analyse the state-of-play of and linkages between the challenges in mainstreaming CCA into land-use planning in Albay, Philippines, and how to overcome these challenges.

This objective supports Objective 1 by providing empirical evidence to the theoretical aspects of mainstreaming CCA. Meanwhile, it also validates and verifies the effectiveness of the outputs of Objective 2. In effect, this objective integrates the various facets of the research. Most importantly, this objective addresses the primary knowledge gap in the field—the lack of practical application, operationalisation, monitoring, and assessment of mainstreaming initiatives—thus, it represents the core of the research.

Objective 4: To generate a more refined understanding of the operationalisation of mainstreaming in local CCA.

This objective is the synthesis of the first three objectives and determines the niche of the study within the realm of adaptation research, which is developing a mechanism to enable planners and policy-makers to operationalise local mainstreaming effectively.

1.5 Structure of thesis

The research thesis is divided into three major parts, namely, the review of literature, the methodology, and the results and discussion (Box 1). The literature review (Chapter 2) answers the first objective and focuses on the mainstreaming approach, its varying definitions, types, benefits, and issues. Chapter 3 addresses the second objective, and presents the modified Institutional Analysis and Development (IAD) framework as applied to CCA mainstreaming research (IAD-CCA framework) and the mixed-methodology devised to investigate the local mainstreaming process and to examine its challenges. The methodology was applied in a case study, specifically, the mainstreaming of CCA into the local land-use planning in Albay, Philippines. The results chapters (Chapters 4 to 6) present the research outputs in relation to Objective 3. Chapter 4 presents the quantitative analysis (i.e., correlation analysis) made possible through the survey data generated using the mixed methodology (Chapter 3); in particular, it presents evidence on the degree of interconnections among the mainstreaming challenges. Chapter 5 introduces the mainstreaming challenges (i.e., factors that affect the effective operationalisation of mainstreaming CCA) and the mainstreaming indicators (i.e., metrics for mainstreaming evaluation and assessment). Chapter 6 provides in-depth qualitative analysis of the indicators highlighted in Chapter 5 (i.e., institutional capacity indicators). It revolves around the theme that developing the institutional capacities of local governments is crucial in the local mainstreaming process. Thus, aside from addressing Objective 3, this chapter contributes also to attaining Objective 4. Likewise, Chapter 7 addresses Objective 4 by further explaining the institutional dimension of CCA, and explores how institutional nestedness (i.e., hierarchical relationships of institutions) and the active participation of local governments can transform a challenge into an opportunity for mainstreaming. Chapter 8 concludes the research by synthesising the results and by discussing the contribution of the research. Lastly, the data collection tools, such as the survey questionnaire and the interview schedule, as well as the related publication entitled "An Analytical Framework for Investigating Complex Institutions in Climate Change Adaptation: The Institutional Environment Matrix", are presented in the Appendix.

Box 1: Summary of dissertation chapters

CHAPTER 1

Introduction

CHAPTER 2

Mainstreaming Climate Change Adaptation, A Review of Theory and Application

Introduces and explains the concept of mainstreaming CCA, and explores the issues and concerns in mainstreaming CCA to determine the gap between mainstreaming theory and application.

CHAPTER 3

Methodology for Examining the Challenges in Mainstreaming Climate Change Adaptation

- Argues that mainstreaming CCA necessitates both climate and institutional perspectives, and that examining the challenges in mainstreaming is a critical part of the mainstreaming process.
- Develops a mixed-methodology to examine mainstreaming challenges. This methodology was successfully applied in a case study in Albay, Philippines.

CHAPTER 4

The Interconnected Nature of Challenges in Mainstreaming Climate Change Adaptation: Evidence from Local Land-Use Planning

Illustrates and measures the linkages between and among the mainstreaming challenges to understand better the relationships among the challenges, and thereby develop schemes to overcome these challenges.

CHAPTER 5

Challenges in Mainstreaming Climate Change Adaptation into Local Land-Use Planning: Evidence from Albay, Philippines

- Introduces the quantitative mainstreaming indicators used in the research that (1) measured the severity of the impacts of the challenges in mainstreaming CCA; and (2) explained the nature of the barriers and opportunities for mainstreaming CCA in Albay, Philippines.
- Presents the results/outputs that have the potential to help planners and decision-makers monitor the adaptation process and implementation, and also track the progress of adaptation efforts.

CHAPTER 6

Institutional Capacity for Long-Term Climate Change Adaptation: Evidence from Land-Use Planning in Albay, Philippines

Assesses the state-of-play of the local mainstreaming process, explores the institutional dimension of mainstreaming CCA, and advocates for strengthening the institutional capacities of systems for a long-term adaptation to climate change.

CHAPTER 7

Barriers and Opportunities in Mainstreaming Adaptation: A Critical Assessment

- Determines the institutional roots from which barriers to mainstreaming CCA arise, and how the barriers potentially can be transformed into opportunities for mainstreaming.
- Explains how institutional nestedness and a polycentric type of governance influence the existence and persistence of barriers to CCA.

CHAPTER 8

Conclusion, Policy Implications, and Future Research Possibilities

Synthesises the research findings and generates a more refined understanding of the operationalisation of mainstreaming CCA.

CHAPTER 2: MAINSTREAMING CLIMATE CHANGE ADAPTATION, A REVIEW OF THEORY AND APPLICATION

2.1 Introduction

In recent years, the strong linkages between development and the impacts of climate change have encouraged developing countries, which are considered most vulnerable to climate change, to apply an adaptation approach called "mainstreaming" (Ayers & Doman 2010; Field et al. 2014; IPCC 2014a). Mainstreaming climate change adaptation (CCA) is an approach that integrates climate change and adaptation concerns into a broader set of actions within the existing development planning, policy-making, and decision-making processes (OECD 2009; UNDP-UNEP 2011). For example, in the Philippines, climate change is integrated into the government processes and activities through the Climate Change Act of 2009. This law was supported by the Disaster Risk Reduction and Management Act of 2010, which decrees integrating not only climate change, but also disaster risk reduction (DRR) into the country's national and local government policy-making processes, socioeconomic development planning and sectoral governance (i.e., environment, agriculture, water, energy, health, education, land-use and urban planning, etc.) (Republic of the Philippines 2009, 2010).

Similarly, Bangladesh accounted for the likely impacts of climate change in its National Perspective Plan 2010–2021, the primary plan responsible for growth and poverty alleviation in the country, and in its National Agricultural Policy (Planning Commission 2010; Ayers et al. 2014). India mainstreamed climate change through the Ministry of Urban Development's National Mission for Sustainable Habitat Plan, which aimed to (1) re-orient urban planning to consider climate change; (2) develop capacity to respond to disaster; and (3) promote climate change awareness, among others (Sharma & Tomar 2010). Likewise, the Republic of Zambia incorporated climate change among the objectives and strategies listed in its Sixth National Development Plan, 2011–2015 (Republic of Zambia 2011). Ethiopia, on the other hand, formulated the Climate-Resilient Green Economy Strategy, which integrated both CCA and climate mitigation objectives into the five-year Growth and Transformation Plan for the country. This effort intended to "protect the country from the adverse effects of climate change and to build a green economy that will help realise its ambition of reaching middle income status before 2025" (Federal Democratic Republic of Ethiopia 2011). Clearly, mainstreaming CCA is advancing as an adaptation approach in several countries.

Amid the advocacy and interest in mainstreaming CCA, its practical on-ground application or operationalisation at the local scale remains very difficult to achieve (Huxtable & Yen 2009; Olhoff & Schaer 2010; Mimura et al. 2014). For example, the development of institutional frameworks for co-ordinating the array of adaptation initiatives involved in the approach is not keeping pace with the rapid progress in national and sub-national policies and strategies that initiate and promote the mainstreaming approach (Field et al. 2014). In theory, mainstreaming involves introducing new or modifying prevailing planning or decision-making processes, procedures, or practices (Lasco et al. 2009). The actual transition from mainstreaming planning to implementation lacks evidence and information (Olhoff & Schaer 2010; Schipper et al. 2010), and hence there is discord between mainstreaming theory and practice (Niang et al. 2014). As the key actors (i.e., planners and decision-makers) struggle to integrate adaptation into the existing development planning and policy domains (Uittenbroek et al. 2013; Noble et al. 2014), a key question persists: "What does mainstreaming look like in practice?" (Ayers et al. 2014, p. 293).

This chapter addresses the first objective, which is to explore the process of mainstreaming CCA, from its theoretical foundations to its operationalisation, with special interest in local land-use planning. In particular, this chapter analyses the mainstreaming process, and the gap between the theory and the operationalisation of mainstreaming. It has two main segments. The first section introduces mainstreaming of CCA; it defines and conceptualises the term, focuses on mainstreaming CCA as an institutional concern, and explains how mainstreaming CCA links to land-use planning. The second section focuses on the developments in the approach in relation to policy-making and planning, and its operationalisation.

2.2 Mainstreaming climate change adaptation – the theory

This section presents the theoretical aspects of mainstreaming CCA, including how it is defined and some examples of how the approach can be operationalised. It also expounds on the institutional dimension of mainstreaming and introduces a crucial aspect of adaptation and sustainable development planning—mainstreaming CCA into local land-use planning.

2.2.1 Definitions and concepts

Mainstreaming, as a concept, is not new and has been used in relation to education for handicapped children, gender issues, environment, disaster risk reduction, HIV/AIDS, and intercultural relations (Gupta & van der Grijp 2010; Olhoff & Schaer 2010). To illustrate, the

term mainstreaming became popular in the education field in the late 1970s, and was defined as integrating students with disabilities within general educational settings (Maheady et al. 2005; Sims & Voltz 2010). Gender mainstreaming, on the other hand, emerged in the mid-1990s following the Beijing Platform for Action during the United Nations (UN) Fourth World Conference on Women in 1995. Gender mainstreaming is the "process by which a gendered perspective (male and female) is integrated into the fabric of our communities, institutions, and lives" (Lyle-Gonga 2013, p. 209).

Thus, mainstreaming is a borrowed concept from other fields, and it is similarly applied under the context of "integration" of CCA (Agrawala 2006; Persson & Klein 2008; Ayers & Dodman 2010; Biesbroek et al. 2010; Chuku 2010; Olhoff & Schaer 2010; UNDP-UNEP 2011; Uittenbroek et al. 2013; Ayers et al. 2014; Giupponi 2014). It is associated with the terms "inclusion," "consideration," and "accounting for" climate change and adaptation in development decision-making (Huxtable & Nguyen 2009; Lasco et al. 2009; Pasquini et al. 2013). Thus, mainstreaming CCA is the process of integrating climate change and adaptation into development planning, programs, projects, policies, and goals, as well as sectoral decision-making, policy-making, budgeting, implementation and monitoring practices and processes (i.e., poverty reduction, livelihood security, water, environment, agriculture, land-use, etc.) (Agrawala 2006; USAID 2009; UNDP-UNEP 2011; Carlson 2012; Ogato 2013).

Mainstreaming CCA is founded on the notion that achieving sustainable development is difficult and cannot be achieved without considering climate change impacts. Likewise, effective climate adaptation is unlikely to happen without accounting for existing and future development actions (Olhoff & Schaer 2010; Schipper et al. 2010). Thus, in mainstreaming, the focus of both CCA and development planning shifts; the path or direction of one is considered in the other and vice versa. Essentially, mainstreaming is the synergy of climate change and sustainable development goals and agendas (Ayers & Doman 2010; Field et al. 2014; IPCC 2014a).

The growing popularity of mainstreaming stems from the notion that sustainable development and CCA are mutually dependent strategies (Ayers et al. 2014; Noble et al. 2014) that can be beneficial to each other. Likewise, when not attuned, each can be detrimental to the other (Klein et al. 2005; IPCC 2007). For example, the observed glacier retreat in Nepal has been attributed to climate change, with the expected risk of glacial lake outburst flooding in the area. This scenario highlights that if climate risks are not considered comprehensively in the country's development planning (i.e., hydropower system design or rural development programs), then the nation's development goals may not be attained under

future conditions (Agrawala & van Aalst 2006). In Ethiopia, future rainfall variability is associated with a high degree of uncertainty. Still, the current higher temperatures and changes in rainfall patterns are expected to influence water availability in the African river basins. Such probable climate risk needs to be integrated into the country's water management and planning to ensure sufficient and safe future water supplies (Conway & Schipper 2011; Oates et al. 2011; Ogato 2013). Water stress is also possible in Egypt where higher temperatures and rainfall pattern changes are likely to affect the Nile River. The climate threat is deeply interwoven in Egypt's economy as 95% of its freshwater needs for household and agricultural use is sourced from the river (Risbey et al. 2006).

Similarly, present development activities may obstruct future adaptation efforts or raise the vulnerability of people and communities if development and adaptation endeavours are not harmonised. For example, clearing of mangroves for commercial purposes or for human settlement can create irreversible consequences in areas likely to be subjected to climate change (OECD 2006). Such resources have been instrumental in minimising the impacts of climate change (i.e., intensified typhoons) in certain areas of the world. For example, the communities in some small islands in the Philippines survived during the most powerful typhoon in recent history, Typhoon Yolanda (international name: Typhoon Haiyan), because of the mangroves in the areas which acted as barriers to the threatening storm surges (Chatterjee 2013; David et al. 2013; Holtz 2013; NASA 2013). Hence, while CCA and development historically have been administered in separate fields, the needs under present and future climate conditions necessitate that they be addressed within a single domain; thus, the need for the mainstreaming approach (Ayers et al. 2014).

2.2.2 Mainstreaming climate change adaptation and institutional linkages

This section establishes the linkages between CCA and institutions, in general, and mainstreaming CCA and institutions, in particular. Fundamentally, it explains why mainstreaming operationalisation needs an institutional perspective.

In dealing with climate change, it is crucial to understand that climate change is a "wicked" problem (FitzGibbon & Mensah 2012; Termeer et al. 2013; Head 2014; Perry 2015). The concept of wicked problems originated from the work of Rittel and Weber (1973) who coined the term to describe social and policy planning problems. They used the term wicked in "a meaning akin to that is malignant (in contrast to benign) or vicious (like a circle) or tricky (like a leprechaun) or aggressive (like a lion, in contrast to the docility of a lamb)" (Rittel & Weber 1973, p. 60). Wicked problems are difficult to address because they are

unsolvable; "at best they are only re-solved-over and over again" (Rittel & Weber 1973, p. 60). Climate change, as a wicked problem, is ambiguous (i.e., defined in varying ways), open ended (i.e., challenging to delineate the boundaries of its effect), unpredictable and intractable (i.e., addressing one generally results to unintended generation of new sets of wicked problems), and multifaceted (i.e., caused by multiple factors from multiple sources) (FitzGibbon & Mensah 2012; Termeer 2013; Head & Alford 2015). Accordingly, CCA has been referred to as a "wicked problem par excellence" (Termeer et al. 2013, p. 27).

Linking CCA with wicked problems expresses the complexity involved in the adaptation process. As such, the traditional methodologies (i.e. scientific and technical) and the available tools in policy analysis are ill-equipped to address CCA issues and concerns. This is so because CCA involves multi-level interventions spanning across households, communities, governments, NGOs, industries, and different sectors at several scales (from local to regional, national and international). Likewise, it warrants changes or adjustments in behaviours and value systems (Pettengell 2010; Berrang-Ford et al. 2011; Hamin et al. 2014; Perry 2015). Thus, as a wicked problem, CCA necessitates interdisciplinary approaches, interactive communication and governance strategies, and integration of varying knowledge systems (Ludwig 2001; FitzGibbon & Mensah 2012). Most importantly, CCA needs to be addressed through institutional means and perspective (Jentoft & Chuenpagdee 2009; Rodima-Taylor et al. 2011; Perry 2015).

The institutional dimension of climate change is critical in adaptation, especially in mainstreaming (OECD 2006; Agrawala & van Aalst 2006; UNDP-UNEP 2011). The whole concept of mainstreaming (i.e., synergy of climate change and sustainable development goals and agenda; and designing new or redesigning existing planning, policy-making, and decision-making structures) is an institutional concern (Young 2002; Ayers & Doman 2010; Field et al. 2014; IPCC 2014a, 2014b) that entails:

(1) multilevel institutional coordination between different political and administrative levels in society; (2) key actors, advocates, and champions initiating, mainstreaming, and sustaining momentum for climate adaptation; (3) horizontal interplay between sectors, actors, and policies operating at similar administrative levels; (4) political dimensions in planning and implementation; and (5) coordination between formal governmental, administrative agencies, and private sectors and stakeholders to increase efficiency, representation, and support for climate adaptation measures (Mimura et al. 2014, p. 871).

The institutional changes created by mainstreaming CCA will affect some areas of reality that are already exposed to existing institutions, which prompts the need to understand

the environment where these institutional changes (e.g., creation of new policies or amendments in prevailing regulations) are to be implemented (Theesfeld et al. 2010). This insight is consistent with the growing realisation of the significance of institutions in CCA research (Adger et al. 2005; Inderberg & Eikeland 2009; Rodima-Taylor 2012). However, the institutional dimension of CCA is the least understood aspect of the challenge (Evans & Stevens 2009; Rodima-Taylor 2012; Rodima-Taylor et al. 2012; Pradhan et al. 2012).

This lack of understanding may be due to the complexity of an institutional analysis; it involves complicated concepts such as institutional arrangements and institutional interplays. Institutional arrangements are the specific guidelines designed to facilitate social interactions. They are the sets of rules or agreements that govern the activities of people and guide individual behaviours toward collective actions (Klein 2000). Meanwhile, institutional interplays are the interactions among institutions that build institutional relationships (Young 2002). Institutional interplays (1) are determined by the impact of one institution on another; (2) are multi-directional and involve functional interdependencies; and (3) include mutual influences or effects among institutions (Young 2002; Linner 2006). This suggests that institutional interplay is an institutional linkage that can result from institutional integration. In this sense, institutional interplay has a significant role in mainstreaming—which is essentially integrating CCA concerns into existing and functioning institutional settings (La Trobe & Davis 2005; Lebel et al. 2012).

Institutional analysis is significant in adaptation, but it is a key component in mainstreaming CCA. As such, the concept of institutions in the context of CCA must be examined. It is difficult to form strategies and actions to solve a problem if significant factors involved in the process are not defined clearly (Forsyth 2012). Having an inconsistent idea of what constitutes an institution may result in conceptual confusion. Consequently, this may make those problems involving complex institutions difficult to understand and address, especially in adaptation planning (Morrison 2006).

As mentioned earlier, adaptation involves multi-scale interventions that span households and communities, involve national and international dimensions, and encompass complex adjustments in behaviour and actions at all levels of society (Pettengell 2010; Berrang-Ford et al. 2011; Perry 2015). Moreover, climate change is a wicked problem (Lazarus 2010; FitzGibbon & Mensah 2012; Termeer et al. 2013; Head 2014; Perry 2015), hence, is a "cross-boundary, multilevel, multi-sectoral and multi-actor challenge" (Fröhlich & Knieling 2013, p. 21). Therefore, institutions under the climate change context: ...should have a synthesis of definition that has cross-disciplinary relevance. Therefore, institutions are the commonly known and acknowledged rules, social structures, and organisations founded on common belief systems that transform individual acts and expectations into collective actions, convert personal values into social norms and shared beliefs, and define the formal and informal behavioral systems of human existence. Rules, social structures, and organisations are all institutions (Cuevas et al. 2014, p. 2).

This chapter therefore argues that the mainstreaming process necessitates institutional analysis—a complicated process that needs to examine institutional arrangements and institutional interplays (Jordan & O'Riordan 1997; O'Riordan & Jordan 1999; Young 2002).

2.2.3 Sectoral mainstreaming: land-use planning

Mainstreaming CCA across sectors is crucial in adaptation and sustainable development planning (Dovers & Hezri 2010). Sustainable development is affected by performances of the varying sectors, and these sectors are impacted by climate change (Ayers et al. 2014; Noble et al. 2014). Moreover, vulnerable people (i.e., poor) rely heavily on climate-sensitive sectors such as agriculture and fishery; thus, addressing the effects of climate change in these sectors becomes a priority (Huxtable & Yen 2009). Attending to the vulnerability of a sector to climate change is an effective way of improving the adaptive capacity of a system, which would reduce the risk of that system to climate change (Table 1) (USAID 2009; Cuevas 2011).

Mainstreaming CCA into land-use planning is one of the most important long term adaptation measures to respond to climate change. This is because land-use planning reflects the (1) rational allocation of space (Jordhal, Jr. 1984; Stewart et al. 2004); (2) the systematic assessment of land-use (FAO 1993); (3) community consensus on debated issues about development and infrastructure, etc. (Kaiser & Godschalk 1995); and (4) management of urban growth and change (Burby et al. 2000; Godschalk 2004; HLURB 2006). Consequently, scholars have explored the linkages between land-use planning and a wide variety of concerns, including ecology, wildlife and conservation, natural resources, welfare economics, transport, disaster and hazards, sustainable development, and lately, CCA (Cheshire & Sheppard 2002; Marshall & Banister 2008; Kaswamila & Songorwa 2009; Tang et al. 2009; Wang 2012; Silberstein & Maser 2013; Gottlieb 2014; Schmitz et al. 2015).

Table 1	Exampl	es of	mainstr	reaming	by	sector
	1			0	~	

Sector	Description
Land-use planning	Expanding the planning horizons of land-use plans to incorporate longer climate predictions
	Planning development according to the medium- to long-term risks posed by climate change on varying geographical zones (i.e., coastal areas vulnerable to sea level rise and storm surges) Revising regulations and standards to reflect climate variability (e.g., areas available for human settlements)
Infrastructure	Introducing disaster risk and climate change assessments in the construction of new roads, bridges, and other major infrastructures for an informed decision-making (i.e., infrastructure designs, materials used, construction techniques, etc.) Using hazard maps, climate forecasts, and other climate-related data to avoid building new infrastructure in areas at high risk of forest fires, flooding, or storm surges Considering climate change projections in designing sewage systems, drains, and
Agriculture	Incorporating climate change adaptation in farming practices, irrigation system designs, community development plans and projects, etc.
Education	Promoting hazard-resilient construction for new and existing schools Incorporating effects of climate change on access to sanitation and safe water in designing schools and in planning for school programs
Water	Including climate forecasts, water resource assessments, and current natural hazard profiles in the designs of new programs (i.e., water safety planning)

Sources: OECD 2009; USAID 2009; AusAID 2010; UNDP-UNEP 2011

Over time, land-use planning concerns and processes have evolved based on society's changing needs and capacities. For instance, land-use planning in the late 20th century has been concerned with synthesising ecological conservation and economic development (van Lier 1998). Likewise, the planning process has progressed from "simple roots in civic design and zoning into an intricate combination of design, policy, and management" (Kaiser & Godschalk 1995, p. 365). On the other hand, land-use planning in the 21st century leans toward achieving sustainable development and addressing climate change (Godschalk 2004; IPCC 2014a), while the planning process has advanced to programming land-use models, utilisation of geographic information systems, and cross-scale and inter-sectoral evaluation and assessments (Dai et al. 2001; Stewart et al. 2004; Verburg et al. 2004). Land-use planning is expected to change further as advancements and practices in CCA planning are incorporated into the (planning) process.

However, integrating these two planning processes is complicated. Land-use and climate change have an intricate relationship that emerges from the impacts of climate change on both the supply of and demand for land (Koomen et al. 2008; Tang et al. 2009). To illustrate, future climate conditions can include accelerated sea level rise, intensified rainfall,
greater extreme temperatures, increased droughts and floods associated with El Nino events, and stronger cyclones and typhoons. The increased quantity of rainfall is projected to raise the incidence of floods, landslides, avalanches, and mudslides and aggravate soil erosion. In turn, the higher incidence of drought and flood are predicted to lessen the agricultural and rangeland productivity in prone regions; whereas the intensified storms will likely increase disaster events and damage to life and infrastructure. Meanwhile, the accelerated sea level rise is expected to inundate low-lying lands, cause greater coastal erosion, aggravate shoreline recession, decrease the number of coastal wetlands, and worsen coastal flooding and submergence of coastal lands (IPCC 2001, 2007, 2014a, 2014b; Ward 2011).

These climate change-related incidents will affect land-uses and will put pressure on decision-makers to address land-use changes. As climate change alters the physical characteristics of land, it will affect land productivity and the land's suitability for certain types of uses (Koomen et al. 2008). Given that land is a limited resource, its supply for certain land-uses (i.e., agriculture, forestry, etc.) will be impacted (Quan & Dyer 2008). Similarly, demand for land will change. For example, good grazing grounds during extreme droughts will be needed; safe shelters (i.e., emergency, transitional and permanent) during disasters caused by intensified typhoons, hurricanes or storms will be required; and new settlement areas will be demanded as people are forced to leave their homes due to varying kinds of climate-related circumstances (i.e., water shortage, flooding) (Orindi & Eriksen 2005; Freudenberger & Miller 2010; UN-HABITAT 2010).

Likewise, the implementation of adaptation measures can affect the supply of and demand for land because "future development options may be confined, and new spatial conflicts, e.g., between risk prevention and land-use interests, may emerge" (Putz et al. 2011, p. 4). For instance, maintaining mangroves as barriers to storm surges during typhoons or expanding risk-zone areas in anticipation of future hazards and disasters will lessen the available land for commercial purposes or for human settlements. Similarly, changing farming practices in rain-fed farms to adapt to extended drought may create a higher demand for irrigated lands (Freudenberger & Miller 2010; Putz et al. 2011).

Clearly, climate change and land-use linkages are complex. On one hand, the physical impacts of climate change on land influence the use, productivity, and access to land. On the other hand, the adaptive interventions also have a variety of implications for land-use (Quan & Dyer 2008). This complexity magnifies the significance of integrating the two planning dimensions (i.e., CCA and land-use) and the urgency of applying "climate conscious" planning (Lindley et al. 2006, p. 545).

Land-use planning can be applied at the national level where planning is concerned with (1) balancing the demand and supply of land among the varying economic sectors; (2) allocating resources for development; (3) coordinating national agencies for land-use; and (4) institutionalising national legislation for land tenure and resource rights (i.e., water, forest). On the other hand, the local level is where planning is focused on the actual use of land in specific areas. It is concerned with who can use the land, and what uses are permissible where and when (FAO 1993). Local land-use planning encompasses the entire planning area of the local jurisdiction, and thus it is crucial for local land management and local development (Tang et al. 2009). Meanwhile, "adapting to climate change is, in many ways, a local issue" (Hamin & Gurran 2014, p. 1), because it is at this level where the direct impacts of climate change are experienced (Sharma & Tomar 2010). Likewise, vulnerabilities and adaptive capacities of systems are founded by local conditions thereby substantiating the notion that adaptation is most effective at the local level (OECD 2009).

Accordingly, mainstreaming CCA into the local land-use planning is an effective means by which to address climate change, and is the response that establishes long-term adaptation action (Pasquini & Shearing 2014; Picketts et al. 2014; Rauken et al. 2015). This is because local land-use planning is the "constitution for future development" (Tang et al. 2009, p. 368), and mainstreaming CCA into local land-use plans enhances the capability of communities to address the present and expected climate change risks, and to respond to and recover from climate change impacts (IPCC 2014a). This approach, therefore, sets the foundation of future development that is sustainable under current, foreseen, and uncertain climate conditions.

2.4 Developments in mainstreaming climate change adaptation

The preceding discussion presented the mainstreaming approach and the key ideas surrounding the concept. This section outlines how mainstreaming gained popularity as a CCA approach. First, it traces the events that led to the policies that support mainstreaming of CCA. Second, it illustrates the conditions in relation to planning and last, it describes the methodologies and approaches surrounding the operationalisation of mainstreaming.

2.4.1 Mainstreaming in policy-making

This section examines the last three Intergovernmental Panel on Climate Change (IPCC) Working Group II's Assessment Reports (AR) (i.e., 3rd in 2001, 4th in 2007, and 5th in 2014) to determine how mainstreaming CCA has developed over the years as a topic in

adaptation research. First, the number of times the term "mainstream" or "mainstreaming" CCA appeared in the reports was collated (Figure 1). The term was not mentioned in the IPCC AR3; mainstreaming first materialised in AR4 where it was formally defined in Chapters 14 and 17 as:

... the integration of policies and measures that address climate change into development planning and ongoing sectoral decision-making (Klein et al. 2007, p. 768 [Chapter 14]).

... the integration of climate change vulnerabilities or adaptation into some aspect of related government policy such as water management, disaster preparedness and emergency planning or land-use planning (Adger et al. 2007, p. 732 [Chapter 17]).





Sources: IPCC 2001, 2007, 2014a, 2014b

Thus, the initial attention given by the IPCC AR4 to mainstreaming CCA influenced the interest of scholars on the approach (Dovers & Hezri 2010; Noble et al. 2014). In 2014, mainstreaming has become more pronounced in IPCC AR5. In this report, mainstreaming discussions have gone beyond the conceptualisation of the term into reporting its actual application. For example, in the Technical Summary chapter, Field et al. (2014) stated that:

In Asia, adaptation is being facilitated in some areas through mainstreaming climate adaptation action into subnational development planning, early warning systems, integrated water resources management, agroforestry, and coastal reforestation of mangroves (p. 51).

Similarly, Wong et al. (2014) mentioned in Chapter 5 that:

In Japan, coastal climate change adaptation has been mainstreamed into the framework of Coastal Disaster Management in the aftermath of the 2011 Tohoku Earthquake Tsunami (p. 390).

The number of times that the term "mainstream" or "mainstreaming" was incorporated into the title of references used in the IPCC AR4 and AR5 was also examined. Based on the frequency of the unique entries of such references in the reports (i.e., peer reviewed papers), it was concluded that the literature which focused on mainstreaming CCA is increasing (Figure 1).

The rising popularity of mainstreaming CCA is influenced largely by international funding mechanisms supporting the adaptation $approach^1$ (Measham et al. 2011; Lal et al. 2012). For example, the Least Developed Countries Fund, established under the United Nations Framework Convention on Climate Change, provides financial support to the formulation and implementation of the National Adaptation Programs of Action (NAPA) (Biagini & Dobardzic 2011). NAPAs are the means for least developed countries to convey their most pressing adaptation needs; they are also the foundations from which least developed countries build their National Adaptation Plans (NAPs) (Kissinger & Namgyel 2014; Mimura 2014). Whereas the NAPA identifies the country's short-term and urgent adaptation actions, the NAP determines its medium- and long-term adaptation needs and outlines the strategies and schemes to address these needs (Noble et al. 2014). Thus, in these countries, CCA has been embedded in the development framework through the NAPAs and NAPs (Mimura et al. 2014). Specifically, concerns related to the adaptation approach include (1) ensuring CCA is mainstreamed effectively into national development; (2) mainstreaming adaptation across core development sectors; (3) using the Least Developed Countries Fund to finance CCA mainstreaming efforts; (4) recognising the barriers to mainstreaming adaptation; and (5) using mainstreaming to transform the planning processes and the promote long-term resilience (Huq et al. 2004; Biagini & Dobardzic 2011; LDC Expert Group 2012).

Advancement in mainstreaming CCA is likewise shaped by the advocacy of the donor, bilateral, and multilateral agencies (Agrawala 2006; Olhoff & C. Schaer 2010; Lal et al. 2012). International funding agencies are concerned with (1) the impacts of climate change on the operationalisation of the agencies' respective projects and investments; and (2)

¹ e.g. Least Developed Countries Fund, Special Climate Change Fund, Multi-donor Trust Fund on Climate Change, Pilot Programme for Climate Resilience under the Climate Investment Fund, Global Environment Facility Trust Fund, and the Adaptation Fund

the climate risks in the countries' own development efforts. Hence, these agencies "push" for the mainstreaming approach. Within this context, these organisational institutions are climate-proofing² their investments and making these investments relevant under future climate conditions (ADB 2005; van Aalst & Agrawala 2006; Ayers et al. 2014).

In 2006, the Organization for Economic Co-operation and Development (OECD) released the Declaration on Integrating Climate Change Adaptation into Development Cooperation. In this proclamation, OECD member countries and members of the European community agreed to "work to better integrate climate change adaptation in development planning and assistance, both within their own governments and in activities undertaken with partner countries" (OECD 2006, p. 6). The declaration was institutionalised in view of "helping vulnerable countries anticipate and adapt to the risks posed by climate variability and climate change," thereby assisting these countries achieve their development goals (OECD 2006, p. 5). Following the declaration, the OECD published the Policy Guidance on Integrating Climate Change Adaptation into Development Co-operation, a document identifying strategies and approaches for mainstreaming CCA into development policies at the national, sectoral, and project levels (OECD 2009; Olhoff & Schaer 2010; Uittenbroek et al. 2013). Similarly, other international and development organisations such as the Asian Development Bank, the World Bank, and the United Nations agencies (i.e., UNEP, UNDP) have been interested in mainstreaming CCA and have developed tools and guidelines to promote the approach (Agrawala 2006; Agrawala & van Aalst 2006; Ayers & Dodman 2010; Lal et al. 2012). These handbooks and related documents provide generic guidance for mainstreaming at the national, sectoral, programme and project levels.

2.4.2 Mainstreaming in planning

Mainstreaming CCA is a multi-scale endeavour. In general, mainstreaming in development plans begins at the national scale where the general framework, within which sectoral and other sub-national levels operate, is presented. At this scale, national policy goals, long-term visions, and development strategies are outlined (Lebel et al. 2012). For example, the Twelfth Five Year Plan (2012–2017) of India mainstreamed CCA through the vulnerability assessments it required from the various sectors of the plan (Planning Commission 2013). Likewise, the National Sustainable Development Strategy 2010–2021 of Bangladesh moved from the traditional planning and budgeting practices to more strategic

² Ensuring climate risks are reduced to acceptable levels through changes implemented in varying stages of a project or planning cycle.

ones by including the environment and climate change issues in the process (Planning Commission 2010). The Bangladesh Climate Change Strategy and Action Plan 2009 also specified mainstreaming climate change into the sectoral and spatial development planning of government ministries and local governments in order to protect vulnerable groups from the impacts of climate change (MoEF 2009).

Meanwhile, the Philippine Development Plan 2011–2016 listed mainstreaming of CCA and disaster risk reduction and management (DRRM) into the "existing policies (i.e., land-use, building code), plans (i.e., comprehensive land-use plan) and programs (i.e., researches, school curricula)" among the plan's priority short-term activities (NDRRMC 2011, p. 15). Similarly, the Philippine National Climate Change Action Plan 2011–2028 outlined the mainstreaming of CCA and DRRM into local plans as a strategy to attain human security, along with climate-proofing energy systems and infrastructures to achieve sustainable energy. The plan also sets climate change adaptive housing and land-use development as target outputs under the climate-smart industries and services agenda (CCC 2011).

While the national plans establish the general direction of mainstreaming CCA, the sectoral level and the local scale plans carry out the specific mainstreaming actions (Lebel et al. 2012). However, as a relatively new approach, there is still little understanding of what mainstreaming requires in practice. Consequently, operationalisation of mainstreaming has been slow, and in some cases, has not been implemented effectively (Uittenbroek et al. 2013; Revi et al. 2014; Lehmann et al. 2015). For example, "progress in mainstreaming adaptation in some of the vulnerable sectors in India is still negligible (Nambi & Prabhakar 2011, p. 444) because of the barriers to adaptation encountered in its on-ground application. Likewise, mainstreaming in Bangladesh faced challenges that caused delays in its operationalisation, such as "inadequate coordination mechanisms among various ministries and line agencies, limited coordination capacity of the Ministry of Environment and Forests (MOEF) and other implementing agencies, losses of institutional memory in relevant agencies and 'brain drain' of trained officials" (Ayers et al. 2014, p. 302). Meanwhile, local planners in the Philippines found it difficult to mainstream CCA due to the lack of "formally issued implementing policy that would outline a clear methodology, procedure and standards on the integration of CCA and DRRM into local plans" (Mercado 2011, p. 7). Accordingly, local government units in the country clamoured for the completion of the Reference Manual on Mainstreaming Disaster Risk Reduction/Climate Change Adaptation in the Comprehensive Land-use Plans to

help local planners (RDC XII 2012).³ Similar challenges and delays were experienced by countries such as Vietnam, Thailand, and Bhutan (Lhendup 2012; Sinh & Toan 2012; Chinvanno & Kerdsuk 2013). Thus, in practice, the operationalisation of mainstreaming has not advanced as fast as its conceptualisation due to the barriers or challenges that affect the mainstreaming process (Huq & Ayer 2008; Lal et al. 2012; Uittenbroek et al. 2013; Ayers et al., 2014; Revi et al. 2014; Lehmann et al. 2015; Rauken et al. 2015).

The subject of barriers to adaptation is a pressing theme in adaptation research, not only in mainstreaming. The term "barriers", as used in the literature, has been referred to as challenges (Mitchell et al. 2006; Burch 2010; Pervin et al. 2013), constraints (Moser et al. 2008; Amundsen et al. 2010; Pasquini et al. 2013; Dang et al. 2014), or limitations (Measham et al. 2011; Oberlack & Eisenack 2014; Picketts et al. 2014). Ekstrom et al. (2011, p. 1) defined barriers as the "obstacles that delay, divert, or temporarily block the adaptation process." Early works on the subject focused on identifying these barriers. Research interest, however, has transcended into (1) knowing the nature of the barriers; (2) determining how the barriers are classified; and (3) understanding how the barriers exist and persist (Amundsen et al. 2010; Farrell 2010; Moser & Ekstrom 2010; Roberts 2010; Sharma & Tomar 2010; Ekstrom et al. 2011). The current knowledge quest has extended to (1) defining the conditions and circumstance surrounding the barriers; (2) evaluating the impacts of the barriers to the adaptation process: and (3) resolving how to overcome the barriers (Burch 2010; Biesbroek et al. 2011; Eisenack et al. 2014; Klein et al. 2014).

The barriers to adaptation have been investigated widely (Amundsen et al. 2010; Burch 2010; Moser & Ekstrom 2010; Ekstrom et al. 2011; Eisenack et al. 2014; Waters et al. 2014), but research on the barriers to mainstreaming CCA is limited in comparison (Biesbroek et al. 2011; Nambi & Prabhakar 2011; Chevallier 2012; Pasquini et al. 2013; Lehmann et al. 2015). With the rapid advancement of mainstreaming CCA into the national planning and policy-making processes of developing countries, there is an immediate need to understand the barriers impeding its effective application on-ground, and to determine how to overcome these barriers.

2.4.3 Operationalising mainstreaming of climate change adaptation

This section analyses the practical implementation of mainstreaming as evidenced by handbooks and guidelines. Specifically, it discusses the information gathered from around 30

³ The Housing and Land-use Regulatory Board released the "Supplemental Guidelines on Mainstreaming Climate and Disaster Risks in the Comprehensive Land-use Plan" in early 2014.

handbooks, guidelines, or mainstreaming documents, published between the years of the IPCC AR4 (2007) and AR5 (2014a, 2014b) releases. These documents were identified through a web search using the key words "handbook," "guide," "guidelines," and "mainstreaming climate change adaptation." This analysis confirmed the findings of Olhoff and Schaer (2010) that these documents typically presented generic guides, ideas, or conceptual frameworks on how to operationalise mainstreaming. This is because the "development of operational measures on integrating adaptation considerations within development activities is still at an early stage" (Gigli & Agrawala 2007, p. 10).

In the context of planning, one of the main problems in mainstreaming stems from the uncertainty about how to proceed with the approach (Hamin et al. 2014). Even with frameworks to follow, planners find it difficult to apply the approach in practice because of the challenges they encounter during the mainstreaming process (Ayers et al. 2014). Consequently, the majority of mainstreaming studies cites the barriers or challenges in mainstreaming CCA, rather than report on successful mainstreaming actions (Sharma & Tomar 2010; Nambi & Prabhakar 2011; Pasquini et al. 2013; Ayers, et al. 2014; Uittenbroek et al. 2014).

A few mainstreaming documents have offered detailed procedures on how to operationalise the approach, and an important concept they have introduced is the mainstreaming "entry point" (i.e., opportunity for mainstreaming) (Huxtable & Yen 2009; OECD 2009; Olhoff & Schaer 2010; UNDP-UNEP 2011; SPREP & UNDP 2013). An entry point can be through either (1) the level, scale, or scope of governance; or (2) the decision-making cycle. The former speaks of assessing climate-development linkages, raising awareness and building partnerships, and evaluating adaptation and capacity needs. Entry points at the national or regional scale include poverty reduction strategy papers, national development plans, national budget allocation processes, NAPA, etc., while the sectoral-level entry points can be through the fishery, water management, land-use, or education sectors. Finally, entry points at the sub-national levels can involve decentralisation policies, district plans, sub-national budgets (activity-specific) or municipalities, districts, provinces, ecosystems, watersheds, etc. (place-specific) (USAID 2009; UNDP-UNEP 2011).

The second perspective—via the decision-making cycle—considers entry points to be opportunities for identifying, implementing, or incorporating measures that support CCA into a given plan, policy, or project (OECD 2009; Olhoff & Schaer 2010). This outlook links the mainstreaming approach to the concept of a "climate lens" (USAID 2009; Hammil & Tanner 2011). Applying a climate lens entails analysing the (1) degree by which the policy, strategy,

or plan may be vulnerable to climate change; (2) degree by which the climate change risks have been accounted in formulating the policy, strategy, or plan; and (3) degree by which the policy, strategy or plan can exacerbate the impacts of climate change, thus, result to maladaptation (i.e., increased vulnerability to climate change) (OECD 2009).

In particular, applying the climate lens speaks of "an analytical tool to examine a strategy, policy, plan, programme or regulation" (OECD 2009, p.17). However, the climate lens is a recent concept (USAID 2009). Its mechanism has not yet been established, such that the OECD (2009) suggested the use of the Strategic Environmental Assessment (SEA) for the task. SEA refers to a variety of "analytical and participatory approaches that aim to integrate environmental considerations into policies, plans and programs" (OECD 2009, p. 185). Although the SEA is effective in integrating environmental concerns into the policy and planning processes to achieve sustainable development, it is limited in addressing climate change concept surrounded by multiple linkages and high levels of uncertainty (Lazarus 2010; FitzGibbon & Mensah 2012; Termeer et al. 2013). This "wickedness" translates to CCA; hence, CCA is a concern "generated by multiple factors from multiple sources" where causal linkages are difficult to define (FitzGibbon & Mensah, 2012, p. 2). Accordingly, current planning practices and techniques need to respond to the challenges associated with climate change (Hamin & Gurran 2014).

A methodology that specifically addresses climate change is the Climate Vulnerability and Capacity Analysis (CVCA). This methodology is recommended by CARE International for applying the climate lens, and has been used to understand how climate change will impact communities (i.e., hazards, vulnerability, and adaptive capacities) (Daze et al. 2009). Because CVCA focuses on community-based adaptation, the methodology links climate and institutional issues in the analytical process. This is an important facet of the methodology because, as argued earlier, mainstreaming CCA is operationalised in an institutional setting. Still, CVCA cannot address the difficulties in the operationalisation of mainstreaming because it lacks focus on a crucial aspect of adaptation—the barriers or challenges to adaptation (Amundsen et al. 2010; Eisenack et al. 2014; Oberlack & Eisenack 2014).

The Community-Based Risk Screening Tool–Adaptation and Livelihoods (CRiSTAL) addresses this limitation. CRisTAL is a software tool that assists local communities and project planners evaluate climate risks in planned and on-going development projects (Gigli and Agrawala 2007). CRiSTAL, advocated by the International Institute for Sustainable Development for applying the climate lens, acknowledges the significance of identifying the

barriers and opportunities for the implementation of an adaptation project (IISD 2012). However, the tool does not answer important questions about barriers to adaptation, such as (1) What is the nature of the barrier? (2) How do the barriers exist and persist? (3) What are the conditions and circumstance surrounding the barriers? (4) How can researchers assess the impacts of the barriers on the adaptation process? and (5) How can the barriers be overcome? (Burch 2010; Biesbroek et al. 2011; Clar et al. 2013; Eisenack et al. 2014). Therefore, the tool is limited in addressing the barrier concern.

Other tools for integrating climate change concerns into projects and plans show similar shortcomings. The Assessment and Design for Adaptation to Climate Change–A Prototype Tool (ADAPT) of the World Bank is primarily a risk-screening tool. Similarly, the Opportunities and Risks from Climate Change and Disasters of the UK Department for International Development concentrates on identifying activities at high risk to climate change and on determining those that provide opportunities for vulnerability and risk reduction (Gigli & Agrawala 2007). These tools focus on the climate change-related aspect of adaptation and overlook the institutional facet of, as well as the barriers to, adaptation.

2.6 Summary and conclusions

Mainstreaming CCA is an adaptation approach that has been growing in popularity in recent decades as evidenced by (1) its prevalence in the IPCC's last two reports (i.e., AR4 and AR5); (2) the growing scholarly interest in the topic; (3) the financial support provided for the approach's adoption in developing countries; and (4) its presence in the NAPA and NAPs of these countries. The appeal of the approach emerges from the integral characteristic of mainstreaming—it is a synergy of CCA and sustainable development goals and agenda (IPCC 2007, 2014a; Biagini & Dobardzic 2011; Ayers et al. 2014; Noble et al. 2014). However, while the questions on the "whys" of mainstreaming CCA have been resolved, the answers on the "how" queries are still lagging. Although mainstreaming is a simple concept, it is poorly understood, which makes it challenging to operationalise (Oates 2011).

Mainstreaming handbooks, guidelines, and related documents have been produced to assist in operationalising the approach (Agrawala 2006; Agrawala & van Aalst 2006; Mitchell et al. 2006; Ayers & Dodman 2010; Lal et al. 2012). However, scholars and practitioners alike agree that the available materials are limited in addressing the difficulties in operationalising mainstreaming (Tang et al. 2009; Ayers et al. 2014; Goosen et al. 2014). Thus, it is often "argued that mainstreaming is not yet sufficiently taking place" (Lehmann et al. 2015, p. 93); and that based on empirical evidence, in practice, "actors are searching for

solutions to integrate the adaptation objective in existing policy domains" (Uittenbroek et al. 2013, p. 399). These notions suggest that there is a gap between the theoretical and practical spheres of the mainstreaming approach.

This chapter therefore addressed the first research objective by proposing that this gap may be due to lack of focus on the institutional facet of mainstreaming operationalisation. Current operational procedures for mainstreaming and the tools and techniques for application concentrate on science of climate change and other climate change-related concerns (i.e., climate change scenario building, vulnerability assessments, climate risk screening, climate change impact analysis, and the like) (OECD 2009; USAID 2009; UNDP-UNEP 2011). However, the operationalisation of mainstreaming needs to go beyond the issues of climate change (Ayers et al. 2014). First, CCA, in general, is a wicked problem that is difficult to tackle using the traditional scientific and technical methodologies; rather, it can be addressed effectively through institutional means. Second, mainstreaming CCA creates institutional changes and entails institutional transformations, and thus is an institutional concern. Yet, the institutional dimension of mainstreaming currently is neglected in its operationalisation.

Likewise, the methodologies for mainstreaming concentrate on what to integrate into existing plans in order to respond to climate change, such as the vulnerability and risks of populations, sectors, communities, etc. They overlook the challenges encountered in the integration process. Consequently, most mainstreaming studies cite the barriers or challenges in operationalising the approach rather than report on successful mainstreaming actions. This review, therefore, asserts that mainstreaming methodologies need to (1) incorporate an institutional perspective in the operational analysis of mainstreaming; and (2) place emphasis on identifying, understanding, and determining the origins of the barriers to operationalisation.

Advancing the knowledge base on the barriers to adaptation needs methodical planning because studying the challenges or barriers to adaptation needs to be context-specific (Biesbroek et al. 2013). It is feasible to generalise their nature and characteristics, but the manner or degree by which the barriers affect the adaptation process depends on individual settings (Mimura et al. 2014). Accordingly, intensive, detailed and rich analyses of the barriers to mainstreaming CCA are possible through a case study approach (Flyvbjerg 2011). These concerns are elaborated in the next chapter where the methodology devised by the research to address the gap in mainstreaming operationalisation is presented in detail.

CHAPTER 3: METHODOLOGY FOR EXAMINING THE CHALLENGES IN MAINSTREAMING CLIMATE CHANGE ADAPTATION*

*Cuevas, S.C., Peterson, A., Morrison, T., & Robinson, C. (In press). Methodology for Examining the Challenges in Mainstreaming Climate Change Adaptation. *International Journal of Climate Change Strategies and Management.*

3.1 Introduction

Chapter 2 illustrated that although interest in mainstreaming climate change adaptation (CCA) is growing, there is limited information on how to operationalise the approach effectively, especially at the local scale (Huxtable & Yen 2009; Olhoff & Schaer 2010). Essentially, in practice, mainstreaming "is not yet sufficiently taking place" (Lehmann et al. 2015, p. 93) and practitioners are still "searching for solutions to integrate the adaptation objective in existing policy domains" (Uittenbroek et al. 2013, p. 399). This research posits that this slow development is caused, in part, by the neglect given to the institutional facet of mainstreaming in operationalisation. Current techniques and operational procedures for mainstreaming, such as vulnerability assessments, impact analysis, risk screening, and the like, focus on climate-related concerns (OECD 2009; USAID 2009; UNDP-UNEP 2011), and overlook (1) the existing institutional settings and institutional arrangements into which CCA will be integrated; (2) the institutional transformations generated by the integration process; and (3) the impacts of these institutional changes to the realities already subjected to the existing institutions. Consequently, mainstreaming CCA encounters a number of difficulties in practice, and most studies on the subject deal with the barriers or challenges in mainstreaming CCA, rather than illustrate effective mainstreaming actions (Lebel et al. 2012; Ayers et al. 2014; Uittenbroek et al. 2014; Lehmann et al. 2015).

Another set-back in mainstreaming operationalisation is the lack of methodologies to investigate how mainstreaming is applied in practice. This is a major gap in the literature primarily because CCA is a wicked problem that is complex, ambiguous, ill-defined, unpredictable, intractable, and multifaceted. The existing CCA methodologies focus on the science of climate change. However, although the technical climate-related issues are significant aspects in adaptation, they are not sufficient to address CCA concerns within the planning and policy-making domains (FitzGibbon & Mensah 2012; Termeer et al. 2013; Head & Alford 2015). Consequently, this condition constrains the capability of researchers, analysts, and practitioners to effectively measure and examine the complex processes involved in mainstreaming CCA (Tang et al. 2009; Ayers et al. 2014).

Likewise, in planning and policy-making, CCA is an "abstract concept" (Persson & Klein 2008, p. 13) that needs to be simplified in a language more familiar to planners and policy-makers. A possible way to address this concern is to develop indicators that can help planners examine and evaluate the state and progress of adaptation efforts. Through quantitative mechanisms, analysts can (1) determine the trade-offs involved in implementing adaptation measures; (2) track the advancements in the implementation; and (3) assess whether the targets of adaptation efforts are attained (Berrang-Ford et al. 2011; Engle 2011; UNDP-UNEP 2011).

This research aims to contribute to this discussion by devising a systematic methodology for examining the challenges in mainstreaming CCA. In particular, it argues that mainstreaming operationalisation necessitates a methodology that focuses on the challenges in applying the approach, and an analytical framework that can examine the mainstreaming process from an institutional perspective. In this research, these challenges refer to the factors that affect the effective operationalisation of mainstreaming CCA at the local scale; and these challenges can be transformed into opportunities that can help the mainstreaming process to be successful. Furthermore, to have in depth insights into these challenges, the research contends that the methodology should be able to monitor and assess the severity of the challenges through metrics; answer causal linkages among challenges; and solve the questions pertaining to the "whys" and "hows" of the subject (Hesse-Biber & Johnson 2013; Weaver-Hightower 2014). Hence, this research utilised a mixed-methods approach to address the task. Since examining the challenges or barriers to adaptation is context specific (Biesbroek et al. 2011), the research conducted a case study in Albay, Philippines to collect robust data and perform intensive and detailed analysis on these data (Flyvbjerg 2011).

This chapter begins with an overview of the four-stage mixed methodology developed by the research. Afterwards, it introduces the Institutional Analysis and Development (IAD) framework, and demonstrates how the IAD can be modified to better fit this CCA research. This is followed by sections on the case study area selection process, the data collection design (i.e., survey, interviews), and the development of quantitative mainstreaming indicators. Later, the issues encountered in applying the mixed methodology and some of the methodology's limitations are presented. Accordingly, this Chapter addresses the second objective of the research which is– *to determine the analytical framework and methodology that can (1) examine effectively the challenges in mainstreaming CCA into land-use* planning; and (2) generate metrics that can be used by planners and decision makers in addressing these challenges.

3.2 Mixed methodology in climate change adaptation research

The research devised a four-stage methodology that (1) included a mixed method that utilised document reviews, interviews, a survey, and key informant consultations (i.e., triangulation by data method) as the main data sources; (2) used the modified Institutional Analysis and Development (IAD) framework as the primary analytical and data collection guide; and (3) employed the scorecard approach to generate quantitative data and indicators. This methodology was developed based on the notion that climate change is a complex (wicked) problem, a cross-sectoral issue, and an inter-disciplinary concern (Huxtable & Nguyen 2009; Nielsen & D'haen 2014).

The current climate change research has lagged behind this inter-disciplinary nature of the problem in terms of research cooperation, citation, and methodologies applied (Bjurstrom & Polk 2011). Thus, an avenue for collaboration among quantitative and qualitative researchers is needed for an effective interdisciplinary communication and cooperation in climate change research (Nielsen & D'haen 2014). Such synergy can be accomplished through the use of a mixed method approach.

Mixed method combines two complementary research methods—qualitative and quantitative—to answer a research question (Hesse-Biber & Johnson 2013). Although this method was formally recognised about thirty years ago, it was only in the last decade that it grew rapidly in popularity (Onwuegbuzie et al. 2009). It is a method that constantly is being tested in varying research problems, conditions, and disciplines (Siddiqui & Fitzgerald 2014). Thus, like CCA research, the mixed method approach is still evolving (Brannen 2005; Onwuegbuzie et al. 2009).

Through a mixed approach, the ability of the quantitative method in establishing or gauging the extent, status, or condition of a phenomenon can be enhanced by the capability of the qualitative method to answer "whys", "hows", and "so whats" queries (Weaver-Hightower 2014). Thus, a combination of the two is generally viewed as the best approach to accomplish an analytical task and answer complex interdisciplinary research questions (Flyvbjerg 2011; Hesse-Biber & Johnson, 2013). In particular, the mixed method in this study was accompanied by data triangulation (i.e., document, interview, survey, and key informant consultations) (Yin 2014).

As mentioned, the research methodology (Figure 2) involved four stages and used quantitative and qualitative methods, which varied according to the stage and purpose of the research (Brannen 2005). The methodology had two key facets, including the modified IAD framework and the case study approach. To illustrate, Stage 1 entailed document reviews and consultations with key informants. Based on the information gathered at this stage, the original evaluation criteria of the IAD were adjusted; and changes were applied to accommodate certain analytical needs of the research. The modified framework—IAD for mainstreaming CCA research (IAD-CCA)—was used as a guide in designing the activities in the next stages of the research.

Stage 2 involved the conduct of a survey among the key actors in the local land-use planning system in Albay and representatives of the national government, non-government agencies, and academic organisations who had experience in implementing projects concerning mainstreaming of CCA in local land-use plans. This design is akin to the "Extended Peer Community" facet of the "Post-Normal Science"⁴ methodology that some scholars (Ludwig 2001; Saloranta 2001; Head 2014) have advocated in addressing wicked problems like climate change. This community is not limited to scientific specialists, public officials and other stakeholders. Rather, it includes all individuals with the interest to partake in resolving the issue. Hence, a variety of perspectives is incorporated in the assessment and decision-making processes (Ravetz 1999; Saloranta 2001)

Meanwhile, a scorecard approach was applied to enable quantification of the respondents' answers, thus allowing for quantitative indicators to be generated (i.e., mainstreaming indicators). The subsequent computed mainstreaming indicator scores helped to establish the key issues that needed to be clarified during the semi-structured in-depth interviews conducted in Stage 3. In turn, the data gathered from the interviews verified and validated the mainstreaming indicator scores and raised additional issues and concerns regarding the mainstreaming process. These (issues and concerns) were further investigated through document reviews and consultations with key informants in Stage 4. All the information gathered were qualitatively analysed using the IAD-CCA framework as a guide (i.e., using the modified evaluation criteria to assess the patterns of interactions and outcomes of these interactions) (Figure 2).

⁴ The post-normal science methodology was introduced by Funtowicz and Ravetz (1993) to manage contemporary problems which the methods and approaches of traditional science are unable to address effectively. Essentially, "the methodology of Post-Normal Science is meant to be applied whenever high stakes, risks and/or high uncertainty are involved in a policy-relevant issue" (Saloranta 2001, p. 396). Post-normal science complements the scientific expertise in order to produce better quality decisions regading complex problems (Saloranta 2001).



Figure 2 The research methodology

3.3 Analytical framework for examining the challenges in mainstreaming climate change adaptation

This section introduces the Institutional Analysis and Development (IAD), argues for the need to modify the framework to suit CCA research and outlines the process by which the IAD was transformed into the IAD as applied in mainstreaming CCA research (i.e., IAD-CCA). Four frameworks (i.e., Advocacy Coalition Framework, Institutional Analysis and Development Framework, Adaptive Capacity Wheel, and Adaptation, Institutions, and Livelihoods) were analysed to determine the most effective framework for examining the challenges in mainstreaming CCA (Table 2). The review showed Ostrom's (2007) IAD to be the most suitable framework for the task. First, the framework is designed specifically to examine institutional settings (Ostrom 2011), so it is equipped to analyse the setting where CCA is to be mainstreamed. Next, the IAD has a variable (i.e., biophysical conditions) that can represent the impacts of climate change concerns in that setting (McGinnis 2011). Furthermore, the IAD has a systematic analytical process that can help users organise vast amounts of data; thus, it is very useful in outlining data collection for CCA research (Koontz 2006; Dick & Meinzen-Dick 2011). Also, the IAD is a framework that has been tested and applied successfully to a variety of institutional conditions and to an extensive range of problems and concerns, including CCA (Koontz 2006; Oberlack & Neumärker 2011).

The IAD focuses on the action arena, which is comprised of institutional arrangements and the actors who follow these arrangements. The action arena is influenced by a number of exogenous variables, namely, biophysical conditions, community attributes, and rules-in-use (Ostrom 2007). Based on the elements in the action arena, analysts can diagnose, explain, and predict the actors' patterns of interaction (i.e., aggregated individual choices, behaviours, and decisions of actors in the action arena) and the outcomes from these interactions (Rudd 2004; Di Gregorio et al. 2012). These patterns of interaction and outcomes are then assessed through a set of evaluation criteria. The criteria may differ based on the action arena; hence, analysts can determine how current institutional arrangements constrain or facilitate desirable outcomes depending on the specific actor or action situation selected, and how the action arena needs to be evaluated (McGinnis 2011; Ostrom 2011).

As stated in Chapter 2, this research used Cuevas et al.'s (2014, p. 2) definition of institutions under the CCA context, thus:

... institutions are the commonly known and acknowledged rules, social structures, and organisations founded on common belief systems that transform individual acts and expectations into collective actions, convert personal values into social norms and shared beliefs, and define the formal and informal behavioural systems of human existence. Rules, social structures, and organisations are all institutions.

	Institutional Concept			_		Advantages and Limitations in Relation to Mainstreaming CCA		
Framework	Rules	Social Structure	Organi- sation	Description	Focus of Analysis	Advantages	Limitations	
Advocacy Coalition Framework	V	V	\checkmark	 analyses problems involving significant goal conflicts, considerable technical disputes, and various actors from several layers of government (i.e., wicked problems) in the policy process offers a theoretical guide for understanding the intricacies of political conflict and mobilisation primary aim is to explain belief change and policy change over long periods of time 	Policy process	 views institutions as rules, social structures, and organisations provides a structured framework for policy analysis with: clear, causal assumptions; empirically testable hypotheses considers the significance of scientific and technical information in policy and political disputes applicable to various governing structures, cultural societies, and policy areas addresses the issue of institutional nestedness 	 can be time consuming, costly, and difficult to apply assumes a decade or more timescale in understanding political conflict and policy change may be ineffective in policy subsystems without clear coalitions or with just one dominant policy coalition needs further evidence and validation for application in CCA research 	
Institutional Analysis and Development Framework	V			 presents how rules, physical and material conditions, and attributes of the community affect the structure of action arenas, the incentives that individuals face, and the resulting outcomes helps analyse situations involving people interacting together in a particular context and following specific rules (i.e., institutional settings) 	Institutional setting	 effective in performing institutional analyses in diverse settings; across different systems; and institutions with intricate patterns of interactions and outcomes provides a structured approach in collecting data and a systematic manner of analysing institutional concerns (i.e., settings, dynamics, influences and interactions) flexible and practical; can be adjusted to suit the needs of the analysis enables analysts to identify the barriers to effective performance of policies addresses the issue of institutional nestedness 	 views institutions as rules only some specific features (i.e., evaluation criteria) are designed to analyse common pool resources, although the framework has been tested and successfully applied to a variety of institutional conditions and into an extensive range of problems and concerns needs further evidence and validation for application in CCA research 	

Table 2 Comparison of institutional frameworks

	Institutional Concept			_		Advantages and Limitations in Relation to Mainstreaming CCA		
Framework	Rules	Social Structure	Organi- sation	Description	Focus of Analysis	Advantages	Limitations	
Adaptive Capacity Wheel	V	V		 tool to assist researchers and policy makers understand, assess, and raise the ability of institutions in advancing the adaptive capacity of society consists of six dimensions, namely, variety, learning capacity, room for autonomous change, leadership, availability of resources, and fair governance encompassing 22 criteria 	Institutions	 provides both qualitative and quantitative analyses presents a comprehensive list of adaptive capacity criteria and indicators applicable to CCA research elements provide a comprehensive idea of the dimensions relevant to assess adaptive capacities simple and can be understood by non-experts focuses on analysis within institutions (internal) 	 views institutions as rules and social structures only provides limited institutional analysis because it does not provide a (an analytical) structure for examining the entire institutional setting cannot assess the varying (external) factors affecting institutions, or examine institutional linkages, relationships, and interactions does not address the concept of institutional nestedness and the issue of (institutional) scale 	
Adaptation, Institutions and Livelihoods Framework			V	 conceptual tool kit for examining institutional partnerships and impacts of these linkages on vulnerable social groups' access to resources examines institutional linkages among public, private, and civil society institutions centers on the importance of institutional partnerships in facilitating adaptation 	Institutions	 presents a structured framework for analysing adaptation practices and the critical role of institutions in the process presents an outline for analyzing adaptation practices through the analytical categories of adaptation responses focuses on the local scale 	 views institutions as organisations only limited capability to perform extensive institutional analysis with rules and social structure institutions 	

Sources: Jenkins-Smith & Sabatier (1994); Koontz (2006); Ostrom (2007, 2011); Sabatier & Weible (2007); Agrawal (2008); Ike (2009); Smajgl, Leitch, & Lynam (2009); Basurto et al. (2010); Gupta et al. (2010); McFadden, Priest, & Green (2010); Albright (2011); Dick & Meinzen-Dick (2011); Bettini et al. (2012); Getchell (2013).

Although the IAD views institutions as rules, the framework's design enables users to examine the social structure-based and organisational institutions, and accordingly map their linkages to one another. To illustrate, the action arena is influenced by a number of exogenous variables including the rules-in-use (i.e., rule-based institutions) and the attributes of the community defined as relevant aspects of the social and cultural context (i.e., social structure-based institutions). Organisational institutions, on the other hand, are incorporated as actors in the action arena (Figure 3).



Figure 3 Basic components of the IAD Source: Ostrom (2011, p. 10)

Furthermore, the framework is practical with a design that can be adjusted according to the needs of the problems being addressed (Rudd 2004). Generally, users change only the composition of the framework's main variables whenever IAD is modified, and still maintain its general premise in mapping institutional linkages and relationships. Any of the IAD components can be adjusted, and there have been a number of scholars, such as Rudd (2004), Di Gregorio et al. (2012), Jones et al. (2013), and Ratner et al. (2013), who modified the evaluation criteria. Following these examples, this research replaced the evaluation criteria of the IAD with factors that influence the effective operationalisation of mainstreaming CCA (i.e., mainstreaming challenges). This research deemed the evaluation criteria to be key variables as they guide the users in (1) assessing the patterns of interactions of institutions; (2) evaluating which outcomes are acceptable and which need improvement; (3) analysing how the current institutional arrangements constrain or facilitate desired outcomes; and (4) formulating ideas on how to attain the preferred outcomes (McGinnis 2011). Most importantly, this flexible characteristic of the IAD allowed the research to address the framework's limitation in examining institutions under the CCA context. That is, the concept of institutions-as rules, social structures, and organisations-was incorporated in the modified evaluation criteria of the IAD, transforming it into the IAD as applied in mainstreaming climate change adaptation or the IAD-CCA framework.

3.3.1 The IAD-CCA framework

In this research, around 80 peer reviewed papers and over 60 book chapters, conference papers, international agency reports, and discussion papers were reviewed to develop the evaluation criteria of the IAD-CCA. For the purposes of this chapter, this section summarises the results of this review. Detailed discussions of each challenge listed in the evaluation criteria are presented in Chapter 4.

The literature on the practical application of the mainstreaming approach and the barriers for its local operationalisation is limited (Uittenbroek et al. 2013; Ayers et al. 2014); thus, this research used a multi-topic strategy to generate a robust set of challenges. The challenges affecting adaptation (in general) and the mainstreaming approach (in particular) were explored both at the national and local scales. The review revealed that the drivers or enablers of adaptation are the opportunities for adaptation, while the barriers are the factors that impede adaptation. Both factors exist at the extremes of the same scale or spectrum (Amundsen et al. 2010; Gardner et al. 2010; Oberlack & Eisenack 2014). For example, the lack of, or the growing awareness of climate change are respectively considered as barriers to and drivers for adaptation. Similarly, effective leadership can help communities prevail over barriers, while the lack of it can be a barrier itself to adaptation (Tang et al. 2009; Moser & Ekstrom 2010; Jones et al. 2013).

The body of research on the linkages between CCA and institutions is increasing, with several authors emphasising the significance of developing the institutional capacity of systems to address climate change (Adger et al. 2005; Burch 2010). Similarly, several studies have identified that the serious barriers to adaptation are institutional in nature (Eisenack et al. 2014; Oberlack & Eisenack 2014; Waters et al. 2014). Using the definition of Cuevas et al. (2014) of institutions in the CCA context, the following barriers or challenges were identified (1) factors influenced by rule-based institutions such as autonomy of local governments, local government prioritisation, commitment to CCA, and other institutional issues relating to policies, regulations, and the like (Pini et al. 2007; OECD 2009; Burch 2010; Ayers et al. 2014; Waters et al. 2014); (2) matters linked to social structure-based institutions such as community support, institutional incentives and local leadership (Burch 2010; Moser & Ekstrom 2010; Biesbroek et al. 2011; Oberlack & Eisenack 2014); and (3) organisational concerns like organisational cohesion, and organisational cooperation and collaboration arrangements (Pini et al. 2007; Amundsen et al. 2010; Biesbroek et al. 2011; Eisenack et al. 2014).

The challenges related to climate change information also are key factors that affect CCA. These include the (1) extent of knowledge and awareness of climate change issues; (2) availability, accessibility, credibility, and reliability of information; (3) manner by which information is communicated and translated by climate change experts; and (4) way the information is received by the users (i.e., planners and decision-makers) (Ekstrom et al. 2011; Ayers et al. 2014; Oberlack & Eisenack 2014). Meanwhile, resource constraints have always been a problem for local governments; however, they are highlighted in CCA because resources are crucial factors of adaptive capacity (Pini et al. 2007; Biesbroek et al. 2011). For example, lack of funds is typically among the primary reasons why the implementation of local adaptation is delayed (Moser & Ekstrom 2010; Lehmann et al. 2015). Local governments have limited capabilities to invest or begin new endeavours since their budgets are often overextended. With the additional responsibility for CCA, these shortcomings are magnified; local governments become more under-resourced, overcommitted, and overtasked (Pini et al. 2007; OECD 2009). Hence, the availability of funds can be a great barrier to CCA when it is lacking, and a significant opportunity when it is sufficient.

This research summarises the mainstreaming challenges into three capacity classifications—*institutional, information,* and *resource* capacities. *Institutional capacity* pertains to the rules, social structures, and organisations involved in mainstreaming CCA. *Information capacity* deals with the ability of a system to integrate climate change information (i.e., technical and scientific knowledge and data) into the information system of the planning and decision-making processes. Finally, *resource capacity* focuses on the financial and human resources that ensure the maintenance and continuation of the integration process. The mainstreaming challenges under these capacity groupings are the factors that replaced the evaluation criteria of the IAD, thus, transforming the framework into the IAD-CCA (Figure 4 and Table 3).



Figure 4 Modified Institutional Analysis and Development framework as applied in mainstreaming climate change adaptation: IAD-CCA framework

3.3.2 Quantitative aspect of the mixed methodology

Converting the mainstreaming indicators into quantitative measures was another significant adjustment in the IAD-CCA framework. In general, quantitative analysis and generating metric tools are relevant research designs in order to understand and evaluate CCA performances. Accordingly, this need was magnified in this CCA mainstreaming research (Horrocks et al. 2012; Miller et al. 2012).

The characteristics of a good adaptation indicator were identified to inform the criteria in designing the mainstreaming indicators. These characteristics are (Harley & van Minnen 2009; Horrocks et al. 2012; Miller et al. 2012):

- simplicity simple to understand and can be communicated readily to users with different backgrounds;
- (2) status check able to track progress;
- (3) continuity and timeliness can be updated;
- (4) relevance significance and applicability at local scale;
- (5) clarity of purpose what, why, and for whom; and
- (6) comparability can be used to compare across sectors or time.

Mainstreaming Challenge	Description
Information canacity	Description
Availability of information	Level of available technical information on climate change
Access to information	Degree of user access to the available climate change information
Credibility and reliability of	Level of trust and confidence of users in the scientific and technical
information	information and their source/s
Communication of information	Quality of the active iterative and inclusive communication
communication of information	between climate experts planners and decision makers
Translation of information	Degree of usefulness of climate change knowledge provided by
Transtation of information	experts to planners
Knowledge and awareness	Degree of knowledge and awareness of planners and community
Miowieuge und awareness	members on climate change concerns and issues
Institutional capacity	nembers on enniae enange concerns and issues
Autonomy of local	Level of local government autonomy (i.e., administrative and
and an and a second sec	legislative powers and authority concerning (CA)
Leadershin (influence over	Absence or existence of a climate change "champion" in the
collective behaviour)	locality and the extent of the champion's influence on the behaviour
concentre benaviour)	of the community
Commitment to CCA	Absence or presence of an administrative and/or legislative
	framework for local adaptation
Community support	Degree of public support and local community participation on
community support	CCA initiatives
Organisational cohesion	Degree of coordination (or diversity) among organisations/agencies
	engaged in local CCA
Organisational cooperation	Absence or existence of cooperation and collaboration
and collaboration	arrangements among organisations concerning CCA
arrangements	
Local government	Level of CCA agenda within the general development priorities in
prioritisation	the local government
Institutional issues	Absence or presence of institutional questions or conflicts (i.e.,
	related to, a product of, or influenced by formal rule-based
	institutions) that inhibit the effective integration of CCA into local
	land-use planning
Institutional incentive	Degree by which the benefits from adaptation encourage actors to
	operationalise the mainstreaming approach
Resource capacity	
Availability of funds	Level of available financial resources to support CCA initiatives
Access to funds	Degree of user access to the available CCA funds
Stability of funds	Level of consistency of available CCA funds
Availability of experts	Existence of climate change experts to train local planners
Availability of human	Existence of people to undertake the CCA tasks and responsibilities
resources	

 Table 3 Components of the IAD-CCA evaluation criteria

Note: CCA - climate change adaptation

The scorecard approach was identified as the most suitable approach for converting the evaluation criteria in the IAD-CCA into quantitative indicators. The approach generates a numerical record of status and condition, which can measure the state-of-play and progress of activities in the setting being analysed (UNDP 2005; Bellamy & Hill 2010). Moreover, it (1) is easy to understand; (2) can be readily communicated to or interpreted by users with

varying backgrounds; (3) can be updated for timeliness and comparability across time; and (4) can be applied at the local scale (Frost 2007). The scorecard approach is also the technique that United Nations Development Programme (UNDP) uses to generate capacity development indicators. The method has been effective in quantifying the qualitative process of capacity development (UNDP/GEF 2003; UNDP 2005; UNDG 2008).

3.3.2.1 Quantifying the challenges: Scorecard approach and valuations

In designing the scorecard valuation, this research relied heavily on UNDP's procedures, particularly on the scorecard rating system used in the UNDP/Global Environment Facility (GEF) capacity development indicator methodology. In this methodology, each capacity result included a number of questions that represented a capacity indicator. The scorecards were in a form of descriptive sentences linked to each capacity development indicator with a numerical rating ranging from 0 to 4 (UNDP/GEF 2003; Bellamy & Hill 2010). The score for each question under a capacity category was averaged and the resulting value was considered as the overall rating (UNDG 2008).

The UNDP scorecard approach has been used in several studies. For example, Haanpää and Peltonen (2007) used it to analyse the institutional vulnerability in the Baltic Sea Region countries. In the survey questionnaire, the respondent chose a condition, among three statements, that best described the adaptive capacity of the nation. The statements were constructed in a progressive manner, from the basic (score = 1) to the highest level (score = 3). Thus, a higher score indicated a lower perceived national vulnerability to climate change impacts, and vice-versa (Haanpää & Peltonen 2007). The survey was distributed among the United Nations Framework Convention on Climate Change focal points, including the main personnel in key government agencies who were responsible for the environmental management and/or climate change issues in their respective countries.

In this research, the procedures described above were used in the scorecard valuation of the 20 mainstreaming indicators. Similar to the approach of Haanpaa and Peltonen (2007), the scorecard was incorporated in a survey conducted among the significant actors in the action arena, the local land-use planning system. First, each question, which represented a challenge in the IAD-CCA evaluation criteria, had three answer choices. The choices were descriptive sentences that exemplified a progressing status of the system's capacity to prevail over the challenges (i.e. worst condition = 1; moderate condition = 2; best condition = 3). A "Don't know" category was also included among the choices to avoid forcing the respondents to make a choice when they had no knowledge of the item (Table 4). Accordingly, the possible scores for each indicator were any value $1 \le n \le 3$. The closer the value of an indicator to 3, the more likely the system overcame or turned the challenge into an opportunity that helped mainstream CCA into the local land-use plans. Conversely, the closer the value of an indicator to 1, the more likely that the challenge was a barrier to mainstreaming; thus, it required intervention (See Appendix A2 for the complete survey questionnaire).

Description	Score
The climate change-related information, with specific focus on typhoons,	
are	
• not available.	1
• available, but limited or inadequate.	2
• are available and comprehensive.	3
The climate change-related information, with specific focus on typhoons,	
are available but/and	
• inaccessible.	1
• partially accessible.	2
• completely accessible.	3
A CCA champion (i.e., staunch advocate, promoter, implementer, of	
CCA initiatives) in the locality	
• does not exist.	1
• exists, but does not influence the behavior of the local community.	2
• exists and influences the behavior of the local community.	3
CCA is	
• not in the local government agenda because there are more important issues.	1
• in the local government agenda but underrepresented because there are more important issues.	2
• a priority local government agenda.	3
Institutional issues concerning land-use and land-use planning	
• exist and are affecting the adaptation approach.	1
• exist but are not affecting the adaptation approach.	2
• do not exist.	3
Other institutional issues	_
• exist and are affecting the adaptation approach.	1
• exist but are not affecting the adaptation approach.	2
• do not exist.	3
	Description The climate change-related information, with specific focus on typhoons, are • not available. • available, but limited or inadequate. • are available and comprehensive. The climate change-related information, with specific focus on typhoons, are available but/and • inaccessible. • partially accessible. • completely accessible. • completely accessible. • does not exist. • exists, but does not influence the behavior of the local community. • exists and influences the behavior of the local community. CCA is • not in the local government agenda because there are more important issues. • a priority local government agenda. Institutional issues concerning land-use and land-use planning • exist and are affecting the adaptation approach. • exist and are affecting the adaptation approach. • exist and are affecting the adaptation approach.

Table 4 Survey scorecard: Selected indicators for mainstreaming climate change adaptation into local land-use planning

Note: The "Don't know" choice is part of the survey questionnaire distributed to respondents in this study. CCA – climate change adaptation

3.3.2.2 Computing for the mainstreaming indicators

To verify and validate the survey results, in-depth interviews were conducted with the same pool of respondents. Equal weights were applied to each response in the survey in order to compute the final indicator score. Employing various weights usually entails a more complex series of assumptions and validations. However, this procedure was beyond what was required for an exploratory study such as the current investigation (Lebel et al. 2013).

Cronbach's alpha statistics—the most reported and used method to measure the reliability of estimates for indices—was computed to measure the reliability of indicator estimates. The alpha (α), which can have a value $0 \le \alpha \le 1$, gauges the survey's reliability by measuring the internal consistency of a test or scale items in the survey (Bravo & Potvin 1991; Santos, 1999). In general, the accepted value of α is between 0.70 to 0.95, with $\alpha > 0.9$ as excellent results; $\alpha > 0.8$, good; and $\alpha > 0.7$, as acceptable results (Gliem & Gliem 2003; Tavakol & Dennick 2011).

The utility of the IAD-CCA framework in examining the challenges in mainstreaming CCA, and the effectiveness of the mainstreaming indicators in assessing the state-of-play of the mainstreaming process were tested in a case study.

3.4 The case study approach

This research asked, "How can mainstreaming of climate change adaptation into local land-use planning be understood?" and "How can the challenges in the operationalisation of mainstreaming be overcome?" Furthermore, it aimed to determine the state-of-play among the challenges in mainstreaming CCA. Hence, this research called for a strategy to answer questions pertaining to the "how" and "why" of a current problem (i.e., mainstreaming CCA) under conditions where the researcher has minimal or no control (i.e., over local mainstreaming operationalisation). Among the research strategies (i.e., experiment, history, archival analysis, survey), the case study was best suited to accomplish these research tasks (Scholz & Tietje 2002; Yin 2014). The case study enables exploration and explanation of a certain phenomenon (i.e., CCA mainstreaming), and is appropriate for answering questions dealing with operational links in real-life contexts (i.e., local mainstreaming implementation) (Yin, 2009; Scholz & Tietje, 2002).

Most importantly, the case study method was suitable for testing the utility of the IAD-CCA framework. The case study described and illustrated how varying conditions, interactions, arrangements, and outcomes were examined through an institutional perspective. It also allowed for an intensive, detailed, and rich analysis, which enabled in-depth

assessment of the phenomenon being examined (Flyvbjerg 2011). The IAD-CCA framework provided the case study with a structured approach for data collection and a systematic approach for data analysis (McFadden, Priest, & Green, 2010; Bettini at al., 2012). Therefore, with the combination of the case study method and the IAD-CCA framework, the research was able to identify and describe the challenges in the local mainstreaming measure, and explain how and why the challenges affected the local implementation of this adaptation strategy.

In particular, this research warranted a multiple-scale investigation; that is, the issues in mainstreaming CCA into land use plans at the higher and lower scales or levels of governance had to be investigated. Accordingly, the research involved a single case (i.e., mainstreaming CCA into local land use planning) with more than one unit of analysis, or a Type 2 case study⁵ (Baxter & Jack 2008; Yin 2009). This Type 2 case study was achieved by choosing a critical case—one that has a strategic importance in relation to the general problem (Flyvbjerg 2011).

3.4.1 Case study area selection

Because the case needed a multiple-scale analysis, the selection involved choosing the country to conduct the case (national/federal), and the localities in this country to investigate (subnational/local scales). To be a critical case, the research should be in a country highly impacted by climate change, actively applying the mainstreaming approach to adapt to climate change, and operationalising the approach in land-use planning at both the national/federal and sub-national/local scales. A web search of the "most vulnerable countries to extreme weather events or climate change" was conducted to narrow the country selection choices. A list of the top 10 countries impacted by climate change was compiled, based on: the Global Climate Risk Index⁶ (generated by the Germanwatch (Kreft & Eckstein 2013; Kreft et al. 2014); the Climate Change Vulnerability Index⁷ developed by the Verisk Maplecroft (Maplecroft 2012, 2013, 2014); countries at most risk to storm compiled by the

⁵ Yin (2009) identified four types of case study design, namely: Type 1, single case, holistic; Type 2, single case, embedded; Type 3, multiple-case, holistic; and Type 4, multiple-case embedded. A single case study may have more than one unit of analysis, which is called the Type 2 design (single-case, embedded). According to Baxter and Jack (2008) a holistic case study with embedded units can only analyse one unique or critical case.

 ⁶ The Climate Risk Index indicates a level of exposure and vulnerability to extreme events that countries should understand as warning to be prepared for more frequent and/or more severe events in the future (Kreft et al. 2014, p. 3).
 ⁷ The new Climate Change Vulnerability Index (CCVI), released by global risks advisory firm Maplecroft,

⁷ The new Climate Change Vulnerability Index (CCVI), released by global risks advisory firm Maplecroft, enables organisations to identify areas of risk within their operations, supply chains and investments. It evaluates 42 social, economic and environmental factors to assess national vulnerabilities across three core areas.

World Bank (2008); and the most vulnerable countries to climate change based on the methodology developed by Wheeler (2011) of the Center for Global Development (Table 5).

The two countries commonly included in the lists were the Philippines (eight times) and Bangladesh (five times) (Table 5). Further investigation showed that both countries have been actively applying mainstreaming of CCA (Huq & Ayers 2008; Lasco et al. 2008, 2009; Mercado 2011; Pervin 2013; Ayers et al. 2014; Florano 2014). Likewise, both countries have CCA related legislation. Bangladesh has the Climate Change Trust Fund Act 2010, a policy that created the Bangladesh Climate Change Trust Fund for financing projects related to climate change (Pervin 2013), while the Philippines has the People's Survival Fund created in 2011 through the Republic Act No. 10174 (Republic of the Philippines 2011). Also, both countries have their own national plans for climate change—the Bangladesh Climate Change Strategy and Action Plan 2009 and the Philippine National Climate Change Action Plan 2011-2028.

However, the Philippines has a legislation that specifically mentions mainstreaming or integrating climate change in land-use planning—the Disaster Risk Reduction and Management Act of 2010 (Republic of the Philippines 2010). No similar legislation was identified during the web search for Bangladesh. Moreover, compared to the Bangladesh climate change plan, the Philippine plan has more specific content regarding land-use and land-use planning, such as (1) allocating the national government budget for direct and indirect CCA and mitigation in the land-use sector; and (2) implementing adaptive housing and land-use development. The Philippine plan also mentions integrating and harmonising CCA and DRR in the national and local agriculture and fisheries policies and plans, particularly targeting on lobbying "for congress to enact a national land use policy" (CCC 2011, p. 53). Given these findings, the research considered the Philippines to be a critical case in helping to understand local mainstreaming of CCA (Flyvbjerg 2011).

Rank	Climate	Risk Index (Ger	manwatch)	Climate Change Vulnerability Index (Maplecroft)		Countries at most risk to storm (World Bank)	Most vulnerable countries (Wheeler)	
	Long-term	2012	2012	2012	2014	2000	2009	2015
	(1994-2013)	2012	2015	2012	2014	2009	2008	2015
1	Honduras	Haiti	Philippines	Haiti	Bangladesh	Philippines	Djibouti	China
2	Myanmar	Philippines	Cambodia	Bangladesh	Guinea-Bissau	Bangladesh	Kenya	Djibouti
3	Haiti	Pakistan	India	Zimbabwe	Sierra Leone	Madagascar	China	India
4	Nicaragua	Madagascar	Mexico	Sierra Leone	Haiti	Vietnam	Mozambique	Kenya
5	Philippines	Fiji	St. Vincent and the Grenadines	Madagascar	South Sudan	Moldova	Ethiopia	Somalia
6	Bangladesh	Serbia	Pakistan	Cambodia	Nigeria	Mongolia	Sri Lanka	Mozambique
7	Vietnam	Samoa	Lao PDR	Mozambique	DR Congo	Haiti	India	Philippines
8	Dominican republic	Bosnia and Herzegovina	Vietnam	DR Congo	Cambodia	Samoa	Somalia	Bangladesh
9	Guatemala	Russia	Argentina	Malawi	Philippines	Tonga	Cuba	Sri Lanka
10	Pakistan	Nigeria	Mozambique	Philippines	Ethiopia	China	Philippines	Ethiopia

Table 5 List of the top 10 most vulnerable or at risk countries to climate change

Sources: World Bank 2009; Wheeler 2011; Maplecroft 2011, 2013; Kreft & Eckstein 2013; Kreft et al. 2014

3.4.1.1 National scale: Philippines

The Philippines is a developing country in Southeast Asia with a population of 93 million as of Census 2010 (PSA 2014). It is an archipelago with around 7,100 islands and is located within the Pacific typhoon belt. It experiences approximately 20 typhoons a year; thus, it is familiar with extreme weather events and weather-related hazards (Lasco et al. 2009; Yumul et al. 2011) (Figure 5). However, as compared to past experiences, the recent typhoons that have crossed the country have significantly intensified (CCC 2011; Cuevas 2012). For example, in November 2006, Typhoon Reming (international name: Durian) carried 466 millimeter (mm) of rainfall in 12 hours, the highest recorded in 40 years; the typhoon's maximum wind speed was recorded at 281 kilometers per hour. Likewise, in September 2009, Typhoon Ondoy (international name: Ketsana) produced 455 mm of rainfall within a 24-hour period, more than the average rainfall for that month, while the 24-hour rainfall delivered by Typhoon Sendong in 2011 (international name: Washi) exceeded the monthly average for December by 60% (Fano et al. 2007; NDRMMC 2012; PAGASA 2014).



Figure 5 Philippines, with the Albay case study area encircled Source: Nearmap database 2015

The Philippines also is the world's third most vulnerable country to extreme events, and among the countries with most people exposed to such extreme events (CCC 2010; World Bank 2013a). As a storm surge "hotspot", the country can potentially experience more than 1,000 deaths from a storm surge (World Bank 2006). Such a catastrophe became real when Typhoon Yolanda (international name: Haiyan) crossed the Philippines in November 2013. Dubbed as the most powerful typhoon to make landfall in recent history, Typhoon Yolanda created storm surges that affected 16 million people and resulted in 6,300 dead; 28,689, injured; and 1,061, missing. The damages from the typhoon were estimated at PHP 89.6 billion (USD 2.1 billion) (Lagmay 2014; NDRRMC 2014). Plans for rehabilitation and rebuilding from the impacts of Typhoon Yolanda are scheduled until 2016 (OPARR, 2014).

These intensified typhoons have been attributed to climate change. Unfortunately, this is projected to continue, thus, Typhoon Yolanda will not be the last powerful typhoon that the Philippines will experience (World Bank 2013b; Lagmay 2014). As such, as early as 2009, the Philippines enacted the Climate Change Act (Republic Act [RA] 9729), a law that institutionalised mainstreaming of climate change into the policy-making, planning, and other decision-making processes of the government. Similarly, the Philippines acknowledged the strong connection between CCA and disaster risk reduction (DRR), and enacted the Disaster Risk Reduction and Management Act of 2010 (RA 10121). This legislation mandated climate change and DRR to be mainstreamed into the development plans (including land-use) of both the national government and the local government units (LGUs). Accordingly, the Philippines' efforts to adapt to climate change are closely tied to the country's DRR initiatives (i.e., CCA-DRR) (CCC 2011).

However, mainstreaming CCA is a new initiative in the Philippines; its operationalisation is still at its early stages, and thus, is a work in progress. Consequently, the serious threat of climate change and the infancy of the mainstreaming efforts in the Philippines make it an ideal area to study.

3.4.1.2. Provincial scale: Albay, Philippines

The impacts of climate change are most experienced by local communities, hence, it is the level at which CCA measures are most important (Burch 2010; Hamin et al. 2014). In the Philippines, the local government units (LGUs) are divided into three major levels: the provinces, cities/municipalities (towns), and barangays (villages). The province is the largest of the three, and is composed of either municipalities, cities, or both. Meanwhile, a barangay is the smallest LGU that makes up the municipalities or cities. The Philippines has 81 provinces, 144 cities, 1,490 municipalities, and 42,029 barangays (NSCB 2014, 2015).

One of the provinces in the country actively pursuing CCA-DRR initiatives is Albay. Albay in 2010 had a population of 1.2 million living in its 2,554.06 square kilometer land area (NSO 2010; Espinas 2013). Poverty incidence among families in Albay in 2012 is 33.9%, about 14.2 percentage points higher than the national incidence, of 19.7% (NSCB 2013). Agriculture is the key economic sector in the province; coconut, rice, sugar and abaca are among the province's main products. Albay is located at the southernmost tip of the main island of Luzon, and is around 550 kilometers (km) from the country's capital, Manila. It is surrounded by the Pacific Ocean, Samar Sea, and Sibuyan Sea from the east, southeast, and southwest directions, respectively (Uy et al. 2011; Espinas 2013) (Figure 5).

Because of its geographical location, Albay is highly vulnerable to climate-related disasters. Accordingly, most of its municipalities (towns), located along its 364 km coastline, is exposed to storm surges during typhoons and other impacts of climate change extreme events (Manila Observatory 2005; Salceda & Rangasa 2011). Around 88 typhoons have crossed within the 50 km radius of Albay within the last 65 years; some of these typhoons were the strongest recorded in the country (PAGASA 2014). This includes typhoon Reming (International name: Durian) in 2006, which resulted in 1,023 deaths, several hundred missing people, and estimated damage of PHP 3.2 billion (USD 64.2 million) (Salceda & Rangasa, 2011). To adapt and build communities that are resilient to climate change, the Provincial Government of Albay has implemented some of the best CCA-DRR practices (i.e., local CCA-DRR policies, projects, and programs) in the country. The provincial government has been successful in its efforts and Albay's CCA endeavours have been recognised both nationally and internationally (Claudio 2012; UNISDR 2012a). In fact, the Albay case has been cited in the Intergovernmental Panel on Climate Change 5th Assessment Report to be among the leading CCA practices in the world (Mimura et al. 2014). Accordingly, the Albay experience offered a robust set of information concerning the challenges in mainstreaming CCA.

3.4.1.3 Municipal/City scale: Camalig Municipality and Legazpi City in Albay, Philippines

The local land-use plan, or the comprehensive land-use plan as commonly referred in the Philippines, is the primary document that contains the physical framework and fundamental basis of spatial development of an LGU. These plans are developed at the municipal and city levels (HLURB 2001). At present, Albay is comprised of 18 LGUs, specifically three cities and 15 municipalities (NSCB 2014). Due to time and financial constraints, two LGUs in Albay province were studied. These areas included an LGU with, and another without, a modified local land-use plan (i.e., with CCA and DRR components). The case study hypothesised that:

- (1) a LGU with a modified local land-use plan would provide information on the challenges it encountered while amending the land-use plan, the manner by which it addressed the challenges, and whether the LGU was able to overcome the challenges; and
- (2) a LGU without a modified land-use plan would provide information on the preliminary challenges in mainstreaming CCA-DRR into land-use plans, and would help identify the factors that currently affect the process of transforming the plan.

The LGU without the modified land-use plan was preselected based on the evidence that LGU personnel have undergone training on integrating CAA-DRR into the local land-use plan. The accessibility of LGU information (i.e., the LGU was visible on the internet and had a website) was also among the selection criteria.

The LGU that best matched the criteria was Camalig, which is a rural municipality with a population of 63,585 (as of 2010) spread across its 50 barangays (NSO 2010). Local land-use plans with CCA-DRR components included Daraga and Malinao municipalities and Legazpi City (Personal communication 2014). Selecting the second LGU was based on the following conditions:

- availability and accessibility of LGU information—as reflected in the visibility of the LGU in the internet, i.e., existence of an LGU website, and
- (2) ease of communication—refers to the condition of correspondence, i.e., whether the LGU responded to the researcher's attempts to communicate via electronic mail.

Pini et al. (2007) illustrated that urban and rural settings experience varying barriers to local natural resource management and environmental sustainability. The same was expected in relation to efforts to mainstream CCA and DRR. Thus, a robust set of data was expected when Camalig (rural area) was paired with a city LGU. Based on these conditions, Legazpi City (the provincial capital) was selected as the second LGU (Table 6). This proved to be a good choice, in terms of time and costs, since Legazpi City also hosts almost all the offices of the provincial respondents.

Table 6 Local scale area selection process

Criteria	Daraga	Malinao	Legazpi City
Availability and accessibility of LGU information	\checkmark	Х	\checkmark
Ease in communication	Х	Х	\checkmark
Urban area	Х	Х	\checkmark

3.4.2 Selection of survey and in-depth interview respondents

The survey conducted in the case study applied a purposive sampling technique to ensure that the respondents were knowledgeable about mainstreaming CCA into local landuse plans. It also employed snowball sampling or chain referral sampling for additional respondents. Accordingly, the survey was conducted among the key actors in the local landuse planning system in Albay and representatives of the national government, nongovernment agencies, and academic organisations who had experience in implementing projects concerning mainstreaming of CCA in local land-use plans.

The composition of the survey respondents has a design similar to the extended peer community—lay and technical stakeholders—facet of the post-normal science methodology. Under the post-normal context, the quality of the decision-making process depends on the open discourse between and among all those affected by the issue, and the extended peer community provides "extended facts" (e.g., personal or anecdotal experience, contextualised insights) regarding the concern at hand (Ravetz 1999; Saloranta 2001). Essentially, this line of thought advocates that "no particular expertise can deliver certainty for policy issues in the post-normal domain, and no expertise can claim a monopoly of wisdom and competence" on a complex issue (Petersen et al. 2011, p. 373).

In the Philippines, the local land-use plan or the comprehensive land use plan is developed at the municipal and city levels. Meanwhile, the plans are reviewed and approved at the provincial level (HLURB 2006). The Housing and Land Use Regulatory Board (HLURB), the primary institution tasked to provide land use plan guidelines, devised the 12-step process to comprehensive land use planning that is followed by all the LGUs in the Philippines. To limit the research coverage, the survey was designed to investigate the mainstreaming conditions under the step concerned with the actual preparation of the local land use plan—Step 7: Preparing the Land Use Plan (Figure 6).

The respondents at the city/municipal level included the members of the (1) Municipal/City Planning Development Offices that generate the plans; and (2) the Disaster Risk Reduction Office (Legazpi City) which is responsible for the CCA-DRR program of the LGU. At the provincial level, the respondents included members of the Provincial Land Use Committee (PLUC), who review and approve the plan. Specifically, the respondents included representatives from the Provincial Planning and Development Office, Housing and Land Use Regulatory Board Region V, Department of Interior and Local Government Region V, Department of Agrarian Reform Region V, Department of Environment and Natural Resources Region V, Albay Public Safety and Emergency, and the Provincial Agriculturist (Table 7).

Another set of respondents was selected based on their capability to provide insights with regard to the issues prevailing at the national scale (Figure 6). They were the:

- (1) key project personnel and national institutional representatives involved in activities and programs for mainstreaming CCA into the local land-use plans; and
- (2) experts on CCA and/or land-use planning from the academe and other institutions with local level experience.

Particularly, the respondents were representatives from the national offices of the Housing and Land Use Regulatory Board, Climate Change Commission, and the National Economic and Development Authority, and the institutions of the UN Human Settlement Programme, Philippine Institute of Environmental Planners, and the University of the Philippines Los Banos. These respondents were implementers of national climate change policies or CCA projects, or advisers of LGUs in the mainstreaming process. The results of the in-depth interviews revealed that this set of respondents had a unique perspective on the mainstreaming endeavour. While having a national level perspective on the mainstreaming concerns, the respondents also had knowledge of the issues existing in localities other than Albay. Therefore, they indirectly represented the assessments for other LGUs in the Philippines. With this, the research was able to generate data for two cases—Albay and LGUs other than Albay.


Figure 6 Survey and interview respondents

Table / Data concentrit seneme 1. Method, seale and coverage
--

	Municipal	Provincial	National	NGOs	Academe	Research
	government	government	government			
Survey	7	13	5	2	2	-
Interviews	5	11	3	0	2	-
Key informants	0	0	3	3	8	-
Documents	20	13	46	5	11	331

Snowball sampling or chain referral sampling was used to obtain additional survey respondents, who included representatives from the Regional Offices of the Environment and Management Bureau and the Mines and Geosciences Bureau at the provincial level, and the City Disaster Risk Reduction Management Council (Legazpi), Municipal Disaster Risk Reduction and Management Officer (Camalig), and the Department of Agrarian Reform (City Office) at the city/municipal level. The search for local respondents ceased when respondents started referring people who had already been surveyed/interviewed.

The same respondents as that of the survey served as the respondents of the interview. Particularly, five were interviewed among the nine survey respondents at the national scale; 11 among 13 respondents at the provincial scale; and five were interviewed among the seven survey respondents at the city/municipal scale (Table 7).

3.4.3 Survey, interview, and consultation set-up

The survey questionnaires were disseminated using two mediums and in two stages. The first stage involved sending the questionnaires via electronic mail in February 2014, where about 35% of the primary respondents replied. The second stage was conducted in the Philippines, in which hard copies of the survey forms were personally delivered to the primary and secondary respondents in April and May 2014.

The one-hour interviews were semi-structured and involved in-depth discussions of the challenges highlighted in the preliminary results of the online survey, namely, *institutional issues, institutional collaboration and cooperation arrangements, organisational cohesion,* and *availability of human resources.* The interview schedule was later adjusted in the field to focus more on *institutional issues* and *leadership*, the indicators deemed most significant based on the survey results and initial interviews conducted. Furthermore, questions were asked regarding the interviewee's perceptions of the following concerns (1) the significance of mainstreaming CCA into the local land use planning process; (2) the mind-set of the local planners regarding mainstreaming CCA into land use planning; and (3) the overall progress in mainstreaming CCA into the local land use plans in Albay (local scale) and the Philippines (national scale). The interviews were recorded and later transcribed.

Key informants from both the local and national governments were further consulted to clarify some concerns raised during the interviews. The informants were representatives of the institutions that played crucial roles in the issues mentioned by the interview respondents, such as the Department of Budget and Management, the Civil Service Commission, and the Climate Change Commission. Documents like government memoranda, national and local laws and regulations, handbooks, and guidelines, were also reviewed for supplemental information.

3.5 Data collection scheme

At this point, the individual stages of the mixed methodology have been discussed, from the modification of the IAD to the conduct of the surveys, interviews and consultations with key informants. This section integrates this information into the data collection scheme used in this research.

Climate change concerns were incorporated in the IAD-CCA framework through the exogenous variables that affect the local mainstreaming setting (action arena). Thus, the data gathered on the biophysical conditions, attributes of the community, and the rules-in-use enabled the identification of the elements that affected mainstreaming CCA at the local scale, which, by transitivity, also influenced the challenges encountered in operationalising the mainstreaming approach. The components of the action arena (i.e., actors and institutional arrangements) provided information on the state-of-play of the local mainstreaming activity, whereas the evaluation criteria explained the patterns of interactions and outcomes from operationalising the adaptation measure.

To gather the data outlined in the IAD-CCA framework, the research applied a strategy for collecting information from multiple data sources-the triangulation by data method (i.e., document, interview, survey, and key informant consultations) (Yin 2014). Through this scheme, the limitations of one data source were supplemented by the strengths of another, thereby enabling the collection of robust data and performance of extensive analyses on these data (Tables 7 and 8).

		Data	a methods	
Data requirements: IAD-CCA components			Consultations	
Data requirements. IAD-CCA components	Survey	Interviews	(key	Documents
			informants)	
Bio-physical conditions				
Climate change events				\checkmark
Land-use patterns				\checkmark
Physical characteristics				\checkmark
Community attributes				
Norms, practices, and traditions that				
influence decision-making		v		v
Administrative/political subdivisions		\checkmark		\checkmark
Economic activities				\checkmark
Demographic characteristics				\checkmark
Rules-in-use				
National laws and policies		\checkmark	\checkmark	\checkmark
Local laws and policies		\checkmark		\checkmark
Action arena				
Key actors in land-use planning		\checkmark	\checkmark	\checkmark
Key actors in climate change		\checkmark	\checkmark	\checkmark
Institutional arrangements, mandates, and		1		
responsibilities of key actors		\checkmark	\checkmark	✓
Patterns of interaction				
Relationships and linkages of key actors	\checkmark	\checkmark	\checkmark	\checkmark
Outcomes				
Mainstreaming of climate change				
adaptation into local land-use plan	\checkmark	\checkmark	\checkmark	\checkmark
conditions				
Evaluation criteria				
Mainstreaming indicators: identifying	/	,	,	,
barriers and opportunities	✓	✓	\checkmark	\checkmark
Mainstreaming indicators: analysis	\checkmark	\checkmark	\checkmark	\checkmark

Table 8 Data collection scheme 2: Requirements and methods

Note: The survey incorporates snowball approach and scorecard technique. Document pertains to desktop or document review.

3.6 Methodology issues and limitations

This section presents the issues encountered in developing the mixed-methodology, from modifying the IAD framework, designing the quantitative aspect of the methodology, to the actual application of the plans and methodological designs in the field. For example, the IAD framework views institutions as rules only (Ostrom 2007). However, the concept of institutions in CCA encompasses rules, social structures, and organisations (Cuevas et al. 2014). The IAD-CCA addressed this issue by incorporating the integrated institutional definition into the institutional capacity component of the evaluation criteria.

Due to the methodology's purposive sampling design—typically applied "to pick a small number of cases that will yield the most information about a particular phenomenon" (Teddlie and Yu, 2007, p. 83)—the survey and interviews included a small sample size. With the aim of gathering "greater depth of information from a smaller number of carefully

selected" people (Teddlie and Yu, 2007, p. 83), all LGU personnel who had intimate knowledge of mainstreaming CCA into the local land use plans were sought as respondents. To minimise the non-response in this highly specialised investigation, the questionnaires were disseminated in two stages—online survey and then field survey. With this strategy, the target people were included successfully among the survey respondents, allowing for in-depth investigation of the critical case of the research. According to Maxwell (2009), a systematically selected small sample (for typicality and relative homogeneity) can enable confident conclusions to be made in qualitative research.

Working with small sample sizes is generally an issue in performing statistical inference and hypothesis testing. However, the works of Bridge and Sawilowsky (1999), Janu'sonis (2009), Fitts (2010), Fritz et al. (2012), and de Winter (2013) showed that applying standard statistical analyses on small sample sizes can be feasible and valid. Likewise, the scorecard approach also has been applied effectively on a small sample size to create quantitative metrics (Haanpää and Peltonen, 2007). Still, to ensure the reliability of estimates, Cronbach's alpha statistics was computed on the data generated.

Other issues encountered pertained to the analysis of the mainstreaming indicators. For example, survey results implied that the questions for the availability of funds, experts, and human resources indicators should be modified. The computed indicator scores were not able to be taken at face value and the indicators required supplementary information and analysis to determine the "true" conditions surrounding the challenges. In particular, the questions for these indicators were concerned with the availability and regularity of resources. However, interviews revealed that the issue was not whether the resources were available, but whether they were sufficient to operationalise mainstreaming CCA efficiently.

While the four-stage mixed methodology was effectively applied in practice, it has some limitations. First, the methodology was restricted to investigating only the variables outlined in the IAD-CCA framework, and thus examined only the challenges identified in its evaluation criteria component. The IAD-CCA offers 20 mainstreaming indicators, and does not claim to exhaust all the possible challenges that systems may encounter in mainstreaming CCA. However, because of the IAD-CCA's flexible design, other researchers can adjust (i.e., augment or lessen) the evaluation criteria depending on their respective research needs.

Also, this mixed research methodology highly depended on the quantitative results. Essentially, if the indicator scores did not reflect the true conditions in the field, then the qualitative interviews would have been misdirected. Thus, it was crucial that the quantitative data collection instruments, especially the survey questionnaire, had the capability to gather data that best reflected the on-ground conditions.

3.7 Summary

The four-stage mixed methodology devised by this research is a systematic and practical process that can be used by analysts, planners, and decision-makers in setting the direction and prioritising activities in mainstreaming CCA. It has two key aspects, the modified IAD framework (i.e. IAD-CCA) and the case study. The IAD-CCA outlined the data requirements and analytical design of the research, whereas the case study method provided the structure, focus, and boundaries for the data collection activities. Thus, the IAD-CCA framework's theoretical underpinnings on the challenges in local mainstreaming were complemented by the case study.

This research recognised the significance of the IAD's flexible design, and replaced the IAD's evaluation criteria with the mainstreaming challenges in Stage 1. Twenty mainstreaming challenges were identified through document reviews and were spread across three capacity classifications, namely, institutional, information and resource capacities. Although this research presents a relatively extensive list of mainstreaming challenges, it does not claim to record all the possible challenges that systems may encounter in mainstreaming CCA. Accordingly, the evaluation criteria can be further modified depending on the needs of the research.

The scorecard approach was effective in converting these challenges into quantitative indicators in Stage 2; afterwards, the in-depth interviews verified and validated the indicator scores in Stages 3. Data gathered through triangulation by source (i.e., document reviews, key informant consultations, and in-depth interviews) were used for the qualitative analysis conducted in Stage 4, with the IAD-CCA framework employed as the analytical guide. In essence, the qualitative analysis provided deeper understanding of the issues highlighted in the survey, thereby strengthening the significance of the quantitative measures. This confirms the notion that qualitative analysis is equally important as the quantitative analysis; the former provides a detailed observation of the phenomenon investigated and presents supplemental understanding of the quantitative results (Yin 2014).

3.8 Conclusion

There is a growing interest in mainstreaming as an adaptation approach, and an increasing number of countries are encouraged to implement it (UNDP-UNEP 2011; World Bank 2013b; Ayers et al., 2014). However, in practice, mainstreaming encounters a number of difficulties, hence, it has been slow to operationalise (Uittenbroek et al. 2013; Lehmann et al. 2015). This research contends that this setback is likely caused by the lack of attention

given to the institutional facet of mainstreaming operationalisation. The current methodologies and tools for mainstreaming concentrate on climate-related concerns, and they neglect (1) the existing institutional settings and institutional arrangements into which CCA will be integrated; (2) the institutional transformations generated by the integration process; and (3) the impacts of the institutional changes caused by mainstreaming to certain realities already subjected to the existing institutions.

Thus, this research argues that mainstreaming operationalisation needs a methodology that focuses on the challenges in mainstreaming CCA and an analytical framework that can examine the mainstreaming process from an institutional perspective. To have intensive insights into these challenges, the methodology should be able to monitor and assess the severity of the challenges through metrics; answer causal linkages among challenges; and solve the questions pertaining to the "whys" and "hows" of the subject. Accordingly, this research devised the four-stage mixed methodology and applied it in examining the challenges in mainstreaming CCA into local land-use planning in Albay, Philippines.

The research attained its second objective by effectively applying the mixed methodology in practice. Although the methodology is at its initial stages of development, it generated results that have the potential to help analysts, planners, and decision-makers determine the state-of-play of the challenges in mainstreaming CCA and make informed decisions for overcoming these challenges. Thus, the mixed methodology can be a useful in advancing the operationalisation of mainstreaming CCA. This notion is supported by the succeeding chapters which illustrate how the research was able to identify and examine the primary barriers and other main challenges that hinder the operationalisation of the approach at the local scale.

The next chapter presents the analysis on the quantitative data generated by the fourstage mixed methodology. Specifically, it discusses the correlation analysis conducted to determine the linkages and interconnections between and among the challenges in mainstreaming CCA.

CHAPTER 4: THE INTERCONNECTED NATURE OF THE CHALLENGES IN MAINSTREAMING CLIMATE CHANGE ADAPTATION: EVIDENCE FROM LOCAL LAND-USE PLANNING*

*Cuevas, S.C. (Under review). The Interconnected Nature of Challenges in Mainstreaming Climate Change Adaptation: Evidence from Local Land-use Planning. *Climatic Change*.

4.1 Introduction

Climate change adaptation (CCA) has become a significant concern in the 21st century, and recent discussions on CCA have focused on its transition from conceptualisation to implementation. This shift has not been easy due to difficulties (i.e., barriers, obstacles, limitations, constraints, deficits, problems, or challenges to adaptation) that obstruct, delay, or deter the adaptation process, thereby undermining progress in CCA research, policy-making, and planning (Moser et al. 2008; Amundsen et al. 2010; Biesbroek et al. 2013; Clar et al. 2013; Eisenack et al. 2014). Hence, analysis of the barriers to adaptation has become important in adaptation research, which has begun to identify, categorise, and determine the characteristics of the barriers, why these barriers emerge, and how they can be overcome and turned into opportunities for adaptation (Burch 2010; Ekstrom et al. 2011; Eisenack et al. 2014; Waters et al. 2014; Lehmann et al. 2015). However, the complex nature of the barriers to CCA is not fully understood. For example, information is lacking on how to analyse systematically the interconnections and relationships among the challenges (Clar et al. 2013) and how to quantify the indicators that can help measure these linkages (Biesbroek et al. 2013). Accordingly, empirical evidence that demonstrates these interconnections and their impacts on the adaptation process is limited (Eisenack et al., 2014; Hamin et al. 2014).

Methodical planning is needed to address these knowledge gaps because the challenges or barriers to adaptation are context-specific (Biesbroek et al. 2013). Although it is possible to generalise the nature of these challenges, determining how they affect the adaptation process and the magnitude of these effects depends on the conditions existing in individual settings (Mimura et al. 2014). Hence, this research investigated barriers to CCA by utilising a case study in Albay, Philippines, and focused on the challenges in mainstreaming CCA into local land-use planning.

Understanding the nature of the challenges in mainstreaming, and the interconnections among these challenges, is important to ensure effective on-ground implementation (Biesbroek et al. 2013; Waters et al. 2014). This research addressed an often overlooked aspect of mainstreaming operationalisation—the institutional dimension of the approach. Essentially, this chapter is the first among the four chapter discussions, providing

evidence that the barriers to mainstreaming operationalisation are better understood by analysing them within the institutional context where CCA is to be mainstreamed.

The chapter begins with a review of the factors affecting adaptation, including mainstreaming (i.e., mainstreaming challenges) and the linkages among these factors. The results of the correlation analysis, the computed values of the mainstreaming indicators, and some qualitative assessments are then presented. This analysis concludes that in Albay, Philippines, mainstreaming challenges are interconnected, but to varying degrees. Moreover, the mainstreaming challenges that demonstrate the most interconnections (to other challenges) are *knowledge and awareness*, *local government prioritisation, institutional incentives, availability of funds, access to funds*, and *stability of funds*. Also, that a tripartite relationship can exist among mainstreaming challenges; in particular, there are strong linkages between *local leadership*, *local government prioritisation*, and the local government's *commitment to CCA*.

This chapter is the first of the three that address the third objective which is—*to* analyse the state-of-play of and linkages between the challenges in mainstreaming CCA into land-use planning in Albay, Philippines, and how to overcome these challenges. The focus of this chapter is on the nature of and interdependencies among the challenges to mainstreaming CCA based on the empirical evidence in the Albay case study. The intensive qualitative analyses of the results, including the in-depth interpretation of the mainstreaming indicator scores, are presented in Chapters 5, 6, and 7.

4.2 Identifying the challenges in climate change adaptation

The challenges or barriers to adaptation are the factors that hinder, impede, restrain, inhibit, limit, block, or delay planning, operationalisation, and overall progress of CCA measures and approaches (Moser et al. 2008; Amundsen et al. 2010; Burch 2010; Ekstrom et al. 2011; Ayers et al. 2014; Eisenack et al. 2014). In this chapter, the expressions "challenges" and "barriers" are used interchangeably; the two terms will be further qualified when the results of the research are revealed in later chapters.

Although there is some research analysing the mainstreaming of CCA, studies that examine its practical local application to planning are still lacking. Likewise, information concerning the barriers and limitations of its operationalisation is limited (OECD 2009; Measham et al. 2011; Biesbroek et al. 2014). Thus, to identify a robust set of mainstreaming challenges, a wide array of literature was reviewed, including studies that focused on the barriers, constraints, limitations, and issues in general and local CCA, and general and local mainstreaming of CCA. Accordingly, the document review of this literature suggested linkages between the barriers and the (low) ability of a system to respond to climate change impacts, i.e., adaptive capacity (Adger et al. 2005; Smit & Wandel 2006; Parry et al. 2007). To further comprehend these linkages, the existing literature on the enablers or determinants of adaptive capacity was examined (OECD 2006; Burch 2010; Gardner et al. 2010; Eisenack et al. 2014).

The document review revealed that the *drivers* or *enablers* of adaptation are the opportunities, while the *barriers* are the factors that impede adaptation; both factors exist at the extremes of the same scale or spectrum (Roberts 2008; Amundsen et al. 2010; Dovers & Hezri 2010; Gardner et al. 2010; Oberlack & Eisenack 2014). For example, the lack of, or the growing awareness to climate change are respectively considered as barriers to or drivers for adaptation. Similarly, effective leadership can help communities prevail over barriers, while the lack of it can be a barrier itself to adaptation (Tang et al. 2009; Roberts 2008; Moser & Ekstrom 2010; Jones et al. 2013).

In this research, mainstreaming challenges are the factors that affect or influence the effective operationalisation of the mainstreaming measure, either as barriers or opportunities. These mainstreaming challenges are summarised into three capacity classifications, namely, *institutional, information* and *resource capacities*. The 20 mainstreaming challenges listed as the evaluation criteria in the IAD-CCA framework were identified from this document review.

4.2.1 Institutional challenges

Planning for climate change is a challenge about "leadership, co-ordination, and collective action", thus it is about institutions (Evans & Stevens 2009, p. 2). The body of research on the linkages between CCA and institutions is increasing, with several authors emphasising the significance of developing the institutional capacity of systems to address climate change (Adger et al. 2005; Agrawal 2008; Amundsen et al. 2010; Burch 2010; Gupta et al. 2010). Similarly, several studies have identified that the serious barriers to adaptation are institutional in nature (Eisenack et al. 2014; Oberlack & Eisenack 2014; Waters et al. 2014). The next discussions present the various institutional challenges identified in literature, classified under the integrated institutional definition—rule-based, social structure-based, and organisational (Cuevas et al. 2014).

4.2.1.1 Rule-based institutions

The rule-based institutional challenges relate to how formal institutions affect the *commitment to CCA* of local governments, the *local government prioritisation of CCA*, *the*

autonomy of local governments to make decisions on the CCA, and other rule-based institutional issues. These challenges are discussed in detail in this section.

For instance, the absence of formal legislation that mandates actors to incorporate adaptation in their activities is a serious barrier to adaptation (Biesbroek et al. 2011; Hamin et al. 2014; Waters et al. 2014). This lack of formal laws creates an "institutional void" in the system—a condition in which the institutions that motivate, support, and facilitate CCA are scarce (Biesbroek et al. 2011, p. 186). On the other hand, the existence of formal laws and regulations promoting CCA facilitates the "institutionalisation of climate change response measures within standard operating procedures" of the local system (Burch 2010, p. 287). Thus, a local regime with a legislative structure dedicated to CCA reflects the local government's *commitment to CCA*; such commitment is an essential factor in effective adaptation (Pini et al. 2007; Tang et al. 2009; Duff 2011).

Still, the CCA agenda is difficult to progress when it competes with other local priorities that local governments argue are more pressing, such as poverty alleviation, access to clean water, access to primary education, and the likes (Measham et al. 2011). This situation often results in CCA concerns being set aside and given a low priority in the local agenda (Tang et al. 2009; Picketts et al. 2014). When adaptation is not a local priority, time, attention, and resources are diverted away from this endeavour (Agrawala & van Aalst 2006; OECD 2009; Biesbroek et al. 2011; Measham et al. 2011). This makes *local government prioritisation* of CCA a significant challenge. Similarly, the *autonomy of local governments* to formulate and implement their own choices concerning their jurisdictions is an important institutional factor in CCA (Pini et al. 2007). Local governments need to be empowered to make decisions (e.g., legislative, political, financial, etc.) and shape their own local plans and actions (OECD 2009; Measham et al. 2011).

A wide array of institutional concerns in different localities and settings influences CCA efforts. Scholars referred to these factors as institutional constraints (Measham et al., 2011) or institutional deficits (Oberlack & Eisenack 2014). In this research, they are known as *institutional issues*. For example, Burch (2010) identified the lack of coherence in climate change approaches in British Columbia, Canada as a considerable barrier to local climate change actions. Ayers et al. (2014, p. 48) cited that the "turbulent political system" in Bangladesh significantly constrains CCA planning in that country. Several studies have identified other barriers, such as mismatched timescales between climate change projections and development planning horizons, and specific sectoral issues (e.g., poorly defined property rights and land tenure schemes) (Agrawala & van Aalst 2006; OECD 2009).

4.2.1.2 Social structure-based institutions

This section discusses the social structures, such as attitudes, values, norms, practices, and beliefs, that influence how and why individuals and societies engage in CCA and adaptation planning (Biesbroek et al. 2011; Cuevas et al. 2014; Lehmann et al. 2015). It presents the challenges relating to *community support* to CCA, *institutional incentives* to engage in CCA, and the local *leadership* supporting CCA. Particularly, the *community's support* for adaptation and the manner by which actors are *incentivised* to adapt to climate change affect people's mindsets toward the endeavour. Likewise, the collective behaviour for climate change can be the result of a type of local *leadership*, specifically, the leadership of a local climate change champion.

The presence or absence of community support can substantially affect CCA. Strong community support defines the readiness of people to act and address climate change and suggests positive behaviour and attitudes toward CCA. Most importantly, it indicates the willingness of the community to accept and abide by the rules and regulations related to CCA (Burch 2010; Biesbroek et al. 2011; Waters et al. 2014). Meanwhile, *incentivising* planners, policy-makers, and other decision-makers to address CCA similarly influences behaviours toward the undertaking (Agrawala & van Aalst 2006; Oberlack & Eisenack 2014). As individual attitudes and the conduct of planners and decision-makers are crucial determinants of adaptation planning, instilling the benefits of CCA into their consciousness would encourage the creation of mindsets that are in favour of CCA (Mitchell et al. 2006; Gardner et al. 2010; Lehmann et al. 2015).

Another key component in local CCA is *leadership* or the "emergence of an identifiable political/administrative champion(s) for climate change issues" (Roberts 2008, p. 527). Leadership is a crucial aspect at any stage of the adaptation process. A climate change champion can raise awareness of climate change, initiate CCA efforts, put CCA high into the local government agenda, and uphold future CCA initiatives. Conversely, the absence of such a leader can weaken the climate change agenda and be a critical barrier to the endeavour (Burch 2010; Moser & Ekstrom 2010; Oberlack & Eisenack 2014; Waters et al. 2014).

4.2.1.3 Organisational institutions

The challenges related to organisational institutions primarily deal with the institutional arrangements between organisations that build cooperation and collaboration across scales, and the linkages among organisations that bring either cohesion or fragmentation among institutions.

The lack of *organisational cohesion* or the existence of organisational fragmentation is an incessant barrier, especially to an interdisciplinary, cross-sectoral and multi-level concern such as CCA (Amundsen et al. 2010; Eisenack et al. 2014). Lack of organisational cohesion can happen when CCA is isolated from the development agenda, which can accordingly manifest in the form of (1) unclear responsibilities or division of duties on CCA by varying organisations; (2) absence of organisational structure that determines the tasks of multiple departments concerning CCA; and (3) disorganised CCA efforts (Pini et al. 2007; Burch 2010; Waters et al. 2014). When responsibilities and tasks are vague, lack of organisational cohesion can lead to "institutional crowdedness," in which there is an overabundance of institutions with authority to influence CCA decisions (Biesbroek et al. 2011, p. 186).

The degree of cooperation and collaboration among organisational institutions is also linked to the lack of organisational cohesion. This factor is treated separately to emphasise the significance of institutional arrangements in structuring the rules governing interactions among individuals and societies. Likewise, *cooperation and collaboration arrangements* focus on the existence of mechanisms that facilitate these arrangements. Essentially, cooperative and collaborative arrangements can ensure compatible and complementary outcomes. On the other hand, ill-defined or conflicting arrangements between organisations can result in redundant, contradicting, or counterproductive CCA efforts (Duff 2011; Cuevas et al. 2014; Oberlack & Eisenack 2014). In fact, this particular concern about the lack of organisational cooperation, collaboration, and coordination "may be considered a main reason why adaptation mainstreaming is lacking" (Lehmann et al. 2015, p. 86).

4.2.2 Information challenges

The challenges related to climate change information are key factors that affect CCA. These include the extent of knowledge and awareness of climate change issues; the availability, accessibility, credibility, and reliability of information; the manner by which information is communicated and translated by climate change experts; and the way the information is received by the users (i.e., planners and decision-makers) (Mitchell et al. 2006; Mukheibir & Ziervogel 2007; Ekstrom et al. 2011; Ayers et al. 2014).

The level of awareness of climate change and its predicted impacts can shape the attitudes, behaviours, priorities, and the actions of local governments toward CCA, and this can result in powerful opportunities or barriers to adaptation (OECD 2009; Biesbroek et al. 2013; Oberlack & Eisenack 2014). To illustrate, sufficient *awareness and knowledge* of climate change impacts can urge local authorities to commit to CCA, and accordingly raise

adaptation high onto the local government agenda. Conversely, decision-makers and communities who refuse to recognise that climate change is real or have limited knowledge of climate change risks can remove CCA from the local agenda (Tang et al. 2009; Eisenack et al. 2014).

In the context of adaptation, scientists and local communities are parts of the same climate change information coin. The scientists generate and transmit the information, whereas the local communities receive and use the information. Hence, the challenges in information need to be perceived as the factors that influence the linkages between the giver and receiver of climate change information. For instance, the *availability of information* (or lack thereof) is a serious problem in adaptation. In retrospect, this factor also becomes a barrier when the available information does not match the needs of the users of information. For example, local planners and decision makers would need hazard maps with high resolution at local scales, but available data are often available only at national scales. Similarly, when the data are not customised for the site where CCA will be operationalised, there is a mismatch of availability and need. In this sense, the outputs of science and technology become irrelevant to CCA (Amundsen et al. 2010; Gardner et al. 2010; Waters et al. 2014).

Climate change data also become ineffective when users cannot *access the information*. This occurs when (1) users are not aware of the existence of the information; (2) users do not know how to retrieve the information; or (3) data are not for public use because they are generated and maintained by private entities (Gardner et al. 2010; Moser & Ekstrom 2010; Ekstrom et al. 2011; Nambi & Prabhakar 2011). Data should also be *credible and reliable* for users to apply the data to CCA. This factor incorporates the attitude and trust of the receiver in relation to the source of the information (Cash et al. 2003; Ekstrom et al. 2011; Measham et al. 2011). It determines the willingness of the receipients to use the data, thereby, leading to informed adaptation actions (Henry & Dietz 2011).

How climate change information is *communicated to users* affects how information is received, thus representing a substantial challenge in CCA. According to Cash et al. (2003, p. 8088), "active, iterative, and inclusive communication between experts and decision makers" is key in mobilising information into action. Thus, effective communication needs frequent and regular two-way dialogue between the source and the receivers (Cash et al. 2003; Mitchell et al. 2006). However, this approach should be accompanied by an efficient *translation of information*. Whereas the way information is communicated affects the attitude of recipients toward the information—specifically, their acceptance and willingness to use the information—the way information is translated influences its utility. That is, the way

complex ideas and scientific findings are explained would determine how users understand and utilise the information (Agrawala and van Aalst 2006; Biesbroek et al. 2011; Jones et al. 2013).

4.2.3 Resource challenges

Resource constraints have always been a problem for local governments. However, they are highlighted in CCA because resources are crucial factors of adaptive capacity (Pini et al. 2007; Biesbroek et al. 2011). For example, lack of funds is typically among the primary reasons why the implementation of local adaptation is delayed (Gardner et al. 2010; Moser & Ekstrom 2010; Lehmann et al. 2015). Local governments have limited capabilities to invest or begin new endeavours since their budgets often are overextended. With the additional responsibility for CCA, these shortcomings are magnified; local governments become more under-resourced, overcommitted, and overtasked (Pini et al. 2007; OECD 2009; Measham et al. 2011). Hence, the *availability of funds* can be a great barrier to CCA when it is lacking, and a significant opportunity when it is sufficient.

Along with the availability of funds, the *access to these funds* is crucial to CCA. Access would include (1) being aware that the funds exist; (2) the convenience in fund processing; and (3) having simple requirements to secure the fund (Biesbroek et al. 2011; Dang et al. 2014). A *stable and predictable fund source* is likewise significant, especially at the local scale (Pini et al. 2007; Kato 2014). Adaptation is an added concern to the local government that can overstretch their already strained financial capacities. Thus, the more likely that a local government will assume the financial responsibility of implementing CCA once funding ceases, the less likely it is that the local government will pursue CCA. Consequently, short-term remedies rather than long-term CCA approaches are implemented (Pini et al. 2007; Measham et al. 2011).

In CCA, the need for adequate resources extends beyond finance. The *availability of staff* dedicated solely to CCA is an important factor in local adaptation (Burch 2010). Because climate change is a long-term challenge, the stability of tenure of the human resources is vital—the permanency of the workforce can help ensure the continuity of CCA activities (Gardner et al. 2010). This issue becomes more complex because the requirements for such positions are specific—trained, skilled, and knowledgeable about climate change (Roberts 2008; Nambi & Prabhakar 2011; Ekstrom & Moser 2014; Oberlack & Eisenack 2014). Accordingly, investing in human resources becomes necessary.

The *availability of experts* to facilitate the training and education of CCA personnel is another issue (Mukheibir & Ziervogel 2007; Amundsen et al. 2010). Experts in this field are

scarce and thus, this problem hinders both the resource and information capacities of localities.

4.2.4 Linkages among the challenges in mainstreaming

As presented in the preceding discussions, there is an expanding literature on the barriers to adaptation, and a number of studies have focused on identifying these barriers. The task now is to go beyond recognising and classifying barriers, to focusing on the interconnections and interdependencies among them (Ekstrom et al. 2011; Eisenack et al. 2014). Understanding these linkages can help stakeholders to determine how these barriers can be overcome (Biesbroek et al. 2013).

The barriers to adaptation are interconnected or interlinked because "barriers do not exist in isolation, but are produced through interrelated processes" (Kuruppu & Willie 2015, p. 77). To illustrate, having unclear delineation of responsibilities among the institutions concerned with CCA affects the level of coordination among these organisations. This condition then results in poor organisational collaboration, which may further result in duplication of organisational efforts (Gardner et al. 2010). Hence, within the organisational context, a single issue can create a domino effect which debilitates the ability of organisations to address CCA concerns effectively. Meanwhile, because the barriers are interdependent, they either support or weaken each other (Eisenack et al. 2014). For example, Hamin et al. (2014) concluded that leadership was the major barrier to local climate change actions in the coastal cities and towns in Massachusetts, USA. This lack of leadership constrained the time and financial resources allocated to adaptation, which consequently restricted the planners' capacity to overcome the technical barriers.

Still, although it is recognised that barriers to adaptation are interconnected (Hamin et al. 2014; Lehmann et al. 2015), little is known about the extent of these relationships. This becomes problematic since "understanding the interdependencies of barriers is central for explaining their occurrence, persistence, and resolution" (Eisenack et al. 2014, p. 869). As such, generating quantifiable connections among these barriers would help researchers to develop a reliable basis for their qualitative assessments (Hamin et al. 2014). A quantitative approach can strengthen analysis and can help establish a baseline against which future actions and developments can be compared (Burch 2010; Lehmann et al. 2015); thus, the significance of the mixed methodology devised by this research.

4.3 Nature of the challenges in mainstreaming climate change adaptation into local land-use planning

This chapter presents only the quantitative analysis conducted in Stage 2 of the mixed-methodology. Specifically, it shows the results of the pair-wise correlation analysis conducted on the survey data and presents the computed mainstreaming indicator scores. A correlation analysis is a statistical technique that determines whether two variables are associated and measures the degree of this association. In general, it can be described as a study of interdependence (Asuero et al. 2006). A correlation coefficient (r) equals to 1 reflects a perfect positive relationship; -1 suggests the inverse; and r = 0 implies no relationship (Moutinho 2011).

It is acknowledged that the correlation analysis was conducted on a set of data collected from small sample sizes of the provincial (i.e., 13 respondents) and city/municipal surveys (i.e., seven respondents). The small sample sizes were due to the highly specialised nature of the investigation; respondents were limited to the members of the Provincial Land-use Committee for the provincial scale and to the city/municipal staff with knowledge of mainstreaming CCA into the local land-use plans. In general, local government units (LGUs) in the Philippines are understaffed, with people assuming multiple functions and positions (Corpuz 2012), and a one-person (staff) Municipal Planning Development Office is not uncommon (Interviews 2014). Still, all possible LGU personnel who had intimate knowledge of mainstreaming CCA into the local land-use plans were sought and were successfully included among the survey respondents.

Working with small sample sizes is usually a concern in conducting statistical inference and hypothesis testing. However, as mentioned in Chapter 3, Bridge and Sawilowsky (1999), Janu^{*}sonis (2009), Fitts (2010), Fritz et al. (2012), and de Winter (2013) agreed that applying standard statistical analyses on small sample sizes can be feasible and valid. In this research, the correlation analysis was used as a descriptive tool for examining the relationships among the mainstreaming challenges. The objective was not to generalise the results to a certain population, but to gather context-specific information. Thus, the results can be considered robust. Still, interpreting the correlation results should be done with caution because correlation only implies interconnections between variables and does not clarify causation (Moutinho 2011). Thus, on top of correlation analysis, additional information and supporting evidence are required for decision-making. This research addressed this through the in-depth interviews conducted in Stage 3, and the additional document reviews and consultations with key informants in Stage 4 of the mixed methodology.

The next discussions focus on the results of the correlation analysis and how they (correlation results) link to the mainstreaming indicator scores. This chapter focuses on the interconnections of the mainstreaming challenges, while the complete mainstreaming indicator scores together with their corresponding interpretations are discussed in Chapter 5. In particular, because correlation analysis only suggests interconnections among variables, this section explores the direction of the interconnections (i.e. causation) through the supplementary information provided by the mainstreaming indicators. The analyses are segmented by provincial and city/municipal scales.

4.3.1 Interconnections among mainstreaming challenges

The correlation analysis presented is restricted to the mainstreaming challenges with $r \ge 0.50$ (i.e., moderate correlation and more), highlighting those with $0.70 \le r \le 0.89$ (i.e., high correlations or strong associations) and $0.90 \le r \le 1.0$ (i.e., very high correlations or very strong associations) (Asuero et al. 2006). The frequencies of interconnections ($r \ge 0.50$) that each challenge has to other challenges also are presented.

The correlation analysis of the provincial data showed strong (i.e., $0.70 \le r \le 0.89$) and very strong ($0.90 \le r \le 1.0$) relationships (Table 9 and Figure 7) between and among resource capacity challenges (i.e., *availability of funds, access to funds, stability of funds, availability of experts*, and *availability of human resources*). Specifically, very strong associations ($0.90 \le r \le 1.0$) between the (1) *stability of funds* and *availability of funds*; (2) *stability of funds* and *access to funds*; (3) *availability of funds* and *access to funds*; (4) *availability of funds* and *availability of experts* were observed. Whereas resource-related challenges have been perennially identified to be important in every stage of a CCA initiative (Measham et al. 2011), and it is acknowledged that resource barriers are linked to one another, these results presented the extent of these linkages. For example, lack of funding oftentimes translates to lack of staff dedicated to CCA activities (Gardner et al. 2010). However, in the Albay case, the relationship between human resources and access to funds (r = 0.86) was stronger than the relationship between human resources and the availability of funds (r = 0.65).

At the city/municipal scale (Table 10 and Figure 8), very high associations were registered between (1) *organisational cooperation arrangements* and *availability of information*; (2) *organisational cohesion* and *access to information*; (3) *leadership* (existence of a climate change champion) and *local government prioritisation* (of CCA); (4) *institutional incentive* and *knowledge and awareness*; and (5) *availability of funds* and *access to funds*.

Meanwhile, at the provincial scale, the challenges that registered the highest frequencies of interconnections (i.e., $r \ge 0.50$) to other mainstreaming challenges included: *stability of funds*, 12 interconnections; *access to funds* and *institutional incentives*, 11 each; and *availability of funds*, *local government prioritisation*, and *knowledge and awareness*, with 10 interconnections each. The notable frequency of interconnections at the city/municipal scale was posted by the *knowledge and awareness* challenge, at 9 interconnections (Figures 7 and 8).

Lastly, at the provincial scale, there were strong tripartite relationships among (1) *knowledge and awareness, stability of funds*, and *access to funds*; (2) *local government prioritisation, availability of experts*, and *access to funds*; (3) *institutional incentive, availability of experts*, and *stability of funds*; and (4) *institutional incentive, availability of experts*, and *access to funds*; (1) *leadership, local government prioritisation* and *commitment to CCA*; (2) *leadership, local government prioritisation*, availability of funds and access to funds; and (4) *availability of funds, access to funds*, and *availability of funds*, and *availability of funds*, and *access to funds*, and *availability of funds* and *access to funds*; and (4) *availability of funds, access to funds*, and *availability of human resources* (Figure 8). The tripartite relationship indicated that a change in the status of one challenge could affect the statuses of the other two challenges, and vice-versa. This notion was confirmed and further explored in the discussion on the mainstreaming indicators.

Mainstreaming challenges	Correlation coefficient	Mainstreaming challenges	
Translation of information	0.7526*	Credibility & reliability of information	
	0.7395*	Credibility & reliability of information	
	0.7449*	Knowledge & awareness	
Stability of funds	0.8470*	Institutional incentive	
	0.9067*	Availability of funds	
	0.8989*	Access to funds	
Local government prioritisation	0.7245*	Community support	
	0.7454*	Local government prioritisation	
Availability of avaarta	0.7652*	Institutional incentive	
Availability of experts	0.8989*	Access to funds	
	0.7059*	Stability of funds	
Institutional incentive	0.7762*	Translation of information	
	0.7094*	Knowledge & awareness	
A googs to funds	0.7454*	Local gov't prioritisation	
Access to funus	0.8148*	Institutional incentive	
	0.8989*	Availability of funds	
	0.8600*	Access to funds	
Availability of human resources	0.7126*	Stability of funds	
	0.8898*	Availability of experts	

Table 9 Mainstreaming indicators with correlation ≥ 0.70 , provincial scale

*Specifies the significance level of correlation coefficients at the 5% level or better (i.e., 95% confidence level).



Figure 7 Interconnections among the mainstreaming challenges, provincial data

Notes: The broken lines depict strong relationships between mainstreaming challenges ($0.70 \le r \le 0.89$). Solid lines illustrate very strong linkage ($0.90 \le r \le 1.0$). The number near the circle depicts the frequency of linkages the mainstreaming challenge has with others at $r \ge 0.50$.

Mainstreaming challenges	Correlation coefficient	Mainstreaming challenges
Translation of information	0.7500	Communication of information
A googg to funds	-0.7500	Translation of information
Access to runus	1.0000*	Availability of funds
Commitment to climate change adaptation	0.7500	Leadership
	0.7303	Communication of information
Availability of experts	0.7102	Knowledge & awareness
Availability of experts	0.7303	Leadership
	0.7303	Local government prioritisation
Organisational cohesion	1.0000*	Access to information
Organisational cooperation arrangements	1.0000*	Availability of information
Local goy't prioritisation	1.0000*	Leadership
	0.7500	Commitment to climate change adaptation
Institutional incentive	0.9169*	Knowledge & awareness
Availability of funds	-0.7500	Translation of information
	0.7500	Availability of funds
Availability of human resources	0.7500	Access to funds
	0.7303	Stability of funds

*Specifies the significance level of correlation coefficients at the 5% level or better (i.e., 95% confidence level). Note: CCA – climate change adaptation



Figure 8 Interconnections among the mainstreaming challenges, city/municipal data

Notes: The broken lines depict strong relationships between mainstreaming challenges ($0.70 \le r \le 0.89$). Solid lines illustrate very strong linkage ($0.90 \le r \le 1.0$). The number near the circle depicts the frequency of linkages the mainstreaming challenge has with others at $r \ge 0.50$.

These sets of information can guide planners and decision-makers to:

- investigate whether addressing the status of the *access to funds* may be more important than changing the status of the *availability of funds*, in relation to addressing the challenge on *the availability of human resources*;
- (2) explore the following challenges in designing a strategy with potential extensive effects on the mainstreaming process: *stability of funds*, *access to funds*, *institutional incentives*, *availability of funds*, *local government prioritisation*, and *knowledge and awareness*;
- (3) investigate whether developing *resource capacity* and *knowledge and awareness* simultaneously can be an optimal strategy to an effective mainstreaming process;
- (4) consider the very strong association between local *leadership* and *local government prioritisation* of CCA in devising viable approaches to mainstreaming CCA; and

(5) exploit the tripartite relationships of the mainstreaming challenges (i.e., *local leadership-local government prioritisation of CCA-commitment to CCA*) (discussed in 4.3.2).

4.3.2 Tripartite relationship among leadership, local government prioritisation, and commitment to CCA

This section focuses on the tripartite relationship among the challenges pertaining to *leadership, local government prioritisation,* and *commitment to CCA*, as illustrated by the correlation results; in particular, how this relationship relates to the mainstreaming indicator scores. Accordingly, the mainstreaming indicators are briefly introduced, while detailed discussions of the indicators are presented in Chapters 5, 6, and 7.

Cronbach's alpha (α) statistics for the local (i.e., provincial and city/municipal) data set was computed at 0.8925; α for the provincial subset registered at 0.9487 and 0.9001 for the city/municipal subset. These results suggest reliable estimates for the mainstreaming indicators, provided that the accepted value of $0.70 \le \alpha \le 0.95$ (Gliem & Gliem 2003; Tavakol & Dennick 2011).

Details from the interviews confirmed the notion that barriers should be prioritised differently as they have varying degrees of severity (Biesbroek et al. 2013; Waters et al. 2014). Essentially, the indicator scores presented the state-of-play related to the mainstreaming challenges at each governance scale (i.e., national, provincial, and municipal/city). The interviewees explained the reasons behind the scores and based on the interview assessments, the mainstreaming indicators were classified into four levels that illustrate the transition of the challenges from barriers to opportunities for mainstreaming. The first-level mainstreaming indicators (i.e., with scores $1.0 \le n < 2.0$) represented the primary barriers that constrained the effective integration of CCA into the local planning system. Conversely, the fourth-level mainstreaming indicators (i.e., with scores ≥ 2.5) indicated those challenges that the system has already overcome, have positive effects on the implementation process, and, therefore, are considered as opportunities for mainstreaming. Indicators with scores $2.0 \leq n < 2.25$ and $2.25 \leq n < 2.5$ were categorised as second- and third-level barriers, respectively. In essence, the indicator scores identified the problems in mainstreaming by level of significance, and hence can help planners and decision-makers prioritise activities.

Analysis of the mainstreaming indicator scores supported the tripartite relationship among *leadership*, *local government prioritisation*, and *commitment to CCA* challenges. *Leadership* pertains to the absence or existence of a climate change champion in the locality, and the extent of the champion's influence on the behaviour of the community. Meanwhile, *local government prioritisation* refers to the level of CCA agenda within the general development priorities in the local government. Finally, *commitment to CCA* pertains to the absence or presence of an administrative and/or legislative framework for local adaptation. The local *leadership* indicator (score: provincial=2.67; city/municipal=2.57) was assessed as an opportunity for mainstreaming CCA due to the existence of a climate change champion—Governor Jose Salceda, the chief executive officer of the province (Figure 9). Governor Salceda was an effective champion, hence the United Nations International Strategy for Disaster Reduction acknowledged him as a "Senior Champion" of CCA and DRR in 2010, as well as a "Champion for Making Cities Resilient" (PreventionWeb 2010; UNISDR 2012b).



Figure 9 Mainstreaming indicators linked to local leadership: Albay, Philippines Note: Grey area represents the region where challenges have been transformed into opportunities for mainstreaming CCA.

Under the governor's leadership, the Provincial Government of Albay institutionalised CCA as a priority agenda by virtue of Provincial Resolutions 2007-04 and 2007-24. Both local regulations aimed to influence the political and social consciousness of the people in the government, private sector, and communities in Albay with regard to CCA (PGA 2007; Lasco et al. 2008). Such policies, along with many others, established a legislative framework for CCA in the province. Consequently, the local government prioritisation indicator (score: provincial=2.77; city/municipal=2.57) was evaluated as an opportunity at both provincial and city/municipal scales. However, the commitment to CCA indicator (score: provincial=2.71; city/municipal=2.43) was an opportunity only at the provincial scale; it was considered as a third-level barrier at the city/municipal scale (Figure 9). These assessments imply that although commitment to CCA is not a serious constraint at

the city/municipal scale, more work is needed to completely overcome the challenge and transform it into an opportunity for mainstreaming. Hence, the implication of a climate change champion at the city/municipal scale can then be explored.

The very strong interconnection between *leadership* and *local government prioritisation*, as suggested by the correlation analysis, was corroborated by the assessments of the respondents, such as:

There are a lot of traditional politicians who are more focused on projects related to poverty, social services, infrastructure, and the like, because the votes are there. Climate change looks into the future and not the present; therefore, climate change would not be in most politicians' agenda and priorities. Instead, they would have short-term insights rather than long-term plans for future climate change impacts....Hence, the importance of a climate change (in the province). If climate change is not among the priorities of the governor, only a small number of people will respond to the climate change efforts.

With regard to the relationship between *leadership* and *commitment to CCA*, one respondent stated:

There should be a (climate change) champion, and this should be the local chief executive. Likewise, this champion should be supported by the local legislative body. Essentially, the local government functionaries—like the heads of departments and technical staff—will not be very effective if these two bodies (executive and legislative bodies of local governments) are not synergised in supporting CCA.

4.3.3 Relationship between knowledge and awareness and institutional incentive

This section explores the direction of the very strong relationship (r=0.92, city/municipal scale) between *knowledge and awareness* and the *institutional incentive* indicators. The *knowledge and awareness* indicator refers to the level of understanding of planners, decision makers, and the communities regarding the implications of climate change (knowledge), or their acknowledgement of the existence of climate change (awareness). Meanwhile, *institutional incentive* measures the degree by which the benefits from adaptation encourage actors to operationalise the mainstreaming approach.

In Albay, the knowledge and awareness indicator (score: provincial=2.70; city/municipal=2.69) was an opportunity (Figure 9). Most of the respondents attributed this condition to the climate change champion, specifically, "because Governor Salceda has been actively advocating for CCA-DRR, hence, the people in Albay, including the planners and the Municipal Planning and Development Council, are aware of climate change" (Interviews 2014). Some of the initiatives that contributed to this high level of knowledge and awareness

included institutionalising organisations that can advance the knowledge of planners, decision-makers and the community on climate change (i.e., Center for Initiatives and Research for Climate Adaptation [CIRCA] and Climate Change Academy). CIRCA was created through the Provincial Executive Order 2007-12-A and became Albay's main organisation for CCA and climate risk reduction research (Lasco et al. n.d.). Meanwhile, the Climate Change Academy, institutionalised through a Memorandum of Agreement between the Provincial Government of Albay and the province's leading university (i.e., Bicol University), facilitated the education of local stakeholders on both climate change and disaster risk concerns (PGA and CIRCA 2010). The Climate Change Academy had since conducted seminars, workshops, and trainings on climate change, CCA, and DRR (Interviews 2014).

The institutional incentive indicator (score: provincial=2.81; city/municipal=2.86) was likewise evaluated as an opportunity in Albay (Figure 9). Most respondents rated mainstreaming of CCA into the local land-use planning process as very important, and credited this mind-set to the improved knowledge and awareness of planners on climate change issues. As such, the respondents stated:

In my opinion, about 80% of the planners in Albay have the mind-set that integrating CCA-DRR into the local land-use plans should be prioritised.

As planners become aware of climate change concepts, impacts, etc., they are encouraged to include climate change mitigation and adaptation into their local plans.

Local government units (staff) have attended trainings and seminars on CCA to improve their understanding of climate change, including their (staff's) appreciation of CCA. So, I think they know the significance of integrating CCA into the local land-use plans.

My technical staff has been attending climate change seminars, workshops, and trainings, and they highly appreciate the importance of integrating CCA-DRR into the local land-use plans. (Interviews 2014)

Whereas the correlation results showed strong interconnection between *knowledge and awareness* and *institutional incentives*, the testaments of the interviewees illustrated the direction of the relationship. That is, high *knowledge and awareness* of climate change was the cause, and a high level of appreciation of the benefits of CCA (i.e., *institutional incentive*) was the effect.

4.4 Summary and conclusions

Understanding the challenges for adaptation is significant for the successful onground application of the mainstreaming approach (Amundsen et al. 2010; Biesbroek et al. 2013; Clar et al. 2013; Eisenack et al. 2014). Mainstreaming challenges are the factors that affect or influence the effective operationalisation of the mainstreaming measure, and can be represented either as barriers or opportunities. Operational advice on mainstreaming CCA typically concentrates on climate-related issues, and recommended tools and techniques to address these issues include vulnerability assessments, climate risk screening, and climate change scenario building, among others (Olhoff & Schaer 2010; Lebel et al. 2012; SPREP & UNDP 2013). The institutional dimension of applying the approach in practice is often overlooked.

This chapter contends that analysing the challenges in mainstreaming within the institutional context can help planners, analysts, and practitioners to improve their understanding of the nature of the challenges. This chapter answered the third research objective—to analyse the state-of-play of and linkages between the challenges in mainstreaming CCA into land-use planning in Albay, Philippines, and how to overcome these *challenges*—by confirming that the barriers to adaptation were interconnected or interlinked, and that these relationships existed at varying degrees of intensity. Furthermore, the challenges had varying frequencies of interconnections; some challenges were more interconnected than others. To illustrate, the challenge relating to knowledge and awareness was evaluated among those with the highest frequencies of interconnections with other challenges. Moreover, this challenge had a strong association with *institutional incentive*. Qualitative assessments verified these relationships and showed that the improved *knowledge* and awareness incentivised the Albay planners to apply the mainstreaming approach in local land-use planning. Another notable result was the tripartite relationship among *leadership*, local government prioritisation and commitment to CCA challenges. Assessments showed that the existence of a climate change champion in Albay resulted in the Provincial Government of Albay prioritising CCA and in creating a provincial government committed to CCA.

This chapter demonstrated how quantitative (i.e., correlation analysis, mainstreaming indicators) and qualitative (i.e., indicator score assessments) techniques, methods, and analyses can be used in formulating strategies for an efficient mainstreaming process. In particular, it showed how the mainstreaming indicators were effective in providing supplementary information to the correlation analysis. That is, the indicator scores confirmed the relationships implied by the correlation coefficients, and also clarified the direction (i.e.,

causation) of these relationships. However, the utility of the mainstreaming indicators go beyond providing supplementary evidence on the relationships among the mainstreaming challenges, a notion that is explained in the next chapter.

CHAPTER 5: CHALLENGES IN MAINSTREAMING CLIMATE CHANGE ADAPTATION INTO LOCAL LAND-USE PLANNING: EVIDENCE FROM ALBAY, PHILIPPINES*

*Cuevas, S.C., Peterson, A., Robinson, C. & Morrison, T. H. (2015). Challenges in Mainstreaming Climate Change Adaptation into Local Land-use Planning: Evidence from Albay, Philippines. *The International Journal of Climate Change: Impacts and Responses*, 7(3):45–65.

5.1 Introduction

The previous chapters argued that while research interest in mainstreaming is growing, information remains limited, especially on the practical application of the approach at the local scale, and on the barriers to the effective operationalisation (Measham et al. 2011; Mangoyana et al. 2012; Uittenbroek et al. 2013; Mimura et al. 2014). This knowledge gap is exacerbated further by the absence of metrics to measure the extent of the barriers' impacts on achieving adaptation outcomes. Adaptation indicators that can track the process, implementation, scope, and the degree to which targeted outcomes are attained would be useful for planning and policy-learning (Noble et al. 2014). This chapter addresses this setback, and analyses the quantitative mainstreaming indicators developed in the research. Specifically, these indicators illustrate how the challenges in adaptation affect the mainstreaming endeavour.

Critical issues in adaptation research pertain to the barriers or challenges obstructing, delaying, diverting or blocking the adaptation process (Ekstrom et al. 2011). Previous research identified the pressing queries regarding the barriers to mainstreaming CCA, and CCA in general, as (1) What are the barriers to adaptation?; (2) How are the barriers categorised?; (3) What is the nature of the barriers?; (4) Why and how do barriers appear and persist?; and (5) How can the barriers be overcome? (Amundsen et al. 2010; Farrell 2010; Moser & Ekstrom 2010; Roberts 2010; Sharma & Tomar 2010; Eisenack et al. 2014). Still, more information is needed to fully understand the complexities in overcoming these barriers and effectively operationalising an adaptation measure.

The mainstreaming indicators introduced in this chapter address the first three queries, while the last two questions are answered in Chapters 6 and 7, respectively. In particular, this chapter argues that understanding the nature of the barriers is crucial in operationalising and analysing the mainstreaming process (Biesbroek et al. 2013; Waters et al. 2014). Thus, quantitative indicators that can assess the state-of-play of a mainstreaming endeavour are significant tools in formulating effective mainstreaming strategies.

This chapter supports Chapter 4 and continues to address the third objective—to analyse the state-of-play of and linkages between the challenges in mainstreaming CCA into land-use *planning in Albay, Philippines, and how to overcome these challenges.* The chapter begins with a review of how the mainstreaming indicators used in this research were generated. These indicators are the quantitative representation of the factors that affect the effective operationalisation of mainstreaming, which are collectively referred to in the previous chapters as "mainstreaming challenges". The characteristics of the challenges are then analysed leading to the following conclusions:

- (1) mainstreaming challenges exist within a certain spectrum, with the barriers and opportunities for adaptation representing the extreme ends of this spectrum; and
- (2) the barriers can affect the mainstreaming process at varying degrees of severity.

The above set of information can help local governments to prioritise the challenges that need to be addressed (i.e., primary barriers) and determine which factors can be utilised to help in the mainstreaming process (i.e., opportunities). Additional key findings include:

- (1) one mainstreaming challenge can be either a key barrier or opportunity to another mainstreaming challenge; and
- (2) when the barriers are overcome, they can become opportunities for mainstreaming CCA.

Based on these results, this chapter concludes that by knowing the relationships among the mainstreaming challenges, planners and decision makers can formulate strategies that can have maximum impacts on mainstreaming CCA.

5.2 Generating the mainstreaming indicators

In Chapter 3, the modified Institutional Analysis and Development (IAD) framework (IAD-CCA) was introduced as the research's primary analytical tool. The IAD is a "multi-tier conceptual map" that examines institutional settings or those situations that involve people interacting together in a particular context (Ostrom 2011, p. 9). It was deemed the most suitable framework for the research since mainstreaming CCA is essentially an institutional concern and should be analysed under an institutional perspective (Agrawala & van Aalst 2006; Ayers & Dodman 2010).

The IAD-CCA evaluation criteria are composed of factors that reflect the challenges in mainstreaming CCA. The evaluation criteria are significant variables in the framework as they are used as guides to identify the patterns of interactions generated in the institutional setting under analysis (i.e. local land-use planning). Moreover, the criteria determine which outcomes are good or poor. The evaluation criteria used in this research's IAD-CCA comprised 20 mainstreaming challenges, classified under the information, institutional, and resource capacity groupings; these challenges were used as the basis for designing the survey questionnaire. The survey focused on the respondents' assessment of the local mainstreaming progress, and scorecards were applied to quantify the participants' responses. In particular, each survey question had three answer choices that illustrated a possible condition surrounding the mainstreaming challenge. The described conditions improved at each subsequent answer choice. The worst condition was given a score of 1 and the best possible state was assigned a score of 3. Accordingly, each mainstreaming challenge was converted into a quantitative mainstreaming indicator with a value ranging from $1 \le n \le 3$. The indicators assessed the state-of-play of the mainstreaming process and evaluated whether a particular indicator was a barrier or an opportunity. Equal weights were applied to the answers supplied by the survey respondents.

Using the statistical software STATA, Cronbach's alpha statistics—a popular method to measure the reliability of estimates for indices—was computed. Alpha (α), expressed as a number between 0 and 1, measures the internal consistency of a test or scale items in a survey to gauge the survey's reliability (Bravo and Potvin 1991; Santos 1999; Gliem and Gliem 2003; Tavakol and Dennick 2011).

The survey results determined the direction of the subsequent semi-structured interviews on the same set of respondents. The interviews focused on the mainstreaming indicators that scored closest to either 1 or 3, and aimed to confirm and further explain the mainstreaming indicator scores. Additional information to support the survey results were acquired by consulting with key informants and reviewing documents such as national and local laws and regulations, government memoranda, LGU reports, and other related studies.

5.3 Challenges mainstreaming CCA-DRR into land-use planning

This section presents the results of the mainstreaming indicator scores; the interpretation of these scores; and the core on-ground conditions relating to the scores. It describes how the mainstreaming indicators illustrated the state-of-play of the mainstreaming process in Albay, Philippines.

5.3.1 Survey results

The Cronbach's alpha statistics was computed to estimate the reliability of the survey instrument used to generate the indicator estimates. In general, the accepted value of alpha (α) ranges from 0.70 to 0.95 (Tavakol and Dennick 2011); specifically, $\alpha > 0.9$ is considered as excellent; $\alpha > 0.8$, good; $\alpha > 0.7$, acceptable; $\alpha > 0.6$, questionable; $\alpha > 0.5$, poor; and the

value of $\alpha < 0.5$ as unacceptable (Gliem and Gliem 2003). Analysis of the entire dataset resulted in $\alpha = 0.8595$. Alpha on the data subsets—national, provincial, city/municipal—was also computed, resulting to α equal to 0.8097, 0.9487, and 0.9001, respectively, suggesting that the survey results provided reliable estimates for the mainstreaming indicators.

As detailed in Chapter 4, the mainstreaming challenges can be either serious barriers or opportunities for mainstreaming, depending on the conditions surrounding the challenges. The assessments of the respondents indicated that the challenges in mainstreaming exist in a certain spectrum that has levels depicting the transition of the challenges from barriers to opportunities for mainstreaming. Consequently, in this research, four categories in the spectrum were designed. First, the indicators with scores $1.0 \le n < 2.0$ were considered firstlevel or the primary barriers to mainstreaming CCA. The first-level mainstreaming indicators represented the primary barriers that constrained the effective integration of CCA into the local planning system. Indicators that scored from $2.0 \le n < 2.25$ were second-level barriers; they were less significant than the primary (barriers) but were considered to be serious problems. Indicators that scored from $2.25 \le n < 2.5$ were third-level barriers that represented those mainstreaming challenges that were transitioning from barriers to opportunities for mainstreaming. Lastly, the fourth-level mainstreaming indicators (i.e., with scores ≥ 2.5) were challenges that the system had already overcome and had positive effects on the implementation process, and, therefore, were opportunities for mainstreaming. In essence, the indicator scores reflected the state-of-play of the mainstreaming challenges at each governance scale (i.e., national, provincial, and municipal/city).

The indicator scores at all scales (Figure 10) identified the *institutional issues* indicator as the primary barrier to mainstreaming CCA into the local land-use plans and therefore a key impediment to the effective operationalisation of the local mainstreaming endeavour. Interviews revealed these issues to be the (1) fragmented national laws; (2) overlapping policy requirements that burden LGUs; and (3) the lack of detailed guidelines for mainstreaming CCA-DRR into the local land-use plans. These issues can be summarised as the lack of institutional support mechanisms for mainstreaming CCA (detailed discussions presented in Chapter 6). Next highlighted, as second-level barriers, were the *availability* and *access to information* indicators. In contrast, the *credibility and reliability of information*, *local government prioritisation, institutional incentive,* and *stability of fund* indicators were assessed as opportunities for mainstreaming CCA (Figure 10).



Figure 10 Mainstreaming indicator scores and levels, by scale

Some variations in the assessments across scales reflected the differences in the conditions in Albay (both provincial and city/municipal) and other LGUs in the country, particularly in terms of the following: *knowledge and awareness, leadership*, and *community support*. These indicators were assessed as opportunities at the provincial and city/municipal scales, but were barriers at the national scale evaluations. Similarly, the *commitment to CCA* and *access to funds* were opportunities at the provincial scale of Albay, but were second-level barriers at the national scale. These differences were caused by the existence of a climate change champion in Albay (i.e., *leadership* indicator), in the person of the provincial chief executive, Governor Jose Clemente Salceda (see Chapters 6 and 7 for more analysis).

Other indicator scores reflected the national versus local perspectives, that is, the national respondents generally reacted to the question in terms of the national institutions involved in climate change and land-use planning. On the other hand, the local respondents (i.e., provincial and city/municipal) evaluated the question in relation to the local institutions in Albay. This case applied to the *organisational cohesion*, *local government prioritisation*, and *institutional incentive* indicators.

Based on the interviews, the scores for the indicators *communication of information*, *translation of information*, *autonomy of local government*, and *organisational cooperation and collaboration arrangements* had to be clarified at the individual scale. That is, the issues encompassing the indicators could not be generalised across scales and were investigated from the national to the provincial and to the city/municipal scales.

Finally, the interview results suggested that the scores for the *availability of funds*, *experts*, and *human resources* could not be taken at face value. The questions related to these indicators needed to more closely reflect the "true" conditions surrounding the challenges. For example, scores on the *availability of funds* were based on questions concerned with the availability and regularity of funds for mainstreaming activities. The interview results showed that although funds were available, they were not sufficient to finance the various adaptation needs of the LGUs. Likewise, what these funds were allocated for was not explicitly defined. The same argument applied to the *availability of experts* and the *availability of human resources* indicators. The interviews also disclosed that although climate change experts were available, their number was insufficient to effectively address the needs of all LGUs. Similarly, the respondents reported that there were deeper institutional concerns than the availability of local personnel to undertake CCA activities that hindered the mainstreaming process. This last point is expounded in Chapter 6, which presents the in-depth qualitative analysis the indicator related to this concern (i.e., institutional issues indicator).

5.3.2 Nature of mainstreaming challenges

This section presents the nature or characteristics of the mainstreaming challenges as gleaned from the case study analysis. It elaborates on the mainstreaming challenge spectrum advocated by the research, expounds the discussion on the mainstreaming challenge levels, and continues to explain the linkages among the mainstreaming challenges.

5.3.2.1 Mainstreaming challenge spectrum

Mainstreaming challenges exist within a certain spectrum, with the barriers and opportunities for adaptation representing the extreme ends of this spectrum. Knowing which challenges are barriers and which are opportunities can help planners and decision makers prioritise activities. For example, if the *institutional issues* indicator is a primary barrier to mainstreaming CCA (at all scales), then this signifies the importance of institutional capacity development (Table 11). This also confirms the notion that despite the level of technology, information, and financial and human resources, weak and ineffective institutional structures inhibit the success of an adaptation measure (Inderberg and Eikeland 2009; Ayers et al. 2014) (See Chapter 6 for extensive discussion on this indicator).

	Mainstreaming indicator levels					
Scale	1 st	Barriers 2 nd	3 rd	Opportunity 4 th		
ALL scales	• Institutional issues	 Availability of information Access to information 		 Credibility and reliability of information Local government prioritisation Institutional incentive Stability of funds 		
National/ other LGUs	 Communication of information Organisational cohesion 	 Knowledge & awareness Commitment to CCA Community support Organisational cooperation and collaboration arrangements Access to funds 	• Leadership	 Translation of information Autonomy of local governments 		
Provincial		Communication of information	 Organisational cohesion Organisational cooperation and collaboration arrangements 	 Translation of information Knowledge & awareness Autonomy of local governments Leadership Commitment to CCA Community support Access to funds 		
City/ Municipal		 Autonomy of local governments Organisational cooperation and collaboration arrangements 	 Translation of information Commitment to CCA Organisational cohesion Access to funds 	 Communication of information Knowledge & awareness Leadership Community support 		

Table 11 Matrix of challenges in mainstreaming climate change adaptation and disaster risk reduction into the local land-use plans: Albay, Philippines

Notes: Mainstreaming indicator levels: $1^{st} - 1.0 \le n < 2$ (primary barrier); $2^{nd} - 2 \le n < 2.25$; $3^{rd} - 2.25 \le n < 2.5$; 4th: $n \ge 2.5$ (opportunity).

* The following indicators are not included in the matrix as they could not be taken at face value: *availability of funds, availability of experts,* and *availability of human resources.* Survey questions relating to these indicators need to be modified to determine the "true" conditions surrounding the challenges.

Similarly, knowing the opportunities can help decision makers determine which factors can be utilised to help in the mainstreaming process. To illustrate, *credibility and reliability of information* as an opportunity implies that planners considered the information

generated by the official government data producers to be reliable and credible. This knowledge can help decision makers allocate resources more efficiently. Further, since LGUs were willing to incorporate the information into the local land-use plans, policy- and decision-makers can instead focus their efforts on matching the data made available by producers to the needs of the users, and improve the user access to the information (2nd level barriers in Albay and other LGUs). Likewise, the stability of funds as an opportunity suggests that (1) CCA-DRR funds were part of the national and local budgets; (2) financial sources for long-term CCA-DRR activities existed; and (3) LGUs can initiate activities without concern that the funding would cease. For example, the People's Survival Fund provides financial funding for CCA-DRR initiatives and the Disaster Risk Reduction and Management Act mandates the creation of the Local Disaster Risk Reduction and Management Fund to support CCA-DRR activities. Meanwhile, the Climate Change Act decrees government financial institutions to provide LGUs with preferential financial packages for climate change-related projects. Moreover, this Act orders the LGUs to allocate climate change funds (i.e., for programs and plans) from their annual appropriations. Therefore, the results imply that creating new fund sources was not a priority problem; instead, efforts should focus on (1) ensuring that the available funds are adequate to support local CCA efforts; (2) improving the coverage of the funds to include CCA planning initiatives such as mainstreaming; and (3) guaranteeing that intended beneficiaries know how to access these financial sources (2nd level barrier in other LGUs).

5.3.2.2 Mainstreaming challenge levels

The mainstreaming barriers exist in varying degrees of severity (i.e., mainstreaming indicator levels). Policy-learning can be improved by understanding how a barrier can transcend from a lower to a higher level within the mainstreaming challenge spectrum. For example, *organisational cohesion* was a primary barrier at the national scale, but was a third-level barrier in Albay. During the interviews, the respondents clarified that the lack of organisational cohesion was an "inherent problem in the Philippine government system." Essentially, the various government departments or organisations have their own key result areas (KRAs) that guide their respective goals and activities. These KRAs are set individually and independently from one organisation to another; hence, the KRAs seldom align. Consequently, even with a common agenda such as CCA-DRR, the organisations tend to formulate disharmonised programs, projects, responsibilities, and tasks because these activities are all based on their individual KRAs. With every organisation concerned with meeting its own KRA, efforts are often not in synergy. The lack of organisational cohesion is

translated at the local scale as the unfamiliarity of government field offices concerning the function of their institution in local CCA-DRR activities. Local offices follow their organisations' KRAs and mandated functions. Since these functions are typically not harmonised with one another, the local offices cannot participate effectively in local CCA-DRR initiatives.

In Albay, organisational cohesion was a third-level barrier, indicating better conditions in the province compared to the national scale, especially in land-use planning. This was because the Provincial Government of Albay had issued local policies and regulations that clarified the roles of the varying field offices in terms of mainstreaming CCA into the local land-use plans. Specifically, the Provincial Executive Order 2007-07 incorporated the Mines and Geosciences Bureau and the Environmental Management Bureau, two key agencies that generate climate change-related data, in the Provincial Land-use Committee. Likewise, the Provincial Executive Order No. 2008-03 included the Albay Public Safety and Emergency Management Office into the committee. Essentially, these pieces of local legislation created an institutional structure that combined the organisations involved in land-use planning and those linked to climate change and CCA into a common effortmainstreaming CCA into local land-use plans. This set of information is important in a variety of ways. First, it indicates that lack of organisational cohesion seriously impedes the mainstreaming endeavours and that national efforts are needed to ease the problems created by this barrier. Second, it confirms that local efforts can help transition the challenge from a key barrier into a lesser constraint (i.e., third-level barrier); however, more is needed to transform organisational cohesion as an opportunity.

The concept of mainstreaming challenge levels is also demonstrated by the *access to funds* indicator scores. This indicator defines the user's awareness of the existence of the CCA funds, the convenience of fund processing, and the simplicity of the requirements of securing the funds. In the other LGUs, as represented by the national score (i.e., 2.11), *access to funds* was considered as a serious problem or a 2nd level barrier. On the other hand, this indicator was an opportunity at the provincial scale and a 3rd level barrier at the city/municipal scale. The difference in scores was due to the local initiatives in Albay which made CCA-DRR funds more accessible in the province. For example, the Provincial Government of Albay had itemised the Albay in Action on Climate Change (A2C2) program in the provincial budget. The A2C2 was an initiative of the Provincial Government of Albay to pioneer the mainstreaming of CCA-DRR into local development efforts and policies (Uy et al. 2011; Salceda 2012). Similarly, the Provincial Government of Albay allocated a regular budget to the Center for Initiatives and Research for Climate Adaptation (CIRCA), the
institution assigned to implement the A2C2 program (PGA 2008; Salceda & Rangasa 2011). Meanwhile, the Albay Public Safety & Emergency Management Office, institutionalised in 1994 as a provincial agency in charge of disaster management, had expanded its concerns to include CCA (Lasco et al. 2008; Interviews 2014). Accordingly, the Provincial Government of Albay strengthened the ability of to spearhead CCA-DRR programs. The agency had a regular allocation from the annual provincial budget and had access to the calamity fund for its operations (Salceda 2012). Thus, access to funds was gauged as an opportunity at the provincial scale (score: 2.50).

The Provincial Government of Albay also "tries to complement the funds and resources" of its component LGUs (Interviews 2014). For example, upon the request of a municipality that was "struggling to use the hazard maps in analysing land-uses," the Provincial Government of Albay funded a two-day training on hazard mapping which was presided by the Mines and Geoscience Bureau Region 5 (Interviews 2014). Still, respondents evaluated access to funds as a third-level barrier at the city/municipality scale (score: 2.42) (Figure 10). The results of the interviews suggested that the LGUs at this scale were aware of the existence of funds for CCA, but they lacked the additional information and institutional mechanisms to have convenient access to the funds. As some respondents shared:

While we know that the (CCA) funds are mandated by law, we are not privy to where the funds are allocated.

A certain percentage of the economic development fund and special appropriations from the local savings can be used for CCA, as long as it is supported by an approved local board ordinance (Interviews 2014).

Thus, in the Albay context, access to funds need not be prioritised by the provincial government and current activities just need to be continued since the indicator was assessed as an opportunity. However, the city/municipal score suggested that while the effects of the provincial efforts trickle down to the city/municipal scale, more was needed to overcome the challenge completely. Actions to consider include (1) improving the city/municipal government (staff, planners, officials) knowledge on the allocation of the CCA funds (i.e., what funds can be used for which activities); and (2) formulating local ordinances regarding local budget support for CCA activities. Meanwhile, the conditions at LGUs other than Albay imply that access to funds is a priority challenge that needs to be addressed. Being a 2nd level barrier, serious interventions from the national, provincial, and city/municipal scales may be in order.

5.3.2.3 Mainstreaming challenge linkages

The results presented in Chapter 4 identified that challenges themselves are interlinked. This chapter expands this finding to indicate that the challenges can either be barriers or opportunities to one another. For example, if climate change-related information is not deemed credible and reliable, then information will not be trusted and used. This condition is a barrier to the advancement of knowledge and awareness of planners, decision makers, and society on climate change and adaptation. Likewise, a challenge, when turned into an opportunity, can positively influence another (challenge). For instance, when CCA is a priority, the local government's commitment to CCA may progress, leading to an institutionalised legislative or administrative framework for adaptation. Similarly, CCA activities may no longer compete for local funding against other goals in the local agenda, which may raise the availability of funds for CCA efforts.

Linkages among the mainstreaming challenges are best illustrated by the ripple effects of having a climate change champion (i.e., *leadership* indicator) in Albay. For one, Governor Salceda was influential in institutionalising CCA as a priority agenda in the local government. Under his leadership, CCA was established as a provincial policy by virtue of Provincial Resolution 2007-04. This was supported by the Provincial Resolution 2007-24, which decreed Albay as "one of the first and pioneering prototype province that will adopt climate change adaptation." This policy aimed to instill CCA into the political and social consciousness of the present and future generations of government officials and people in Albay (PGA 2007). These provincial policies and regulations, along with many others, established a legislative framework for CCA in the province and created a provincial government *committed to CCA*. Likewise, the varying initiatives under the governance of the climate change champion helped (1) decision makers, planners, and the local communities in Albay become aware and knowledgeable on CCA concerns; (2) incentivise planners and decision makers to mainstream adaptation into planning and other decision-making processes; and (3) gain community support for CCA initiatives (Table 12). The role of the climate change champion in the local mainstreaming process in Albay is elaborated in Chapter 6.

uauptunon	
Policy/Activity/Program	Description
Provincial Executive Order 2007-	Established the Center for Initiatives and Research for Climate
12-A	Adaptation (CIRCA) as a key institution on climate change.
Memorandum of Agreement	Institutionalised the Climate Change Academy as Albay's main
between the Provincial	arm in enhancing and strengthening the theoretical knowledge
Government of Albay and Bicol	and practical skills of major local stakeholders on climate and
University	disaster risk assessment.
Provincial Council Appropriation	Identified Albay in Action on Climate Change (A2C2) program
Ordinance 2007-01	as a provincial budgetary item with corresponding funds for its
	activities.
Provincial Council Resolution	Urged the Provincial Government of Albay-CIRCA,
2008-44	Department of Agriculture, Department of Environment and
	Natural Resources, and the Department of Agrarian Reform to
	conduct training and workshops to incorporate climate change
	in the local land-use plans.
Provincial Executive Order No.	Stipulated for the Climate Change Academy to hold
2011-02	environmental classes at the key university in the region (Bicol
	University) starting 2011.
2007 First National Conferences	Provincial Government of Albay spearheaded the first National
on Climate Change Adaptation	Conferences on Climate Change Adaptation which was at the
	provincial capital (Legazpi City). This brought the concept of
	climate change into the doorstep of the people in Albay.
2008 First Interfaith Forum on	Brought together people from the Catholic, Islam, Iglesia ni
Climate Change Adaptation	Cristo, and Seventh Day Adventist faiths to discuss CCA and
	DRR.

Table 12 Primary local laws, programs and activities in Albay related to climate change adaptation

Sources: Lasco et al. n.d.; Lasco, et al. 2008; UNISDR 2012; various Provincial Resolutions; Personal Communication 2014

5.4 Summary and conclusions

In this research, mainstreaming challenges are the factors that affect or influence the effective operationalisation of the mainstreaming measure, and can be represented either as barriers or opportunities. These mainstreaming challenges are summarised into three capacity classifications—institutional, information, and resource capacities. The mainstreaming challenges, when transformed as quantitative mainstreaming indicators, effectively assessed the conditions surrounding the process of mainstreaming CCA into the local land-use plans in Albay, Philippines.

To understand better *the state-of-play of and linkages between the challenges in mainstreaming CCA (Objective 3)* this chapter identified the barriers to mainstreaming CCA according to the severity of their effects on the endeavour. Specifically, the *institutional issues* indicator was assessed as the primary barrier in operationalising the approach, followed by the *availability of* and *access to information* as second-level barriers. Knowing the extent of a barrier's impact on the mainstreaming process can assist planners and decision- makers in prioritising the barriers to address. This is significant, especially in the local context where local governments face a number of governance and development

constraints. For example, the literature identifies barriers related to adaptation funds to be among the primary reasons why the implementation of local adaptation is delayed (Gardner et al. 2010; Moser & Ekstrom 2010; Lehmann et al. 2015). In response, the government may concentrate on creating new funds for CCA. However, the case study suggests that there is no immediate need to create new funds in the Philippines, as stable funds for CCA already exist. Instead, the more pressing concern involves improving the access of users to these funds (i.e., second-level barrier in LGUs other than Albay). Similarly, understanding the severity of the impacts of a certain barrier can help in policy-learning. To illustrate, how the organisational cohesion indicator became a third-level barrier in Albay can be a learning experience for the national government and the other LGUs in the country.

Meanwhile, several indicators were evaluated as opportunities for mainstreaming CCA, including—*credibility and reliability of information, local government prioritisation, institutional incentive*, and *stability of funds*. These assessments suggest that barriers can be overcome to transcend into opportunities for mainstreaming CCA. Also, the mainstreaming challenges themselves are linked and can be barriers or opportunities to one another. By knowing the relationships among the mainstreaming challenges, planners and decision-makers can formulate strategies that can have maximum impacts on mainstreaming CCA. This chapter introduced the quantitative mainstreaming indicators the research developed to help planners and decision-makers monitor the adaptation process and its implementation, and also track the progress of adaptation efforts. Also, it illustrated how the mainstreaming indicators can be used in prioritising and devising strategies to address the adaptation challenges, thereby illustrating the promise of these tools in adaptation planning and decision-making. The next chapter solidifies the importance of these metrics, as it presents the in-depth interpretations of the two very important challenges in mainstreaming CCA in the Albay case—*institutional issues* and local *leadership*.

CHAPTER 6: INSTITUTIONAL CAPACITY FOR LONG-TERM CLIMATE CHANGE ADAPTATION: EVIDENCE FROM LAND-USE PLANNING IN ALBAY, PHILIPPINES*

* Cuevas, S.C., Peterson A., Robinson, C. & Morrison, T.H. (2015). Institutional Capacity for Long-Term Climate Change Adaptation: Evidence from Land-use Planning in Albay, Philippines. *Regional Environmental Change*. DOI 10.1007/s10113-015-0909-8

6.1 Introduction

One of the key challenges in climate change adaptation is the tendency of individuals to resist and delay change, or the failure of institutions to create an enabling environment that can promote efforts to plan for and respond to the effects of a changing climate. For this reason, improving an adaptation approach utilising existing schemes (i.e., plans, strategies, organisational institutions, etc.) is more appropriate than designing and creating new or separate institutions for managing climate change adaptation (CCA) (Klein et al. 2005).

Chapter 4 introduced the various challenges in mainstreaming operationalisation and discussed the nature of these challenges through quantitative analysis, particularly, correlation analysis and assessments by mainstreaming indicator scorings. Chapter 5 further explained how the indicator scores can be interpreted and illustrated how the challenges in mainstreaming exist within a certain spectrum, with the barriers and opportunities for adaptation representing the extreme ends of this spectrum. This chapter continues this discussion and explores further the primary barriers (i.e., *institutional issues*) and the substantial opportunity (i.e., local *leadership*) for mainstreaming CCA that were highlighted by the mainstreaming indicators.

CCA is generally viewed as a technical problem; hence, most adaptation studies have focused on assessing the environmental impacts of climate change and the resulting vulnerabilities due to the system's biological risk exposure to these hazards (Resurreccion et al. 2008; Lebel et al. 2012). However, climate change and CCA are challenges about "leadership, coordination, and collective action," and thus they are about institutions (Evans & Stevens 2009, p. 2). Regardless of the existing technology, information, and financial and human resources, weak and inefficient institutional structures significantly constrain the success of an adaptation measure (Inderberg & Eikeland 2009; Ayers et al. 2014).

This chapter moves closer to *determining how to overcome the challenges in mainstreaming CCA* (i.e., Objective 3) by verifying the significance of institutions in the context of CCA through quantitative indicators and qualitative assessments. The mainstreaming indicators provided the quantitative aspect of the analysis, while the IAD-CCA framework was the tool used in qualifying the indicator scores. The IAD-CCA was most useful in organising the data collected and in providing structure for analysing the institutional setting where CCA was being mainstreamed. For example, the mainstreaming indicator that garnered the lowest score (i.e., closest to the value of 1) was examined first. The responsibilities and linkages among the key actors involved in this challenge were then identified. Next, the institutional arrangements that guided the actions of these actors were mapped to determine their patterns of interactions. Subsequently, the outcomes of these patterns of interaction were determined. The underlying issues related to the patterns of interaction and outcomes were examined to understand further the score associated with the mainstreaming challenge

By using the IAD-CCA framework in the analysis, this chapter explains how institutional issues impede the effective operationalisation of mainstreaming CCA and establishes the significance of developing the institutional capacity of systems to address the challenges in mainstreaming. Also, it illustrates how the existing institutional support mechanisms helped the local leadership (i.e., climate change champion) in Albay, Philippines to become a substantial opportunity for mainstreaming. Consequently, this chapter proposes that, in mainstreaming CCA, institutional capacity development is more important than the need to advance information and resource capacities. The key finding identified in this chapter also generated a more refined understanding of the operationalisation of mainstreaming in local CCA (Objective 4).

6.2 Significance of developing institutional capacity for mainstreaming CCA

This section presents the core on-ground conditions relating to the mainstreaming indicator scores, specifically, the two indicators—*institutional issues* and *local leadership*— highlighted as the primary barrier and substantial opportunity, respectively, for mainstreaming CCA into the local land-use plans (Table 13). Institutional issues refer to the absence or presence of rule-based institutional questions or conflicts that inhibit the effective integration of CCA into local land-use planning. On the other hand, leadership pertains to the absence/existence of a CCA "champion" in the locality and the extent of the champion's influence on the community's behaviour.

Table 13 Primary b	parrier and su	ubstantial o	opportunit	y for mai	instreami	ng cli	imate c	hange
adaptation into loc	al land-use p	olans in All	bay, Philip	opines				

Mainstreaming indicator	National	Provincial	City/Municipal
Institutional capacity			
Leadership	2.38	2.67	2.57
Institutional issues	1.00	1.46	1.36
Note: Indicator levels = $1^{st} - 1.0 \le n < 2$; $2^{nd} - 2 \le 2$	\leq n < 2.25; 3 rd - 2	$.25 \le n < 2.5; 4$	th: $n \ge 2.5$

6.2.1 Institutional issues

As presented in Chapter 5, the indicator scores suggested that the primary barriers to mainstreaming CCA into the local land-use plans in the Philippines, and Albay in particular, were linked to institutional capacity (i.e., *institutional issues*). Specifically, these issues included (1) fragmented national laws and regulations; (2) overlapping or multiplicity of policy requirements; and (3) a shortage in guidelines for mainstreaming CCA into the local land-use plan. The issues all relate to the lack of institutional mechanisms that support the mainstreaming initiative. Another identified institutional issue was associated with political concerns (i.e., decision-making influenced by personal interests of politicians).

6.2.1.1 Fragmented laws and regulations

During the interviews (at all scales), the respondents cited the institutionalisation of the Local Disaster Risk Reduction and Management (LDRRM) officer as a key concern in mainstreaming CCA. At first glance, this issue seemed to fall under the *availability of human resources* challenge. However, intensive analysis revealed the key issue was discord among the regulations affecting the creation of this government position (i.e., fragmented laws and regulations). Creating the LDRRM Officer position in cities and municipalities is critical in advancing local climate change concerns in the Philippines. This position assumes the tasks and responsibilities related to CCA-DRR at the local scale, including spearheading the mainstreaming of CCA into the local land-use plan.

By virtue of the Disaster Risk Reduction and Management Act of 2010 (RA 10121), all local government units (LGUs) are mandated to create the LDRRM officer position, subject to the rules and regulations of the Department of Budget and Management (i.e., the key agency responsible for the government budget) concerning the budgetary source, and dependent on the standards and guidelines provided by the Civil Service Commission (i.e., central personnel agency mandated to formulate policies and regulations for government employment). Based on the Department of Budget and Management Memorandum dated March 15, 2012, all personal services requirements (i.e., salaries and compensations of government employees) for the LDRRM officer were required to be sourced from the LGU funds, and be subject to the personal services limitation of the LGU budgets. This personal services limitation is governed by Section 325 of the Local Government Code (RA 7160), which states that the personal services of LGUs should not exceed 45% and 55% of their funds, for first to third income class provinces, cities, and municipalities, and fourth class or lower, respectively. As one of the respondent commented, the Local Government Code is regarded as the "bible" of LGUs as it outlines all the power, authority, responsibilities, and

(allowable and prohibited) acts of LGUs. This rule on personal services limitation posed a significant problem since most (if not all) LGUs had reached their respective budgetary ceilings.

However, LGUs are mandated to comply with the law. Thus, without sufficient financial resources to create a permanent LDRRM officer, LGUs resorted to designating the position to existing regular and permanent LGU personnel. This meant that aside from the usual responsibilities of the staff, s/he was allocated additional tasks and "great accountability especially when there are disasters", without additional compensation, financial or otherwise. This scenario explains the high score for the mainstreaming indicator *availability of human resources*, especially at the city/municipal scale. Hence, although there were available LGU personnel tasked to attend to CCA concerns, these personnel were the designated LDRRM officers who were typically overworked, underpaid, and unmotivated (Interviews 2014).

The few LGUs that had the funds to create the LDRRM officer position were faced with another institutional constraint, that is, the lack of standards and guidelines for creating such a position. Prior to April 2014, the Civil Service Commission has not determined the said guidelines. In such cases, the local chief executives (i.e., mayors) typically exercised their authority to create casual local government positions "without need of approval or attestation by the Civil Service Commission," as provided by Section 77 of the Local Government Code. However, this authority was oftentimes influenced by the "padrino" system, the norm or value system of political patronage where a person gives or gains favour, promotion, or political appointment through social (friendship) or familial affiliation, instead of merit and qualifications. These political appointments often result in LDRRM officers with less than the desired knowledge, experience, and appreciation of climate change concerns. In cases where the appointed LDRRM officer is qualified and/or experienced, the temporary status of the position presents another difficulty. As a political appointee, the LDRMM officer is co-terminus with the mayor (chief executive at the city/municipal scale). Thus, a change in local political power signifies losses in human resource investments, which in this case is the CCA-DRR trained LDRRM officer (Interviews 2014).

In early 2014, these budgetary and guidelines issues were addressed by a Technical Working Group composed of the Civil Service Commission, National Disaster Risk Reduction and Management Development Council, Department of Interior and Local Government, and the Department of Budget and Management. The Technical Working Group formulated the implementing guidelines for establishing the Local Disaster Risk Reduction and Management Offices in provinces, cities, and municipalities. These guidelines were issued on April 4, 2014 through the Joint Memorandum Circular No. 2014-1 among the

four agencies. Specifically, Section 4.3 of the Joint Memorandum waives the enforcement of the budgetary personal services limit, thus enabling the LGUs to finance the initial year requirements for creating the LDRRM officer position. As the "usual practice" in these cases, the personal services for the LDRRM officer can now be incorporated as a regular item in the succeeding fiscal budget allocations, thereby providing steady funds for the position (Interviews 2014). Similarly, the Joint Memorandum provides the necessary standards and guidelines for creating the position as stated in "Section 6: Position Titles, Qualification Standards, and Salary Grades for the Technical Staff of the LDRRMO."

While the impacts of the Joint Memorandum are yet to be realised, the amendments and the new sets of rules it provides are evidence of institutional mechanisms that support the institutional foundation for CCA. This institutional support is relevant to improve the resource capacity of LGUs, especially since the national directives for CCA-DRR are continuously being implemented. For example, in June 2014, the Department of Interior and Local Government issued Memorandum Circular 2014-70 instructed the LGUs to formulate their local disaster preparedness plans in anticipation of the erratic typhoons expected in 2014.

6.2.1.2 Lack of guidance for mainstreaming CCA into the local land-use plan

In 2009 and 2010, the Climate Change Act (RA 9729) and the Disaster Risk Reduction and Management Act (RA 10121) were implemented, respectively, thereby institutionalising the CCA-DRR agenda in the Philippines. However, mainstreaming CCA-DRR is a new initiative; hence, information regarding its operational procedures is limited. Consequently, its operationalisation becomes a challenge, especially at the local scale (Olhoff & Schaer 2010; Mercado 2011). Since the Climate Change Act has been enacted, the mechanisms to support its implementation have been insufficient (i.e., lack of guidelines to support the mainstreaming efforts). The LGUs have found it difficult to comply with the existing laws (Interviews 2014). Perhaps the closest to an LGU-oriented CCA-DRR mainstreaming guideline is the document produced by the National Economic and Development Authority in 2008 entitled, Mainstreaming Disaster Risk Reduction in Subnational Development and Land-use/Physical Planning in the Philippines. Still, this guideline is more effective at the provincial rather than at the city or municipal scale. Likewise, it is too technical for LGU decision-makers and planners (Mercado 2011). Thus, LGUs have appealed for additional documents to support mainstreaming CCA into the local land-use plan (RDC Region XII 2012).

In early 2014, the Housing and Land-use Regulatory Board (HLURB) released the "Supplemental Guidelines on Mainstreaming Climate and Disaster Risks in the Comprehensive Land-use Plan" (Personal communication 2014). This document was produced in collaboration with the Climate Change Commission, and in consultation with other partner agencies (i.e., National Economic and Development Authority, Department of Interior and Local Government, etc.). The feat was accomplished through the Project Climate Twin Phoenix, with assistance from the United Nations Development Programme and Australian Agency for International Development (ReliefWeb 2014; Interviews 2014). Although this development is expected to improve LGUs' capacities to mainstream CCA-DRR into the local land-use plan, its impacts are yet to be determined.

6.2.1.3 Overlapping and multiple policy requirements

LGUs are mandated to comply with approximately 30 sectoral plans as stipulated by various laws and regulations (Mercado 2011) (Table 14). Complying with these rules has been a challenge for LGUs not only due to the lack of resources, but also because LGUs regard some of the mandated plans as repetitive, unnecessary, and generally overwhelming in number (Gotis 2008; Interviews 2014). According to local respondents, the multiple requirements prevent the understaffed LGUs from focusing on mainstreaming CCA into local plans, including the local land-use plan.

Mandated local plans	Legal basis
Comprehensive Land-use Plan	RA 7279: Urban Development and Housing Act of 1992
Local Shelter Plan	RA 7160: Local Government Code of 1991
Comprehensive Development Plan	RA 7160: Local Government Code of 1991
Local Development Investment Plan	
Annual Investment Program	
Executive and Legislative Agenda	
Local Climate Change Action Plan	RA 9729: Climate Change Act of 2009
Local Disaster Risk Reduction and	RA 10121: Disaster Risk Reduction and Management Act of
Management Plan	2010
Local Poverty Reduction Action Plan	RA 8425: Social Reform and Poverty Alleviation Act of 1997
	DBM-DILG-DSWD-NAPC Joint Memorandum Circular No. 1
	Series of 2012: Policy Guidelines and Procedures in the
	Implementation of Bottom-Up Planning and Budgeting for the
	FY 2013 Budget Preparation
Local Solid Waste Management Plan	RA 9003: Ecological Solid Waste Management Act of 2000
Local Tourism Development Plan	RA 9593: The Tourism Act of 2009
Strategic Agricultural and Fisheries	RA 8435: Agriculture and Fisheries Modernization Act of 1998
Development Zones Plan	
Local Nutrition Action Plan	DILG Memorandum Circular 2012-89: Adoption of the
	Philippine Plan of Action for Nutrition (PPAN) 2011-2016

Table 14 Selected local government unit mandated plans and their legal basis

Notes: RA – Republic Act; DBM – Department of Budget and Management; DILG – Department of Interior and Local Government; DSWD – Department of Social Welfare and Management; NAPC – National Anti-Poverty Commission

6.2.1.4 Political concerns

Politics also affected the mainstreaming of CCA, particularly when members of councils or the local chief executives decided on local land-use plan concerns for personal gains. Most of the institutional arrangements created by the Local Government Code provide local politicians a number of avenues to influence the land-use planning procedure. For one, the Local Development Councils, the main body that formulates the local plan, is predominantly comprised of politicians (Serote 2004). Similarly, politicians rule the legislative body that enacts the plan into zoning ordinances (Sec. 446 and Sec. 457 of the Local Government Code). Sections 54 and 55 of the Local Government Code also authorise local executives to approve or veto local ordinances, including those related to zoning regulations. Hence, local politicians can reject proposals for the conversion or reclassification of lands in critical sites or danger zones when such changes threaten their personal investments located at the sites (Interviews 2014). This practice is predominant in the Philippines where "zoning classifications are likewise the subjects of political trade-offs, compromise, and corrupt practices" (Corpuz 2012, p. 9).

These issues show that institutions significantly influence the process of mainstreaming CCA into the local land-use plan. This is further supported by the next section which provides evidence that an institutional challenge, when transformed into an opportunity, also is key to the effective operationalisation of the mainstreaming approach.

6.2.2 Leadership: Climate change champion

A key component in local CCA is leadership or the "emergence of an identifiable political/administrative champion(s) for climate change issues" (Roberts 2008, p. 527). Leadership is a crucial aspect at any stage of the adaptation process. Furthermore, a climate change champion can raise awareness of climate change, initiate CCA efforts, put CCA high into the local government agenda, and uphold future CCA initiatives. Hence, having such a leader can be a significant opportunity, whereas the absence of a champion can weaken the climate change agenda and be a critical barrier to the endeavour (Roberts 2008; Burch 2010; Moser & Ekstrom 2010; Oberlack & Eisenack 2014; Waters et al. 2014). In this research's context, the champion may take the form of an institutional organisation or an individual whose position symbolises a social structure-based institution that can form individual and social expectations, and can influence relations, interactions, behaviours, and conduct of people (Cuevas et al. 2014).

6.2.2.1 National scale

The *leadership* indicator was assessed as a third-level challenge at the national scale (Table 13). The Climate Change Act institutionalised the Climate Change Commission as the key agency tasked to coordinate, monitor, and evaluate government programs and action plans relating to climate change, thus making the agency the main climate change champion in the country.

In relation to land-use planning, the Climate Change Commission spearheads the Ecotowns (ecosystems town) project, an initiative that aims to develop climate changeresilient towns with improved adaptive capacities. The project also aims to demonstrate the convergence of CCA and mitigation actions, as well as the integrated ecosystem-based management approach in planning (CCC 2011). Through Memorandum of Agreements, the Climate Change Commission works closely with the 10 LGUs involved in the project. In the beginning (of the project), the mayors of the participating LGUs were a mix of climate change skeptics, non-believers, believers, and individuals who lacked interest in the issue. As the Ecotown project progressed and the Climate Change Commission and LGUs worked together, some of the mayors realised the significance of CCA (in general) and mainstreaming CCA into local plans (in particular). As such, they were influenced to champion CCA in their localities. This development paved the way for more effective and efficient transactions of CCA initiatives (Interviews 2014).

However, the Climate Change Commission is a national agency. Although it can be effective at the national scale, it has limited impact at the local scale. Therefore, for the onground mainstreaming of CCA initiatives, having a climate change champion at the local scale is a significant factor. This was confirmed by the evidence from Albay.

6.2.2.2 Local scale: Albay province

By virtue of the Local Government Code, LGUs in the Philippines have extensive local autonomy and increased powers, authority, responsibilities, and resources to govern their localities (Gonzales 1997; Serote 2004). Consequently, local leadership is critical in the development of an LGU. Similarly, it can be a major factor in the demise of a LGU due to corrupt local leaders. Although the latter are prevalent in the Philippines, as evidenced by the earlier discussion on political concerns, this chapter focuses on how local leadership can be vital in advancing local CCA. More importantly, it aims to demonstrate that the institutional mechanisms to support local leaders (i.e., Local Government Code) are already established in the Philippines. In Albay, these mechanisms were utilised by the climate change champion in the person of the provincial chief executive (i.e., Governor Jose Salceda).

The provisions of the Local Government Code offered Governor Salceda the essential institutional support to advocate and execute CCA initiatives effectively in the province of Albay. For example, according to the Local Government Code [Section 465 (b) (2) (ii)] the provincial governor has the authority to call for conventions, conferences, seminars, or meetings on concerns he deems significant to promote the general welfare of the province and its constituents. Thus, in 2007, the Provincial Government of Albay assembled government officials, academics, researchers, NGOs, the business sector, local community representatives, and donor communities into the First National Conference on Climate Change Adaptation to discuss the climate change agenda. The conference resulted in the "Albay Declaration on Climate Change Adaptation" that called for the early passage of the Climate Change Act, and consequently the creation of the Climate Change Commission. In 2009, the provincial government again organised and co-hosted the Second National Conference on Climate Change Adaptation (Benson 2009; Salceda and Rangasa 2011). These activities, along with numerous others, helped raise the knowledge and awareness of the public on climate change issues and gain community support on CCA endeavours. This condition is expected to assist the people who will be affected by modifications in the local land-use plan to understand the need and significance of the changes.

Section 465 of the Local Government Code also authorises the provincial governor to (1) initiate and propose legislative measures to the provincial council; (2) issue executive orders for the enforcement and execution of laws; (3) exercise general supervision and control over all programs, projects, services, and activities of the provincial government; and (4) initiate and maximise the generation of resources and revenues, and apply the same to the implementation of development plans, program objectives and priorities. Hence, Governor Salceda, through the powers and authority of his position, implemented and influenced a number of CCA-DRR initiatives.

For example, Governor Salceda promoted mainstreaming CCA-DRR into the local land-use plan through the Provincial Executive Order 2007-07, which incorporates the Mines and Geosciences Bureau and the Environmental Management Bureau in the Provincial Land-use Committee (i.e., committee that reviews and approves the plan). These two agencies are among the key government institutions that generate climate change-related data. Likewise, through the Provincial Executive Order No. 2008-03, the Albay Public Safety and Emergency Management Office was included in the committee. These developments are expected to ensure that CCA-DRR will be incorporated in the revised local land-use plans of LGUs. Likewise, the legislative actions helped improve *organisational cohesion* in the province's land-use planning system by clarifying the responsibilities of varying institutions concerning

mainstreaming of CCA. Hence, the indicator was assessed as a third level barrier, illustrating its transition from a potential primary barrier toward becoming an opportunity for mainstreaming. According to one respondent:

> In Albay, institutions are working well in relation to CCA-DRR. For example, in other provinces, only the Department of Environment and Natural Resources is a member of the Provincial Land Use Committee; but in Albay, the Mines and Geosciences Bureau, Environmental Management Bureau, and Philippine Institute of Volcanology and Seismology are members too, via Provincial Executive Orders. Thus, climate change data producers are able to provide inputs on how CCA-DRR should be incorporated in the local land-use plans. Because of this arrangement, a direct communication line between data producers and the data users was established.

Moreover, a Memorandum of Agreement between the Provincial Government of Albay and Bicol University has established the Climate Change Academy (now known as the Disaster Risk Reduction Management Training Institute) as Albay's main arm in enhancing and strengthening the knowledge and skills of major local stakeholders on climate and disaster risk assessment. This was followed by the Provincial Executive Order 2011-02 that stipulates for the Climate Change Academy to hold environmental classes at the key university in the region (i.e., Bicol University) starting 2011. Like the activities that promoted the dissemination of climate change information, these regulations enabled land-use planners to become more knowledgeable about climate change issues. Such understanding helped them see the benefits of mainstreaming CCA into the local land-use plan, *incentivised* them to operationalise the mainstreaming approach, and improved their *commitment to CCA*. As one respondent stated:

In my opinion, about 80% of the planners in Albay have the mindset that integrating CCA-DRR into the local land-use plan should be prioritised. This is because of Governor Salceda, who has been actively advocating for CCA-DRR. That is also why the people in Albay, including the planners and the Municipal Planning and Development Council, are aware of climate change.

Finally, as a climate change champion, Governor Salceda was able to place CCA-DRR as a priority agenda of the local government, as evidenced by the number of climate change-related activities in the province. Consequently, the *local government prioritisation* indicator was assessed as an opportunity at the local scale (provincial and city/municipal).

6.3 Summary and conclusion

This chapter comprehensively assessed the *state-of-play of the local mainstreaming process in Albay, Philippines* (i.e., Objective 3), and established the initial foundations for *generating a more refined understanding of the operationalisation of mainstreaming in local CCA* (i.e. Objective 4). The quantitative aspect of the research provided an evaluation of the conditions on-ground and therefore served as guides in determining the challenges that need to be prioritised to effectively mainstream CCA-DRR into the land-use planning process. Based on the indicator scores, this chapter focused on the *institutional issues* surrounding the operationalisation of the approach. The qualitative analysis highlighted why and how institutions can be primary barriers to the local mainstreaming process in the Philippines, as can be seen through the (1) fragmented national laws and regulations; (2) overlapping and multiple policy requirements; and (3) lack of guidelines for mainstreaming CCA into the local land-use plans. These barriers can be summarised as the absence of institutional mechanisms that support the foundations for CCA, specifically the Climate Change Act of 2009 (RA 9729) and the Disaster Risk Reduction Management Act of 2010 (RA 10121).

Mainstreaming CCA is a change that will require broader institutional reforms. Thus, understanding the planning context where these institutional changes (e.g., creation of new policies or amendments in prevailing regulations) are to be implemented is critical (Theesfeld et al. 2010). For example, the provisions in the Disaster Risk Reduction Management Act with regard to the institutionalisation of the LDRRM officer could not be implemented effectively due to budgetary constraints and limited standards and guidelines for implementation. The case implies that institutional mechanisms to support the institutional foundations for CCA are essential to mainstream CCA effectively and to transform mainstreaming challenges into opportunities.

This point is demonstrated also by the circumstances surrounding the leadership challenge, that is, the existence of a climate change champion in Albay. A significant feature included in this mainstreaming indicator was the ability of the champion to influence the behaviour of people and initiate collective action. Leadership became an opportunity to raise the knowledge and awareness of planners, decision-makers, and the community on climate change concerns; positively influence the commitment of the local governments to CCA-DRR initiatives; place CCA-DRR among the priority agenda of the local governments; gain community support for CCA-DRR; and provide institutional incentive through motivating planners and decision-makers to mainstream CCA-DRR into the local plans. Hence, these commonly identified "obstacles to mainstreaming in the Philippines" (Lasco et al. 2008, p. 14) were transformed into opportunities in Albay. The analysis also suggested that in the

Philippines, the local government chief executives are important to champion CCA at the local scale. Vital to this (analysis) is the recognition that local leaders, as climate change champions, have the institutional support to initiate and execute a number of CCA-DRR activities, policies, and orders, by virtue of the Local Government Code.

The institutional dimension of climate change is a crucial facet of adaptation (Adger 2000; Lebel et. al 2012). Institutional changes and concerns are among the important factors that determine the success or failure of an adaptation measure, especially at the local scale (Orindi & Eriksen 2005). This is particularly true in the Philippine context where the improvements in institutional capacities of LGUs can result in a reduction in climate change-associated risks, and where local government institutions are crucial in facilitating local adaptation (Lasco et al. 2008; Uy et al. 2011; Cuevas 2012).

This chapter has strengthened the significance of developing institutional capacities of systems for a long-term adaptation to climate change. To further improve the knowledge on the operationalisation of the mainstreaming initiative and fully achieve the research Objective 4 (i.e., *to generate a more refined understanding of the operationalisation of mainstreaming in local CCA*), Chapter 7 explores how institutional nestedness affects the mainstreaming process in Philippines and how the polycentric type of governance relates to the challenges in mainstreaming CCA in the country.

CHAPTER 7: BARRIERS AND OPPORTUNITIES IN MAINSTREAMING ADAPTATION: A CRITICAL ASSESSMENT

7.1 Introduction

Climate change presents unprecedented risks to society. Thus, the climate change debate has changed from whether there is a need to adapt to how to adapt (Amundsen et al. 2010; Biesbroek et al. 2011). Chapter 2 illustrated that adaptation literature is focusing on the cross-sectoral and interdisciplinary nature of climate change concerns. Hence, adaptation research now is exploring the linkages between CCA and sustainable development, and addressing them as mutually dependent strategies through mainstreaming of CCA (Agrawala & van Aalst 2006; Huxtable & Nguyen 2009).

However, while the queries on the "whys" of mainstreaming have been settled, the "how" questions remain unanswered. Research indicates that "mainstreaming is not yet sufficiently taking place" (Lehmann et al. 2015, p. 93), and that in practice, "actors are searching for solutions to integrate the adaptation objective in existing policy domains" (Uittenbroek et al. 2013, p. 399). While there are handbooks and guidelines to assist in operationalising mainstreaming, scholars and practitioners alike agree that these documents are limited in addressing the challenges in applying the approach on-ground (Ayers et al. 2014; Goosen et al. 2014). Thus, mainstreaming studies commonly cite the barriers or challenges in mainstreaming CCA, rather than account for successful mainstreaming actions (Sharma & Tomar 2010; Nambi & Prabhakar 2011; Uittenbroek et al. 2013; Ayers, et al. 2014).

This research proposes that this situation is caused partly by a lack of understanding of the institutional dimension of the approach. Essentially, the operational procedures devised to apply mainstreaming (i.e., vulnerability assessments, climate risk screening, impact models) focus on climate-related issues (Olhoff & Schaer 2010; Schipper et al. 2010; Lebel et al. 2012). However, mainstreaming operationalisation necessitates more than a climate change perspective. The entire concept of mainstreaming—designing new or modifying existing planning, policy-making, and decision-making structures—is an institutional concern (Ayers & Dodman 2010; Ayers et al. 2014) [Refer to Chapter 3].

Understanding the barriers in adaptation (in general) and mainstreaming CCA (in particular) is important as it can help planners and policy-makers develop the appropriate institutional support to address climate change (Dovers & Hezri 2010; Ekstrom et al. 2011). Research in CCA barriers is emerging, and there are a number of concerns that need to be explored (Biesbroek et al. 2011). Currently, clear conclusions on the subject are that barriers

in adaptation exist, most barriers have historical roots, some barriers are deep-rooted in the system, and it takes time to overcome these barriers. What is less clear includes (1) what these barriers are and their nature, which the research addressed in Chapters 4 and 5; (2) why and how barriers arise and endure, which the research clarified in Chapter 6; and (3) how the barriers can be overcome, which the research discusses in this chapter (Amundsen et al. 2010; Burch 2010; Moser & Ekstrom 2010; Biesbroek et al. 2011; Eisenack et al. 2014).

This chapter begins by investigating the institutional dimension of CCA, and presents the concepts of institutional nestedness, institutional environment, and polycentricity. Next, discussion moves to how the barriers and opportunities for mainstreaming CCA were identified and qualitatively analysed. This chapter concludes that (1) institutional nestedness affects how some factors either become significant barriers or opportunities for mainstreaming CCA; (2) mainstreaming CCA involves a network of interacting institutions and institutional arrangements that transcend across governance scales; (3) local efforts are crucial in transforming potential barriers into opportunities for local mainstreaming; and (4) overcoming these challenges necessitates broad institutional reforms that go beyond the institutional setting where CCA is to be integrated. The findings of this chapter address the fourth objective which is—to generate a more refined understanding of the operationalisation of mainstreaming in local CCA.

7.2 Institutional dimension and the barriers in mainstreaming climate change adaptation

Identifying the source or origin of a barrier (i.e., spatial/jurisdictional or temporal origins) helps to understand how that barrier fits into the CCA system (Moser & Ekstrom 2010). Likewise, it is important to determine the circumstances surrounding the barriers, and how the changes in institutional arrangements and institutional structures are linked to these barriers (Eisenack & Stecker 2012; Eisenack et al. 2014). These concerns involve multiple levels of governance. However, there is limited understanding of the roles of the federal/national, state, and local governments with regard to the barriers to CCA, and how the multilevel governance relationships affect these barriers (Farber 2009; Mukheibir et al. 2013).

Key to this line of investigation is the concept of institutional nestedness in the context of CCA (Corfee-Morlot et al. 2009; Measham et al. 2011). Institutional nestedness involves hierarchical relationships and rules, and deals with the "appropriation, provision, monitoring, enforcement, conflict resolution, and governance activities" of institutions at multiple levels (Ostrom 1990, p. 90). This means that actions at higher levels of government

affect the responses of lower levels, and that clear directions on climate change actions from higher authorities are critical in local policy- and decision-making processes (Burch 2010; Amundsen et al. 2010). Accordingly, the concept of institutional nestedness is crucial in identifying the barriers in local mainstreaming (Corfee-Morlot et al. 2009; Measham et al. 2011).

However, climate change is a "wicked problem" (Lazarus 2010; FitzGibbon & Mensah 2012), and is characterised as a "cross-boundary, multi-level, multi-sectoral and multi-actor challenge" (Fröhlich & Knieling 2013, p. 21). Thus, effective adaptation requires a system in decision-making spread across multiple centers at multiple levels. That is, CCA responses should incorporate both the higher and lower levels of government, and should strike a balance between centralised and decentralised decision-making and governance (Lebel et al. 2006). Thus, some argue that the research on adaptation barriers needs to address multi-governance and cross-scale facets of adaptation (Amundsen et al. 2010; Biesbroek et al. 2011; Mukheibir et al. 2013). Similarly, others focus on the polycentric governance system for CCA in which multiple authorities under multi-layered institutional settings contribute to CCA policy formulation and implementation (Lebel et al. 2006; Ostrom 2010; Cuevas et al. 2014; Jordan & Huitema 2014).

Cuevas et al. (2014, p. 22) stated that the "institutional dimension of climate change adaptation involves an intricate web of relationships between and among institutions." Institutions exist in an institutional environment or the "array of institutions that influence and affect climate change adaptation behaviours and decisions" (Cuevas et al., p. 2). Through a system of institutional arrangements, these institutions-rules, social structures, and organisations—(1) interact with and impact one another; (2) build institutional relationships; and (3) establish institutional linkages (Ostrom 1990, 2007; Heikkila et al. 2011). In relation to CCA, the interactions and interplays between and among institutions are exhibited when an adaptation measure is introduced into an institutional setting. For example, integrating CCA into local land-use plans introduces climate change related data into the current local land-use planning system. Because institutional interplays are not one directional, and therefore involve mutual influences or effects, the institutions existing in both the land-use planning system and the climate change information system will interact (Cuevas et al., 2014). Hence, the institutions in both settings will be affected by the mainstreaming approach. To understand better these kinds of institutional relationships and linkages, this chapter investigated the influence of institutional nestedness and the roles of local and national governments in mainstreaming CCA in Albay, Philippines, and determined how these concepts relate to the institutional dimension of CCA.

7.3 Influence of institutional nestedness on the mainstreaming process

To explain how institutional nestedness affects the mainstreaming process, this section examines the following mainstreaming indicators–*institutional issues, availability of information, access to information, and credibility and reliability of information.*

7.3.1 Primary barriers: Institutional issues

The Albay experience identified the *institutional issues* indicator as the sole primary barrier in mainstreaming CCA-DRR into local land-use planning. This indicator is defined as the absence or presence of rule-based institutional questions or conflicts that inhibit the effective integration of CCA into local land-use planning. In-depth interviews revealed these key institutional issues as (1) fragmented national laws and regulations; (2) lack of guidelines for mainstreaming CCA-DRR into the local land-use plans; and (3) overlapping policy requirements (Chapter 6).

The difficulties experienced by the local government units (LGUs) in mainstreaming CCA stemmed from national institutional issues, the effects of which filtered down to the local environment. As discussed in Chapter 6, the first issue (i.e., fragmented national laws) involves the Disaster Risk Reduction and Management Act of 2010, a national law that mandates all LGUs to create the Local Disaster Risk Reduction and Management (LDRRM) officer position. This position is crucial as it assumes the tasks and responsibilities related to CCA-DRR at the city/municipal scale, including spearheading mainstreaming of CCA-DRR into the local land-use plan. Second, based on this Act, a national agency (i.e., Department of Budget and Management) identifies the fund source for the LDRRM officer's personal services requirements (i.e., salaries and compensations of government employees). The Department of Budget and Management Memorandum (March 15, 2012) names this fund source as the regular LGU fund. Third, usage of the LGU fund for personal services is restricted by the Local Government Code of 1991; thus, LGUs are constrained from hiring an LDRRM officer. Fourth, a national agency, the Civil Service Commission, is responsible for determining the standards for hiring the LDRRM officer; however, these standards were only issued in April 2014.

Hence, the disharmonies among rule-based national institutions and the inaction or the delayed actions of national organisational institutions were key sources of difficulties for the LGUs on this matter. Eventually, the issues were addressed through a collaborative effort of national government agencies, specifically, the issuance of a national directive (i.e., Joint Memorandum Circular No. 2014-1). The Memorandum Circular outlined the qualifications and standards for hiring an LDRRM officer, and the process that reconciled the provisions in

the Disaster Risk Reduction and Management Act and the Local Government Code. However, while this development may address the issue, its effects are yet to be determined (Chapter 6).

Similarly, the second issue, (i.e. lack of guidelines for mainstreaming CCA-DRR into the local land-use plan) was addressed by the "Supplemental Guidelines on Mainstreaming Climate and Disaster Risks in the Comprehensive Land-use Plan" released in the second quarter of 2014. This document was published by the Housing and Land-Use Regulatory Board (HLURB) and was an inter-agency "coordinated effort at the highest levels" (Interviews 2014). HLURB is a national government agency tasked to provide land-use planning guidelines in the Philippines by virtue of the Executive Order No. 648 (HLURB 2015). Although the HLURB has nine regional or field offices, the task of generating guidelines lies with the national office; the regional or field offices only implement and apply these guidelines in their respective areas (Republic of the Philippines 1987). Accordingly, the national office spearheaded the production of the said publication.

Finally, the difficulties arising from the third issue (i.e., overlapping national policies) can be addressed only at the national scale. In essence, there are approximately 30 local plans required from LGUs, by virtue of various national laws and regulations (Mercado 2011). These requirements overload the understaffed LGUs and prevent them from focusing on CCA-DRR related initiatives, including mainstreaming CCA-DRR into the local land-use plans (Interviews 2014).

Consequently, the resolve or failure of national institutions to act on matters affecting LGUs can either enable or constrain the ability of local governments to function efficiently in relation to CCA-DRR matters. As such, linkages between and among institutions across scales should be considered seriously, especially in CCA-DRR endeavours focused at the local scale (Cuevas et al. 2014).

7.3.2 Second level barriers: Availability and access to information

The information capacity mainstreaming indicators primarily deal with the integration of technical and scientific climate change and land-use data. The indicator scores revealed two substantial barriers in mainstreaming CCA-DRR into the local land-use plans. These included the level of available technical information on climate change (i.e., *availability of information*) and the degree of user access to the available climate change information (i.e., *access to information*) (Figure 11). The survey and interviews presented several concerns regarding the available climate change-related data including (1) limited data in terms of availability at the lowest governance scale (municipal/city, villages); (2) substandard

technical features (e.g., low resolution of hazard maps); (3) obsolete data; and (4) insufficient data to conduct total risk and vulnerability assessments. For example, the Mines and Geoscience Bureau, one of the primary sources of geohazard maps in the Philippines, provides 1:50,000 scaled maps to the LGUs in Albay (MGB Region 5 2014). However, the survey respondents clarified that 1:20,000 or 1:10,000 scaled maps are needed to produce a local land-use plan with CCA-DRR components.



Figure 11 Radar chart of the indicators used to assess mainstreaming of CCA-DRR in Albay, Philippines: Barriers or opportunities?

Notes: Scores for the availability of funds, experts, and human resources indicators did not capture the "true" conditions relating to these challenges, thus, answer choices relating to these indicators need to be modified. Grey area represents the region where challenges have been transformed into opportunities for mainstreaming CCA.

The problem with *access to information* predominantly stemmed from the lack of local users' knowledge of what information to use and how to access it. Electronic copies of the needed data were not available, thus causing access problems because users needed to personally collect the data from the agency source. In other instances, although digital data existed and was accessible from the internet, the data were in portable document formats (pdf files) and could not be edited to suit the needs of users. Finally, some data were available for a fee, which at times, LGUs were unable to afford (Interviews 2014).

Institutional nestedness is linked to the *availability and access to information* challenges through the organisational structure of the Philippine government. Government institutions are the main data sources of climate change information used by all LGUs, including Albay (refer next section). These institutions are part of the national government and are referred to as the national or central offices. They typically have field offices to help

with administrative tasks regarding field operations and engagements in local concerns (Republic of the Philippines 1987). In general, the functions of field government offices are governed by national rules, specifically the Executive Order No. 292, which present the key structures, procedures, functions, and rules of governance in the Philippines. Based on this rule, field offices are authorised only to implement laws, policies, plans, programs, rules, and regulations of the department or agency in the regional/local area; and to coordinate with the other regional offices of other departments, bureaus, agencies and LGUs in the area. Thus, the seat of authority of a government agency is with the national/central offices, although there are field offices tasked to address local affairs. Accordingly, for agencies that generate scientific data, the scientific tools, technologies, and the qualities of climate information available in the national offices dictate those existing in the field offices.

Some national rules also govern the national and field offices of specific government institutions. For example, the Department of Environment and Natural Resources Administrative Orders No. 95-23, No. 96- 40, and 97-11 instituted the functions of the central and regional offices of the Mines and Geoscience Bureau. Likewise, through the Executive Order 366, the Philippine Atmospheric, Geophysical, and Astronomical Services Administration, another primary source of climate change data, was re-organised into five technical divisions, five regional services divisions and two support divisions (PAGASA 2015).

In terms of access, the long-awaited Supplemental Guidelines is expected to improve the local users' familiarity with the data needs and their corresponding sources. For example, the Supplemental Guidelines identify the Mines and Geoscience Bureau; Philippine Atmospheric, Geophysical, and Astronomical Services Administration; and the Nationwide Operational Assessment of Hazards (a project of the Department of Science and Technology) as the chief sources of climate change information for land-use planning. This set of guidelines was a product of the national office of the HLURB and was a collaborative effort of other national institutions (HLURB 2014). In essence, the process of generating and accessing climate-related data in the Philippines is nested in a broader set of institutional arrangements characterised by a certain level of institutional hierarchy. As one respondent mentioned:

All development plans have guidelines on how they (plans) should be prepared... but basically, all guidelines originate from the top level (Interviews 2014).

7.3. 3 Opportunity: Credibility and reliability of information

The analysis of the *credibility and reliability of information* (i.e., level of trust and confidence of users in the scientific and technical information and their source/s) provided evidence that the efforts and resolutions implemented at the higher levels of government also enable potential barriers to become opportunities for local mainstreaming.

Government institutions, based on their respective mandates, are the primary sources of scientific data for official government documents such as local land-use plans. To illustrate, Presidential Decrees No. 78 and No. 1149, and Executive Order 128 assign the Philippine Atmospheric, Geophysical, and Astronomical Services Administration to provide the official climatological, atmospheric, geophysical, and astronomical data of the country, including information on natural disasters such as typhoons, floods, earthquakes, and tsunamis. Meanwhile, by virtue of Department of Environment and Natural Resources Administrative Order No. 97-11, geohazard mapping is among the decreed responsibilities of the Mines and Geoscience Bureau. As such, the bureau implements the Geohazard Mapping and Assessment program that determines the areas in the country that are vulnerable to geologic hazards (DENR 2014).

Accordingly, the Comprehensive Land-use Plan Guidebook Volume 1, one of the key guides for land-use planners, identifies the Mines and Geoscience Bureau as a data source for geological or environmental hazard areas (HLURB 2006). Similarly, Department of Environment and Natural Resources Memorandum Circular No. 2011-005 names the Philippine Atmospheric, Geophysical, and Astronomical Services Administration as the data source for climate projections; and the Mines and Geoscience Bureau and the Philippine Institute of Volcanology and Seismology for natural hazards.

Essentially, these rules established "accountability" for the data. This was the primary reason why "as long as data are official," local planners in Albay accepted the data provided by the government agencies. Clearly, institutional nestedness, that is, the institutional process created by national rule-based institutions, works in favour of transforming the issue of credibility and reliability of information from a potential barrier to an opportunity for mainstreaming CCA-DRR. Specifically, the climate change-related data were accepted by users, and local planners were willing to incorporate the information into the plans; hence the high score recorded for the credibility and reliability of information indicator (Interviews 2014).

7.4 Impacts of local initiatives on the barriers and opportunities for mainstreaming CCA-DRR into local land-use plans in Albay

This section elaborates the significance of local initiatives on the operationalisation of the mainstreaming endeavour by exploring the circumstances surrounding the *organisational cohesion*, *local leadership*, *knowledge and awareness*, *local government prioritisation*, *community support*, and *institutional incentive* indicators. It shows that local initiatives were key factors that helped transcend most of the challenges from potential barriers into opportunities for mainstreaming in Albay.

7.4.1 Third level barrier: Organisational cohesion

Organisational cohesion measures the degree of diversity or coordination among organisations/agencies engaged in local CCA. A low *organisational cohesion* score indicates a high degree of diversity among organisations; conversely, high scores signify that organisations are effectively coordinating with each other.

In Albay, *organisational cohesion* registered as a third-level barrier (Figure 11), denoting a certain level of coordination among organisations. However, the relationships among the organisations need to be improved for the indicator to be considered as an opportunity. The lack of cohesion among the institutions in Albay was manifested through the unfamiliarity of the field/local office heads of government agencies regarding their respective institution's role or function in local CCA-DRR activities. In these cases, organisations adhered to their mandated functions, and these functions typically did not align with each other. In other instances, these agencies were more concerned with their own CCA-DRR programs, thus, were unable to participate effectively in local CCA-DRR initiatives (Interviews 2014). The Provincial Government of Albay addressed this lack of organisational cohesion in land-use planning by providing the varying field offices with specific purposes and tasks. The Provincial Government of Albay issued local policies and regulations that established inter-institutional cooperation on CCA-DRR activities (Table 15). Thus, although organisational cohesion was a (third-level) barrier in mainstreaming CCA-DRR in Albay, it was not a serious threat.

7.4.2 Opportunity: Local leadership

The leadership indicator refers to the absence or existence of a "climate change champion" in the locality, and to the extent of the champion's influence on the behaviour of the community. The leadership indicator was an opportunity in Albay due to the presence of Governor Jose Salceda, the chief executive officer of Albay who championed climate change concerns in the province (Figure 11). Governor Salceda was very effective in influencing the behaviour of people and in initiating activities for CCA-DRR that the United Nations International Strategy for Disaster Reduction recognised him as a "Senior Champion" of CCA and DRR in 2010 and one of the "Champion for Making Cities Resilient" (PreventionWeb 2010; UNISDR 2012b).

Local Policy	Description	Effect
Provincial Executive	Included the Mines and Geosciences	Clarified the roles of
Order 2007-07	Bureau and the Environmental	institutions that generate
	Management Bureau in the Provincial	climate change-related data and
	Land-use Committee	engaged in climate change
Provincial Executive	Included the Philippine Institute of	adaptation-disaster risk
Order 2007-12	Volcanology and Seismology in the	reduction activities concerning
	Provincial Land-use Committee	integrating climate change
Provincial Executive	Included the Albay Public Safety and	adaptation into the local land-
Order No. 2008-03	Emergency Management Office in the	use plans
	Provincial Land-use Committee	-
Provincial Council	Urged the Provincial Government of	Encouraged collaboration
Resolution 2008-44	Albay-Centre on Initiatives and Research	among organisations
	on Climate Adaptation, Department of	concerning mainstreaming
	Agriculture, Department of Environment	climate change adaptation and
	and Natural Resources, and Department of	disaster risk reduction
	Agrarian Reform to work together and	
	train local government units in	
	incorporating climate-related hazards into	
	the local land-use plans	

Table 15 List of local policies in Albay that helped improve organisational cohesion

The Local Government Code of 1991 played a significant role in making the governor an effective climate change champion. This law provides LGUs in the Philippines extensive local autonomy in terms of powers, authority, responsibilities, and resources to govern their localities (Gonzales 1997; Serote 2004). Through this law, a governor is given substantial executive and legislative power and authority to undertake activities on concerns s/he deems significant to the general welfare of the province and its constituents (Republic of the Philippines 1991). Thus, as the head of Provincial Government of Albay, Governor Salceda initiated and has sustained a number of local CCA-DRR activities, policies, projects, and programs. As will be illustrated in the later sections, having a climate change champion was an opportunity in Albay that raised the *knowledge and awareness* of planners, decisionmakers, and the community on climate change concerns; elevated CCA-DRR as a *priority agenda* of the LGUs; and was instrumental in getting *community support* for CCA-DRR initiatives and creating *institutional incentives* for planners and decision-makers to mainstream CCA-DRR into the local plans.

7.4.3 Opportunity: Knowledge and awareness

The *knowledge and awareness* indicator refers to the degree of understanding of planners and community members about the implications of climate change (knowledge), as well as their familiarity or recognition of the existence of climate change (awareness). In Albay, the *knowledge and awareness* indicator was an opportunity (Figure 11), suggesting that planners, decision-makers, and the community members alike are aware of the concept of climate change and the need to adapt.

The local activities, programs, and policies initiated by the Provincial Government of Albay highly contributed to this level of climate change knowledge and awareness. In 2007 and 2009, the Provincial Government of Albay spearheaded the first and second National Conference on Climate Change Adaptation, respectively (Lasco et al. 2008). The first conference was held at the provincial capital (Legazpi City) and brought the concept climate change to the people in Albay. In addition, the conference produced the "Albay Declaration on Climate Change Adaptation," which was an initiative to promote CCA, mitigation, and enhanced resilience to hazards through information, legislation, and programs. This "declaration", in turn, became instrumental in the passage of the Climate Change Act (Salceda & Rangasa 2011). Although the second conference was conducted at the national capital (i.e., Metro Manila), the active participation of the Provincial Government of Albay in this initiative helped introduce climate change into the consciousness of the local planners and decision-makers in Albay (Interviews 2014). In 2008, the Provincial Government of Albay also organised an inter-faith forum in Legazpi City, which brought people from the Catholic, Islam, Iglesia ni Cristo, and Seventh Day Adventist faiths to discuss CCA and DRR (Salceda & Rangasa 2011; UNISDR 2012a).

Policy-wise, the Provincial Executive Order 2007-12-A established the Center for Initiatives and Research for Climate Adaptation (CIRCA) as a key institution on climate change. CIRCA is Albay's primary research institution for CCA and climate risk reduction; it also implements some of the province's major adaptation programs and projects (Lasco et al. n.d.). The Provincial Government of Albay also signed a Memorandum of Agreement with Bicol University to establish the Climate Change Academy (now known as the Climate Change Adaptation and Disaster Risk Reduction Management Training Institute), which became the province's main arm in enhancing and strengthening the theoretical knowledge and practical skills of major local stakeholders on climate and disaster risk assessment (PGA & CIRCA 2010). This was followed by the Provincial Executive Order No. 2011-02 that stipulated for the Climate Change Academy to hold environmental classes at the key university in the region (i.e., Bicol University) starting 2011.

Likewise, the Memorandum of Understanding between the Provincial Government of Albay-CIRCA and the University of the Philippines Los Baños sanctioned activities geared toward mainstreaming climate change into the academic curriculum of primary, secondary, and tertiary education in the province. With this, Albay became the first province to mainstream CCA in the education sector (Lasco et al. 2008).

These developments contributed to the relatively high level of knowledge and awareness to climate change in the province. Accordingly, this condition is expected to help community members understand the need for a local land-use plan with CCA-DRR concerns, and thereby support the endeavour (i.e., *community support* indicator); and planners and decision-makers see the benefits of mainstreaming CCA-DRR into the local land-use plans (i.e., *institutional incentive* indicator).

7.4.4 Opportunity: Local government prioritisation

The *local government prioritisation* indicator defines the CCA agenda within the general development priorities in the local government. CCA is a clear priority in Albay, as evidenced by the number of related local government programs and policies. The Provincial Council Resolution 2007-04 cemented this prioritisation by institutionalising CCA as a provincial policy. Similarly, through the Provincial Resolution 2007-24, which decreed Albay as "one of the first and pioneering prototype province that will adopt climate change adaptation," CCA was further embedded as a priority of the province. Institutionalising CCA as a local government priority was an opportunity to impart CCA into the political and social consciousness of the present and future generations of government officials and people in Albay (PGA 2007).

With CCA as a priority, the Provincial Government of Albay also provided the LGUs legislative means to mainstream CCA-DRR into the local land-use plans. For example, Provincial Resolution 2008-44 urged the Provincial Government of Albay-CIRCA and other government agencies like the Department of Agriculture, Department of Agrarian Reform, among others, to conduct trainings and workshops on how LGUs can revise their local land-use plans (i.e., incorporate climate change impacts and risks). As one respondent shared:

One time, the Albay provincial government requested the Mines and Geoscience Bureau Region 5 to train an LGU that was struggling to use the hazard maps in analysing land-uses. The two-day training was funded by the provincial government.

7.4.5 Opportunity: Community support and institutional incentive

The *community support* and *institutional incentive* indicators refers to the degree of public support and local community participation on CCA initiatives, and the degree to which the benefits from adaptation encourage actors to operationalise the mainstreaming approach, respectively.

In Albay, this community support for CCA-DRR is reflected in the participation of community members in various CCA-DRR-related local government activities such as the "Mural Painting Competition on Climate Change Adaptation" among colleges and universities; first "Mayon Trail Run," a marathon in celebration of the National Conference on Climate Change Adaptation first anniversary, among others (Lasco et al. 2008; Plan International 2010; Salceda & Rangasa 2011). Such support suggests the willingness of the community to accept CCA initiatives, including possible changes in land-uses.

When the respondents were asked to rate (from 1 to 5) the importance of mainstreaming CCA into the local land-use planning process, almost all gave the highest score of 5. Several respondents justified their answers based on economic reasons, such as the need to assure investors that the LGU offers investment areas that are safe, low risk, and not vulnerable to disasters. Some were concerned with minimising human casualties during future intensified typhoons, and indicated the need for safe and permanent settlement areas. Others rationalised their responses by considering climate change and DRR strategies as either opportunities or constraints in the context of land-use planning.

The actors involved in local land-use planning in Albay recognised the benefits of mainstreaming CCA-DRR into the local land-use plans, thereby minimising the possible resistance to or maximising the expected cooperation of planners, reviewers, and implementers in generating a local land-use plan with CCA-DRR components. This perception was highly influenced by the efforts of the local climate change champion. As one respondent stated:

In my opinion, about 80% of the planners in Albay have the mindset that integrating CCA-DRR into the local land-use plans should be prioritised. This is because of Governor Salceda who has been actively advocating for CCA-DRR. That is also why the people in Albay, including the planners and the Municipal Planning and Development Council, are aware of climate change.

7.5 Institutional settings and institutional networks in climate change adaptation

The preceding analyses demonstrate how the interplays between and among institutions existing within a hierarchical structure (i.e., institutional nestedness) influence the conditions surrounding the challenges in mainstreaming CCA, and indicate how the institutions at the local scale can define the circumstances relating to the barriers and opportunities for effective operationalisation of mainstreaming. These assessments imply that the challenges in mainstreaming CCA involve a network of interacting institutions and institutional arrangements that transcend governance scales. This network is comprised of the institutional setting wherein CCA is being integrated and the institutional settings where the mainstreaming challenges operate. Essentially, each mainstreaming challenge exists and persists in another institutional setting with its own working or functioning institutions and institutional arrangements (Figure 12).



Figure 12 Network of selected institutions, institutional settings, and institutional arrangements for mainstreaming climate change adaptation: Case of local land-use planning in Albay, Philippines

Notes: The lines depict the institutional arrangements governing the varying institutional settings. Thick lines connote arrangements that directly link institutions or institutional settings to the mainstreaming setting. The thin lines signify arrangements that are not directly associated with the mainstreaming setting.
DENR – Department of Environment and Natural Resources; PAGASA – Philippine Atmospheric, Geophysical, and Astronomical Services Administration

For example, the challenges related to the availability of information and access to information were assessed as second-level barriers, thereby suggesting that developing the information capacity of systems (i.e., ability to integrate climate change information into the information system of the planning and decision-making processes) is an important feat in mainstreaming. However, the producers, communicators, and translators of data are functioning in institutional settings outside land-use planning, and thus are governed by different sets of institutional arrangements. Specifically, the Mines and Geosciences Bureau and the Philippine Atmospheric, Geophysical and Astronomical Services Administration, key agencies that generate climate change-related data in the Philippines, are governed by national policies and regulations such as the Executive Order No. 292, the policy that presents the key structures, procedures, functions, and rules of governance in the Philippines. The Mines and Geosciences Bureau also adheres to the institutional arrangements stipulated in the Department of Environment and Natural Resources Administrative Orders No. 95-23, No. 96- 40, and No. 97-11. On the other hand, the Philippine Atmospheric, Geophysical and Astronomical Services Administration's powers, duties and functions are dictated by Presidential Decrees No. 78 and No. 1149, and Executive Order 128, and its organisational structure are influenced by Executive Order 366 (Figure 12).

Meanwhile, local leadership was assessed as an opportunity because of the existence of the climate change champion in the person of the provincial governor. The champion is linked to the socio-political status of the governor position and is effective because of the Local Government Code that provided the governor with the crucial institutional support to execute, formulate, and conduct local CCA policies, programs, and activities.

All these rule-based institutions that transformed the challenges into either barriers or opportunities for mainstreaming were not parts of either the land-use planning system or the CCA mainstreaming institutional setting. Likewise, although these rules affected the local setting (i.e., local land-use planning system), they originated from the highest level of governance (i.e., national). These conditions illustrate that the challenges in mainstreaming CCA encompass a chain of interactions or interplays within the network of institutional settings and this notion suggests that overcoming mainstreaming challenges necessitates broad institutional reforms that go beyond the institutional setting where CCA is to be integrated.

Previous studies have associated CCA with network analysis. For example, Pahl-Wostl (2009) emphasised the significance of network governance in analysing the adaptive capacities of systems; Juhola and Westerhoff (2011) investigated the role of networks in adaptation governance; Funfgeld (2015) examined how transnational municipal networks

facilitate local CCA; and Ingold and Balsiger (2015) studied how collaboration networks or network relations affect the sustainability of local CCA actions. The results of this research in Albay support this direction of analysis. Essentially, by contending that effective mainstreaming operationalisation necessitates an institutional perspective, and by proving this notion to be critical in analysis, the research has produced empirical evidence that links the mainstreaming approach to institutional networks. This then unlocks a wide range of theoretical and analytical research possibilities associated with network analysis—network theory¹ and complex systems² (Amarala & Ottino 2004; Newman 2011; Kim 2013)—that can be explored in future research on mainstreaming operationalisation.

7.6 Summary and conclusion

There is a void in CCA literature concerning the practical on-ground application of mainstreaming, especially regarding the challenges or barriers that hinder its effective operationalisation (Ekstrom et al. 2011; Uittenbroek et al. 2013; Lehmann et al. 2015). This chapter filled this gap by offering empirical evidence on the conditions surrounding the challenges in mainstreaming CCA into local land-use planning in Albay. It clarified how the some factors were evaluated as either barriers or opportunities for mainstreaming, thereby raising the level of understanding on the persistence of some challenges as barriers to CCA. Thus, this chapter *generated a more refined understanding of the operationalisation of mainstreaming in local CCA* (Objective 4).

The findings in this chapter support the ideas this research argued in Chapter 2, namely, that:

- the operationalisation of mainstreaming needs to go beyond the issues of climate change;
- (2) the mainstreaming process necessitates institutional analysis—a complicated process that needs to examine institutional arrangements and institutional interplays; and
- (3) the manner or degree by which the barriers affect the adaptation process depends on individual institutional settings.

¹ Network theory has its origins in graph theory and has developed in different areas of the social and natural sciences and led to current trends in theoretical and empirical network research. Several scholars now agree that the study of networks, including the exploration of their variety and dynamics, is an important key to understanding the evolution and stability of physical and social structures (Odella 2011).

² A complex system is a system with a large number of elements, building blocks or agents, capable of interacting with each other and with their environment (Amaral & Ottino 2004, p. 148).

Likewise, the results validate the mixed methodology devised in Chapter 3, especially the methodology's IAD-CCA framework that effectively examined the local mainstreaming process from an institutional perspective.

Based on the analysis in this chapter, two significant concepts were identified as influential in the transition of the barriers into opportunities for local mainstreaming, namely, institutional nestedness and active participation of local governments. Institutional nestedness suggests that operationalising the mainstreaming approach is dependent on other levels, and that the actions of local governments on climate change concerns are often nested in institutional frameworks and processes at higher levels (Corfee-Morlot et al. 2009; Measham et al. 2011). The circumstances surrounding the primary and second-level barriers in Albay support this notion. Thus, the process of mainstreaming CCA-DRR at the local scale is nested in a broader set institutional arrangements characterised by a certain level of institutional hierarchy. Essentially, planners encountered barriers created at multiple levels of government (Hamin et al. 2014).

This is an important insight because part of the system's ability to overcome barriers in mainstreaming CCA stems from knowing the origin of the barrier. Specific barriers caused by issues at the higher level imply that local actors would not have direct control over the matter (Moser & Ekstrom 2010). However, as evidenced by the "2007 Albay Declaration on Climate Change Adaptation", an Albay initiative that influenced the passage of the Climate Change Act in 2010, local governments can indirectly address the challenge that originated from higher levels.

The chapter also demonstrated how mainstreaming challenges (i.e., *credibility and reliability of information*) can be addressed by higher level (i.e., national) decisions. Likewise, it shows how other challenges (i.e., *availability of information* and *access to information*) were affected by hierarchical institutional structures. Thus, although the chapter does not recommend possible solutions for overcoming these specific barriers, it argues that realising the effects of institutional nestedness in creating barriers in mainstreaming CCA can help actors determine the suitable institutional support to transform these barriers into opportunities.

The Albay case also offers evidence of the critical role of local efforts in preventing the challenges from becoming barriers, and in turning them into opportunities for local mainstreaming. This makes local administration a key component in CCA. The numerous local policies, projects, programs, and initiatives implemented by the Provincial Government of Albay and spearheaded by the climate change champion, Albay Governor Salceda, raised the knowledge and awareness of the people on climate change matters. These efforts resulted in (1) a well-informed community that supports local government CCA-DRR initiatives; (2) knowledgeable planners and decision-makers who are incentivised to integrate CCA-DRR into the local land-use plans; and (3) local governments that prioritised CCA in their agendas. Essentially, the effective local climate governance advanced the participation, cooperation, and commitment of stakeholders in addressing climate change tasks and problems (Fröhlich & Knieling 2013).

A number of studies have emphasised the significance of local governments in addressing the challenges in CCA (Schreurs 2008; Burch 2010; Sharma & Tomar 2010; Pasquini & Shearing 2014). Similarly, others have highlighted the importance of the arrangements governing institutions at different levels—polycentric governance—in the context of CCA (Ostrom 2010; Corfee-Morlot et al. 2009). The findings of this chapter substantiate these notions. However, it concludes that operationalisation of mainstreaming is not an "either or" debate, nor is it a hierarchical versus decentralised decision-making dilemma. Local mainstreaming of CCA should account for both the institutional linkages across levels of government and the active participation of local governments. This is because the barriers to adaptation exist in different institutional settings that cut across governance levels. These settings become linked when an introduced adaptation measure interacts with the prevailing institutions functioning in these settings (Cuevas et al. 2014).

These findings imply that operationalisation of mainstreaming CCA is a multi-scale, multi-setting endeavour. This chapter contends that in designing strategies to address the challenges in mainstreaming CCA, analysts, planners, and decision-makers must understand that the challenges exist within a network of institutional settings, and that these challenges encompass a chain of institutional interactions or interplays within this network. Accordingly, overcoming these challenges necessitates broad institutional reforms that go beyond the institutional setting where CCA is to be mainstreamed. This line of thought opens a variety of analytical possibilities for mainstreaming research; in particular, scholars can explore network analysis for future research and investigation on mainstreaming operationalisation.

CHAPTER 8: CONCLUSION, POLICY IMPLICATIONS, AND FUTURE RESEARCH POSSIBILITIES

8.1 Introduction

This dissertation began by presenting evidence that although interest in mainstreaming CCA has been growing, and several countries have applied the approach at the national scale, there is still limited information regarding its on-ground operationalisation. This has been identified as a crucial setback because it is at the local scale where specific mainstreaming actions are implemented, thus suggesting that mainstreaming has not advanced from conceptualisation and planning to the effective practical application. Many mainstreaming studies have cited the barriers or challenges in operationalising the approach (Sharma & Tomar 2010; Nambi & Prabhakar 2011; Pasquini et al. 2013; Ayers et al. 2014; Uittenbroek et al. 2014) rather than reported on successful mainstreaming undertakings. These circumstances highlighted the gap between mainstreaming theory and operationalisation, and prompted the need for a critical analysis of the local mainstreaming processes. This research focused on mainstreaming CCA in land-use planning, and asked the questions "How can mainstreaming of climate change adaptation into local land-use planning be understood?", and "How can the challenges in the operationalisation of mainstreaming be overcome?"

Four objectives were formulated in this research to answer the above questions. The first three objectives were to (1) *explore the process of mainstreaming CCA, from its theoretical foundations to its operationalisation, with special interest in local land-use planning* (Chapter 2); (2) *determine an analytical framework and methodology that can examine effectively the challenges in mainstreaming CCA into land-use planning and generate metrics that can be used by planners and decision-makers in addressing these challenges* (Chapter 3); and (3) *analyse the state-of-play of and linkages between the challenges in mainstreaming CCA into land-use planning in Albay, Philippines, and determine how these challenges can be overcome* (Chapters 4, 5, and 6).

This final chapter synthesises the research findings and, along with Chapter 7, addresses the last research objective—*to generate a more refined understanding of the operationalisation of mainstreaming CCA*. Accordingly, Section 8.2 recalls the gap between mainstreaming theory and application that set the foundations for this research—which is the need to understand the institutional dimension on how to operationalise mainstreaming of CCA. Section 8.3 discusses the methodology that the thesis devised in order to answer the research questions. Section 8.4 highlights the key findings of the research analyses, focusing

on the nature of the mainstreaming challenges, and elaborates on the policy implications of these findings. Section 8.5 intensively examines the institutional dimension of mainstreaming CCA based on the research. Section 8.6 concludes with a statement on the challenges in mainstreaming CCA and the contribution of this research to the knowledge base on the operationalisation of mainstreaming. Lastly, Section 8.7 extends the chapter by presenting future research possibilities based on the findings from this research.

8.2 Gap between theory and application of mainstreaming climate change adaptation

General insight: Understanding the mainstreaming process entails examining the institutional dimension of the approach; thus, analysis of and planning for mainstreaming operationalisation necessitate an institutional perspective.

To address the first objective, the research investigated the theoretical underpinnings of the mainstreaming approach, tracked its path to popularity, and examined the status of its operationalisation. These activities entailed conducting intensive reviews of CCA literature, including the IPCC AR3, AR4, and AR5; handbooks and guidelines in mainstreaming CCA; peer reviewed papers related to mainstreaming CCA; government reports on country mainstreaming efforts; and documents relating to outputs of projects concerning mainstreaming of CCA (Chapter 2).

Two notable points were realised from the review. The first is that research interest in mainstreaming CCA has been increasing over the years, as outlined in the discussions in the IPCC reports released in 2001, 2007, and 2014. The concept of mainstreaming is not new and has been used in relation to education for handicapped children, gender issues, environment, disaster risk reduction (DRR), HIV/AIDS, and intercultural relations (Gupta & van der Grijp 2010; Olhoff & Schaer 2010); mainstreaming became associated with CCA only at the beginning of the 2000s. In fact, mainstreaming CCA as an adaptation approach was not mentioned in the IPCC AR3 in 2001. However, in 2007, the IPCC AR4 defined mainstreaming in several chapters (e.g., Chapters 2, 16, 17, and 18), advocated for the mainstreaming approach, and stated in its Technical Summary that "mainstreaming climate change issues into decision-making is a key prerequisite for sustainability" (Parry et al. 2007, p. 55). In 2014, the IPCC AR5 went beyond defining the term and presented explicit examples of how countries applied the approach at the national scale. For example, Wong et al. (2014, p. 390) stated that in Japan: "... coastal climate change adaptation has been mainstreamed into the framework of Coastal Disaster Management in the aftermath of the 2011 Tohoku Earthquake Tsunami." Likewise, the number of references on mainstreaming
CCA increased significantly in the reports from 2007 to 2014. As such, mainstreaming CCA has become a more prominent topic in adaptation literature over the last decade.

Document reviews also revealed that more countries have acknowledged that (1) the difficulties in achieving sustainable development cannot be overcome without considering climate change impacts; and (2) effective climate adaptation is unlikely without accounting for existing and future development actions (Olhoff & Schaer 2010; Schipper et al. 2010). Consequently, mainstreaming CCA has become a popular adaptation approach, especially in developing countries. For example, Bangladesh, Ethiopia, India, the Philippines, and Zambia, among others, have integrated climate change issues and concerns into their respective country development plans. The growing recognition of mainstreaming has been influenced substantially by external forces that advocate for its implementation. Examples of these external influences are the international funding mechanisms such as the Least Developed Countries Fund, Special Climate Change Fund, Multi-donor Trust Fund on Climate Change, Pilot Programme for Climate Resilience under the Climate Investment Fund, Global Environment Facility Trust Fund, and the Adaptation Fund (Measham et al. 2011; Lal et al. 2012). Likewise, mainstreaming CCA as a strategy has advanced due to the active advocacy of donor, bilateral, and multilateral agencies (Agrawala 2006; Olhoff & Schaer 2010; Lal et al. 2012). Accordingly, most (if not all) mainstreaming handbooks and guidelines that prescribe its operationalisation have been produced by international organisations such as the Asian Development Bank, OECD, the World Bank, and the United Nations agencies (i.e., UNEP, UNDP), among others.

Based on these handbooks and guidelines, the operational procedures (i.e., vulnerability assessments, risk screening, impact modelling), the tools, and techniques to accomplish the mainstreaming task focus on the climate-related aspect of adaptation. Specifically, they concentrate on risk reduction and environmental impact assessment procedures (Klein et al. 2007). For example, the Assessment and Design for Adaptation to Climate Change–A Prototype Tool (ADAPT) of the World Bank is primarily a risk-screening tool. Meanwhile, the Opportunities and Risks from Climate Change and Disasters of the UK Department for International Development concentrates on identifying activities at high risk to climate change and on determining those that provide opportunities for vulnerability and risk reduction (Gigli & Agrawala 2007). Some, like the Climate Vulnerability and Capacity Analysis (CVCA) of CARE International and Community-based Risk Screening Tool-Adaptation and Livelihoods by the International Institute for Sustainable Development, went beyond vulnerability, risk, and environmental assessments; however, they still focused on climate change-related issues.

There have been significant advancements in mainstreaming CCA from its conceptualisation to its recognition as an important adaptation approach, and to its actual application to address climate change. With the encouragement of external influences (i.e., funding mechanisms, international organisations), several developing countries have incorporated climate change into their development plans, thereby setting the direction of their respective nations' linked CCA and sustainable development agendas. Similarly, there have been substantial developments in relation to tools and mechanisms to integrate climate change into the planning and decision-making processes. However, the progress of mainstreaming as an adaptation approach has been more of a top-down path, with less focus on the elements that can affect the actual integration process on-ground. That is, the existing operational procedures and tools neglect the most important aspect of mainstreaming, which is the institutional dimension of adaptation (Chapter 2).

This is a major gap in CCA literature primarily because climate change is a wicked problem that is complex, ambiguous, ill-defined, unpredictable, intractable, and multifaceted (Lazarus 2010; FitzGibbon & Mensah 2012; Head & Alford 2015). Consequently, CCA has been referred to as a "wicked problem par excellence" (Termeer et al. 2013, p. 27). The traditional methodologies (i.e. scientific and technical) and available tools in policy analysis are ill-equipped to address CCA issues and concerns. This is so because CCA involves multilevel interventions spanning across households, communities, governments, NGOs, industries, and different sectors at several scales (from local to regional, national and international) (Tol et al. 2000; Perry 2015). Likewise, it warrants changes or adjustments in behaviours and in value systems (Lorenzoni et al 2007; Pettengell 2010; Berrang-Ford et al. 2011; Hamin et al. 2014). As such, CCA, especially mainstreaming CCA, needs to be addressed through institutional means (Jentoft & Chuenpagdee 2009; Rodima-Taylor et al. 2011; Perry 2015).

Fundamentally, the whole concept of mainstreaming (i.e., synergy of climate change and sustainable development goals and agenda; designing new or re-designing existing planning, policy-making and decision-making structures) is an institutional concern (Young 2002; Ayers & Doman 2010). Particularly, the institutional aspects of mainstreaming are incorporated in the (1) guidelines for applying the adaptation measure; (2) governance arrangements accompanying the mainstreaming efforts; (3) institutional conditions where the mainstreaming policy is implemented; and (4) varying institutional levels or scales where the approach is applied (Brondizio et al. 2009; Theesfeld et al. 2010; Bettini et al. 2012). Therefore, since the settings where CCA will be integrated already exist within functioning institutions and institutional arrangements, mainstreaming CCA essentially is an approach dominated by institutional changes, institutional transformations, and institutional reforms. Consequently, because the institutional aspect of mainstreaming CCA has been neglected, very few reports of successful on-ground application of the approach were identified during the course of this research. Instead, studies have cited the barriers or challenges in mainstreaming operationalisation.

8.3 Methodology for examining the challenges in mainstreaming climate change adaptation

General insight: Mainstreaming operationalisation necessitates a methodology that focuses on the challenges in applying the approach, an analytical framework that can examine the mainstreaming process from an institutional perspective, and metrics that can translate climate change adaptation from an abstract to a more concrete concept.

To address the second objective, the research investigated literature from the fields of institutional analysis, methodological procedures, and CCA. The previous section established the need to focus on the institutional dimension of mainstreaming in operationalisation; hence, the review of institutional literature. Meanwhile, because information is lacking on how mainstreaming CCA can be applied on-ground, which consequently highlights a knowledge deficit regarding the mainstreaming process (Lal et al. 2012; Ayers et al. 2014; Picketts et al. 2014), methodological literature was also examined in this research. Finally, CCA literature was explored to determine how the overall methodology of the research should be designed.

The existing literature on mainstreaming CCA has identified a number of setbacks in its knowledge base. First is the lack of understanding regarding the barriers or challenges that slow or delay the effective operationalisation of the approach (Uittenbroek et al. 2013; Revi et al. 2014; Lehmann et al. 2015). Second is the absence of metrics to determine the state-of-play of the adaptation effort, to evaluate and assess adaptation outcomes, and to measure adaptation progress and its effectiveness (Berrang-Ford et al. 2011; Mimura et al. 2014; Noble et al. 2014). Lastly, climate change research has lagged behind the interdisciplinary nature of the problem in terms of research cooperation, citation, and the methodologies applied (Bjurstrom & Polk 2011). Hence, to have effective interdisciplinary communication and cooperation in climate change research, quantitative and qualitative researchers need an avenue for collaboration (Nielsen & D'haen 2014).

To address these setbacks, the research designed a mixed methodology that (1) focused on examining the challenges in mainstreaming CCA; (2) had an analytical framework that was able to investigate institutional settings; and (3) generated indicators for assessing mainstreaming endeavours. The methodological design incorporated a case study in order to answer the research question within a real-life context (Scholz & Tietje 2002; Yin 2014). In particular, the research's four-stage methodology (1) included a mixed method that utilised document reviews, interviews, a survey, and key informant consultations (i.e., triangulation by data method) as the main data sources; (2) used the modified Institutional Analysis and Development (IAD) framework as the primary analytical and data collection guide; and (3) employed the scorecard approach to generate quantitative data and indicators. The research methodology was successfully applied to the case of mainstreaming CCA into the local land-use planning system in Albay, Philippines (Chapter 3).

The four-stage mixed methodology was a systematic and practical process. Each stage in the methodology produced its own output, and each individual output significantly contributed to adaptation literature. Stage 1 developed the IAD-CCA framework; Stage 2 generated the quantitative mainstreaming indicators; Stage 3 devised the varying levels of severity by which the challenges (as represented by the indicators) impact the mainstreaming process; and Stage 4 produced the in-depth qualitative analyses of the challenges in mainstreaming CCA. Likewise, each stage was an important part in a chain of actions within the methodology, in which an output of one stage was an input in another (stage).

To illustrate, the research modified the IAD in Stage 1, thus, transforming it into the IAD as applied in mainstreaming CCA research (i.e., IAD-CCA). The IAD was the most suitable framework for this CCA research because it is designed specifically to examine institutional settings (Ostrom 2011). Hence, it can analyse the setting where CCA is to be mainstreamed and is equipped to examine the complexities in institutional arrangements and institutional interplays (Jordan & O'Riordan 1997; O'Riordan & Jordan 1999; Young 2002). Furthermore, the IAD variable that refers to the biophysical conditions of the system can represent the impacts of climate change concerns in that setting (McGinnis 2011). The IAD has a methodical analytical process that can help users organise vast amount of data (Koontz 2006; Dick & Meinzen-Dick 2011). This feature was very useful in designing the data collection activities of the CCA research. Also, the IAD is a framework that has been tested and applied successfully in a variety of institutional conditions and to an extensive range of problems and concerns, including CCA (Koontz 2006; Oberlack & Neumärker 2011).

Most importantly, the IAD has a flexible design which allowed the research to address the framework's limitation in examining institutions under the CCA context. Particularly, this research replaced the evaluation criteria of the IAD with factors that influence the effective operationalisation of mainstreaming CCA (i.e., 20 mainstreaming challenges). The research summarised the mainstreaming challenges into three capacity classifications (i.e., *institutional, information,* and *resource* capacities). Meanwhile, the concept of institutions as rules, social structures, and organisations (Cuevas et al. 2014) was incorporated in the mainstreaming challenges under the institutional capacity classification. This research considered the evaluation criteria to be key variables as they guide the users in (1) assessing the patterns of interactions of institutions; (2) evaluating which outcomes are acceptable and which need improvement; (3) analysing how the current institutional arrangements constrain or facilitate desired outcomes; and (4) formulating ideas on how to attain the preferred outcomes (McGinnis 2011).

In Stage 2, the list of mainstreaming challenges guided the survey design. The survey was conducted among the key actors in the local land-use planning system in Albay and the representatives of national government and non-governmental organisations engaged in local mainstreaming of CCA. Specifically, each survey question represented a challenge in the IAD-CCA evaluation criteria. The scorecard approach was effective in converting these challenges into quantitative indicators (Chapters 4 and 5). The three-choice design of the survey questions was simple and straightforward. The respondents easily answered the survey, and thus the researcher processed the survey data without difficulty. Essentially, the scorecard approach was easy to understand and was readily communicated to or interpreted by users with varying backgrounds. Most importantly, it was applied at the local scale successfully. Accordingly, the resulting indicators have the potential to be used as benchmarks for determining the status of the mainstreaming challenges and they can be updated for timeliness and comparability across time.

The partial results of the mainstreaming indicators were generated in the field, and the indicator scores guided the line of questioning during the interviews in Stage 3. To illustrate, because the institutional issues indicator was assessed as a primary barrier, interview questions were directed at exploring these issues in greater depth. Consequently, the following were identified as the factors that constrained the capacity of local governments in Albay in mainstreaming CCA into the local land-use plans (1) fragmented national laws; (2) overlapping policy requirements; (3) lack of guidelines for mainstreaming CCA into the local land-use plans; and (4) political concerns.

The additional issues raised during the interviews were further investigated in Stage 4 through the document reviews (i.e., national and local laws and regulations, government memoranda, local government reports, and other related studies) and consultations with key

informants. For example, a representative from the Department of Budget and Management was interviewed to determine how the Joint Memorandum Circular No. 2014-1 issued in April 4, 2014 could address the problems created by the fragmented national laws. This problem speaks of the conflict between the Disaster Risk Reduction and Management Act and the Local Government Code regarding the budgetary requirements for hiring a local disaster risk reduction and management (LDRRM) officer (Chapter 6).

Also in Stage 4, the IAD-CCA framework helped organise the analysis of the data collected. To illustrate, based on the quantitative results of the evaluation criteria, the institutional issues indicator was prioritised in the analysis. One of the specific concerns related to this indicator was the fragmented national laws. At this point, the aim of the analysis was to determine the conditions surrounding this problem. The next step was to identify the rule-based institutions (i.e., policies) involved in the fragmented national laws issue (e.g., these were the Disaster Risk Reduction and Management Act and the Local Government Code). The specific institutional arrangements and institutional actors in these two policies that affect the mainstreaming process were examined. In particular, by virtue of the Disaster Risk Reduction and Management Act, the local government units in the Philippines are mandated to create the LDRRM officer position. This person is the primary local government staff concerned with CCA and DRR activities of the local government, including mainstreaming of CCA into the local land-use plans. However, hiring of the officer is subject to the rules and regulations of the Department of Budget and Management concerning budgetary source. The department, on the other hand, directed that the salary of the officer to be governed by the provisions of the Local Government Code on the rules regarding the local government budget (Chapter 6). Essentially, using the IAD-CCA structure as guide, (1) the interplays between and among the rule-based and institutional organisations connected to the mainstreaming of CCA into local land-use planning were examined; and (2) the existing and introduced institutional arrangements governing the actions of these institutions were analysed. Overall, the mixed methodology was applied effectively in practice; and it generated results that can assist analysts, planners, and decision-makers determine the state-of-play of the challenges in mainstreaming CCA and make informed decisions for overcoming these challenges.

8.4 Nature of challenges in mainstreaming climate change adaptation: Assessing linkages and state-of-play

General insights:

- (1) The mainstreaming challenges are interconnected at varying levels of intensity.
- (2) The challenges exist within a certain spectrum, with the barriers and opportunities for adaptation representing the extreme ends of this spectrum.
- (3) The barriers can affect the mainstreaming process at varying degrees of severity.
- (4) The barriers can be overcome and can transcend into opportunities for mainstreaming CCA.

Several researchers who have studied CCA measures and efforts have agreed that the barriers exist and that they are impeding the progress and success of adaptation efforts (Amundsen et al. 2010; Burch 2010; Biesbroek et al. 2011, 2013). One of the pressing questions addressed by existing research in this area is "What are the barriers to adaptation?" Accordingly, the knowledge base on identifying the barriers has grown. However, the current research need has moved beyond merely recognising and classifying these barriers. This is especially true in mainstreaming CCA since it deals with planning and policy- and decision-making processes. Planners and policy-makers need to understand fully these barriers and the circumstances surrounding them (barriers) in order to formulate effective plans and policies. Therefore, CCA research needs to (1) determine the nature of these barriers; (2) establish the interconnections and interdependencies among the barriers; (3) understand why and how barriers arise and endure; and (4) determine how the barriers can be overcome (Amundsen et al. 2010; Burch 2010; Moser & Ekstrom 2010; Biesbroek et al. 2011; Eisenack et al. 2014; Waters et al. 2014).

This thesis generated quantitative data that allowed for comprehensive data analysis to be conducted (i.e., ranking, descriptive statistics, etc.) on the challenges in mainstreaming CCA. Since this research was concerned with understanding the relationships among the barriers, correlation analyses were performed among the mainstreaming challenges. For example, based on the resulting correlation coefficients (r) at the provincial scale, the highest frequencies of interconnections (i.e., $r \ge 0.50$) among the mainstreaming challenges stood at *stability of funds* (12 interconnections); *access to funds* and *institutional incentives* (11); and *availability of funds*, *local government prioritisation*, and *knowledge and awareness* (10). In comparison, the highest frequency of interconnections at the city/municipal scale was the *knowledge and awareness* challenge (9). Also, the correlation coefficients provided a concrete idea of the kinds of relationships that existed among the challenges. To illustrate, at

the provincial scale, while both the challenges pertaining to the *knowledge and awareness* (r = 0.740) and *institutional incentive* (r = 0.847) were linked to the *stability of funds*, the latter relationship was stronger than the former (Chapter 4).

Although the correlation analysis was context-specific, that is, the relationships among the challenges depended on the circumstances or conditions surrounding the mainstreaming process in Albay, Philippines, the results suggest that:

- the mainstreaming challenges are interconnected to each other and that the interconnections among the mainstreaming challenges exist at varying levels of intensity;
- (2) different types of relationships among the challenges may exist (i.e., dual or tripartite relationships); and
- (3) tripartite relationships among the mainstreaming challenges can be significant factors in formulating strategies to overcome the challenges.

These sets of information have important planning and policy implications. In the case of Albay, Philippines, planners and decision-makers can use these results to (Chapter 4):

- (1) explore the following challenges in designing a strategy with potential extensive effects on the mainstreaming process: *stability of funds, access to funds, institutional incentives, availability of funds, local government prioritisation,* and *knowledge and awareness*;
- (2) investigate whether developing *resource capacity* and *knowledge and awareness* simultaneously can be an optimal strategy to an effective mainstreaming process;
- (3) consider the very strong association between *local leadership* and *local government prioritisation* of CCA in devising viable approaches to mainstreaming CCA; and
- (4) exploit the tripartite relationships of the mainstreaming challenges (i.e., *local leadership-local government prioritisation of CCA-commitment to CCA*) in planning for mainstreaming CCA.

Because of the case-study facet of the mixed methodology, similar analytical possibilities can be gathered in other studies, although results are likely to present different combinations of relationships among varying types of mainstreaming challenges. A case study is a "detailed examination of a single example" that can be used "in the preliminary stages of an investigation to generate hypotheses" (Flyvbjerg 2011, p. 301). As presented in Chapter 3, the Albay case is a "critical case" that has a strategic importance in relation to the

problem of local mainstreaming of CCA. Thus, the hypotheses gleaned from the case "could be expected to be valid for all" (Flyvbjerg 2006, p. 226) or at least to a wide range of cases. Chiefly, the Albay case study provides concrete, context-dependent knowledge regarding the local mainstreaming process, and offers a methodology that can be replicated under different conditions and circumstances (Flyvbjerg 2006, 2011). Accordingly, the results of other cases can validate the hypotheses advocated by this case study.

The quantitative mainstreaming indicators developed in this research can also help other researchers investigate their own cases. As discussed in Chapter 3, each mainstreaming indicator represented a challenge in the IAD-CCA evaluation criteria. Although the process by which these indicators were computed can be improved (i.e., three of the 20 indicators did not reflect the actual on-ground conditions), in general, the indicators measured effectively the severity of the impacts of the challenges to the mainstreaming process. Specifically, at all scales (i.e., national, provincial, city/municipal), the *institutional issues* indicator was assessed as the primary barrier in operationalising the approach. Interviews revealed these issues to be (1) fragmented national laws; (2) overlapping policy requirements; (3) lack of guidelines for mainstreaming CCA into the local land-use plans; and (4) political concerns. The indicator scores also implied that the *availability* and *access to information* were among the major challenges that should be prioritised. Conversely, the *credibility and reliability of information* and *stability of funds* were considered as opportunities to the mainstreaming endeavour.

The other indicator scores reflected the national versus local perspectives, that is, the national respondents generally reacted to the question in terms of the national institutions involved in climate change and land-use planning. On the other hand, the local respondents (i.e., provincial and city/municipal) evaluated the question in relation to the local institutions in Albay. This case applied to the *organisational cohesion*, *local government prioritisation*, and *institutional incentive* indicators. These results provided direction for additional and more intensive inquiries. To illustrate, organisational cohesion was a primary barrier at the national scale, but was a third-level barrier in Albay. As discussed in Chapter 5, the investigation revealed that the lack of organisational cohesion was an "inherent problem in the Philippine government system." Government agencies or organisations in the Philippines design their goals and implement their activities based on their respective key result areas (KRAs). These KRAs are set individually and independently from one organisation to another; thus, the KRAs often do not align. Under this condition, the programs, projects, responsibilities, and tasks of varying organisations tend to be disharmonised even within a common agenda, such as CCA-DRR. At the local scale, this lack of organisational cohesion was translated as the

unfamiliarity of government field/local offices with the function of their institution in local CCA-DRR activities. Local offices follow their organisation's KRAs and mandated functions, and not the local government's CCA-DRR agenda.

In Albay, *organisational cohesion* was a third-level barrier, suggesting better conditions in the province as compared to the national scale, especially in land-use planning. This is because the Provincial Government of Albay issued local policies and regulations that clarified the roles of the varying field offices in terms of mainstreaming CCA into the local land-use plans. Particularly, the provincial government issued local regulations that included key agencies that generate climate-related data (i.e., Mines and Geosciences Bureau and the Environmental Management Bureau) and the main agency responsible for DRR in the province (Albay Public Safety and Emergency Management Office) among the members of the committee that reviews and approves the local land-use plans (i.e. Provincial Land-Use Committee). In essence, the provincial government created institutional mechanisms that linked the organisations involved in land-use planning, climate change, and CCA-DRR. These mechanisms clarified their individual roles and functions toward a common CCA-DRR agenda—mainstreaming CCA into local land-use plans (Chapter 5).

However, some variations in the assessments across scales reflected the differences in the conditions in Albay (both provincial and city/municipal) and other LGUs in the country, especially in relation to the following: knowledge and awareness, leadership, and community support. These indicators were assessed as opportunities at the provincial and city/municipal scales, but were barriers at the national scale. Similarly, the commitment to CCA and access to funds were opportunities at the provincial scale, but were second-level barriers at the national scale. These results suggested that there were conditions in Albay that separated the province from the other local government units in the Philippines. Further investigation showed that the primary cause of this difference was the existence of a climate change champion in Albay (i.e., *leadership* indicator), in the person of the provincial chief executive, Governor Jose Clemente Salceda. The governor was instrumental in turning a number of challenges from possible barriers into actual opportunities (i.e., barriers that have been overcome or factors that have positive influence on the mainstreaming process) for mainstreaming CCA in Albay. To illustrate, under the governor's leadership, CCA was institutionalised as a priority agenda in the provincial government, resulting in an assessment of the local government prioritisation of CCA as an opportunity at both provincial and city/municipal scales. The very strong correlation between the two mainstreaming challenges verified the relationship between the two indicators, and the qualitative assessments established the direction of this relationship. That is, *leadership* or existence of a climate change champion was the cause, and the high *local government prioritisation of CCA* was the effect.

This research contends that knowing the extent of a barrier's impact on the mainstreaming process can assist planners and decision-makers in prioritising the barriers to address. This is significant especially in the local context where local governments face a number of governance and development constraints. Thus, in the Albay case, solving the institutional issues should be the priority. More importantly, knowing and understanding the nature of the challenges will help in formulating strategies to overcome these challenges. Although the results of this research are context-specific, some generalisations apply as follows (Chapters 4 and 5):

- (1) mainstreaming challenges exist within a certain spectrum, with the barriers and opportunities for adaptation representing the extreme ends of this spectrum;
- (2) depending on the conditions surrounding the barriers, they (barriers) can affect the mainstreaming process at varying degrees of severity;
- (3) a challenge can be a significant barrier or opportunity to another challenge; and
- (4) when the barriers are overcome, they can become opportunities for mainstreaming CCA.

8.5 Institutional dimension of mainstreaming climate change adaptation

General insights:

- (1) The institutional mechanisms to support mainstreaming actions need to be in place for effective operationalisation.
- (2) Active participation of local governments helps transition the barriers into opportunities for local mainstreaming.
- (3) Mainstreaming CCA involves a network of interacting institutions and institutional arrangements that transcend across governance scales.
- (4) The challenges in mainstreaming CCA encompass a chain of interactions or interplays within the network of institutions.

This research is founded on the notion that mainstreaming operationalisation is an institutional concern that necessitates an institutional analysis. Based on this principle, the institutional setting where CCA is being mainstreamed (i.e., local land-use planning) was examined using the IAD-CCA framework; the evaluation criteria of the framework outlined the structure of the examination. Four major conclusions regarding the mainstreaming process were gathered from the analysis as follows:

- institutional mechanisms to support the mainstreaming process are necessary for effective operationalisation;
- (2) local governments can significantly contribute to CCA policy formulation and implementation; thus, the active participation of local governments help transition the barriers into opportunities for local mainstreaming;
- (3) on-ground application of mainstreaming involves a network of interacting institutions and institutional arrangements that transcend across governance scales; thus, mainstreaming operationalisation should account for the institutional linkages across levels of government; and
- (4) the challenges in mainstreaming CCA encompass a chain of interactions or interplays within the network of institutions; hence, overcoming these challenges necessitates broad institutional reforms that go beyond the institutional setting where CCA is to be integrated.

First, the institutional mechanisms to support mainstreaming actions need to be in place for its effective operationalisation. This notion was confirmed through the assessment of the *institutional issues* indicator as a primary barrier to the effective mainstreaming operationalisation in Albay and the Philippines as a whole. In particular, these issues were fragmented national laws, overlapping policy requirements, lack of guidelines for mainstreaming CCA into the local land-use plans, and political concerns (Chapter 6).

To illustrate, when the Climate Change Act of 2009 and the Disaster Risk Reduction and Management Act of 2010 mandated for climate change and DRR to be mainstreamed in the government development planning and policy-making processes, plans needed to be revised. In the case of land-use planning, this involved the existing Comprehensive Land-Use Plan. However, the two introduced laws also required additional new plans to be generated. The Climate Change Act required the development of a Local Climate Change Action Plan, while the Disaster Risk Reduction and Management Act mandated the Local Disaster Risk Reduction and Management Plan. Meanwhile, by virtue of the Local Government Code of 1991 and the Urban Development and Housing Act of 1992, local governments need to produce not only the Comprehensive Land-Use Plan (local land-use plan) but also a Local Shelter Plan. Likewise, the Agriculture and Fisheries Modernization Act of 1998 called for a Strategic Agricultural and Fisheries Development Zones Plan. Hence, local government units in the Philippines are obligated to produce multiple plans based on varying laws, regulations, memorandum circulars, and other rule-based institutions. Consequently, the revisions and modifications needed to mainstream climate change into the local land-use plans brought additional pressure and stress to the already under-staffed and under-resourced local government units. In essence, when the mainstreaming process was operationalised, institutional mechanisms to address the overlapping policy requirements were neglected. There was no institutional rule that directed the merging of related plans into a single all-inclusive plan with CCA-DRR components; nor were there amendments to the existing rules to avoid duplication of planning effort. Most especially, the impacts of designing the Local Climate Change Action Plan and Local Disaster Risk Reduction and Management Plan on the local government resources (i.e., additional burden) and the local planning system (i.e., repetitive plans if CCA-DRR is mainstreamed) were overlooked.

Another important issue was the lack of guidelines for mainstreaming CCA into the local land-use plans. When the Climate Change Act and the Disaster Risk Reduction and Management Act were enacted in 2009 and 2010, respectively, local government units were then required to produce updated local land-use plans with CCA-DRR components. However, the operational procedures for mainstreaming CCA-DRR (i.e., Supplemental Guidelines on Mainstreaming Climate and Disaster Risks in the Comprehensive Land-use Plan) were only released in early 2014. Thus, during the intervening years, local government units were unsure of how to fulfil their obligations as required by the legislation. As such, in 2012, the Regional Development Council (RDC) in General Santos City, Philippines filed a resolution requesting the National Economic and Development Authority, the main socio-economic planning agency in the Philippines, to "fast track the completion of the reference manual on mainstreaming disaster risk reduction/climate change adaptation in the comprehensive land-use plans" (RDC XII 2012, p. 1).

The assessments on the *leadership* indicator also illustrated the significance of institutional-support mechanisms for effective mainstreaming of CCA. Likewise, it revealed the second conclusion—that active participation of local governments help transition the barriers into opportunities for local mainstreaming. The *leadership* indicator was an opportunity for mainstreaming CCA in Albay due to the existence of the climate change champion in the person of the provincial chief executive, Governor Salceda. The position of Governor Salceda provided him the authority and power to be a positive influence in the operationalisation of the mainstreaming endeavour. In the Philippines, local chief executives have extensive administrative and legislative authority by virtue of the Local Government Code. Accordingly, this legislation offered institutional support, specifically, the autonomy needed by the provincial government to execute CCA-DRR activities, policies, and regulations for its jurisdiction. For example, Section 465 of the Local Government Code authorises the provincial governor to (1) initiate and propose legislative measures to the

provincial council; (2) issue executive orders for the enforcement and execution of laws; (3) exercise general supervision and control over all programs, projects, services, and activities of the provincial government; and (4) initiate and maximise the generation of resources and revenues, and apply the same to the implementation of development plans, program objectives and priorities. Hence, the climate change champion in Albay was able to implement and influence a number of CCA-DRR initiatives. Moreover, under the governor's leadership, the Provincial Government of Albay institutionalised CCA as a priority agenda by virtue of Provincial Resolutions 2007-04 and 2007-24. Both local regulations aimed to influence the political and social consciousness of the people in the government, private sector, and communities in Albay with regard to CCA (PGA 2007; Lasco et al. 2008). Such policies, along with many others, established a legislative framework for CCA in the province. Consequently, the local government prioritisation indicator was evaluated as an opportunity at both provincial and city/municipal scales. This condition (i.e., local leadership and *local government prioritisation* as opportunities) (1) raised the knowledge and awareness of planners, decision-makers, and the community on climate change concerns; (2) positively influenced the commitment of the local governments to CCA-DRR initiatives; (3) placed CCA-DRR among the priority agenda of the local governments; (4) helped gain community support for CCA-DRR; and (5) provided institutional incentive through motivating planners and decision-makers to mainstream CCA-DRR into the local plans.

The third conclusion relates to institutional nestedness, a concept that involves hierarchical relationships and rules at multiple levels of governance (Ostrom 1990). Institutions are typically nested in institutional frameworks and processes at higher level; thus, on-ground application of mainstreaming involves a network of interacting institutions and institutional arrangements that transcend across governance scales. For example, based on a number of national legislation, government institutions are the primary sources of scientific data for official government documents such as local land-use plans. These pieces of legislation turned the *credibility and reliability of information* challenge into an opportunity in Albay because the information provided by "official" sources were believed and accepted by planners and decision-makers. This acceptance was built on the principle that official data sources can be held accountable for the data produced and disseminated, and such accountability cannot be imposed on unofficial data sources.

The impacts of institutional nestedness are illustrated also by the circumstances surrounding the *institutional issues* indicator. Fundamentally, the complications that the local government units in Albay experienced with mainstreaming CCA stemmed from the effects of the actions of national institutions, which trickled down to the local environment. To

illustrate, the Disaster Risk Reduction and Management Act stipulated that local governments should create the position of the LDRRM officer in their respective units. However, there had been no national directives, regulations or other formal institutional rules that addressed the possible conflicts that may arise from this provision. As explained in Chapter 6, the rule that governed the source of the officer's salary is the Local Government Code of 1991; Section 325 of the Code stated that the personal services (salaries) of local government personnel should not exceed 45%–55% of the local government funds. However, this rule posed a significant problem to local governments because most (if not all) had already reached their respective budgetary ceilings for personal services. This meant that no budget was available to hire a new officer. This was only addressed in early 2014 when the a Technical Working Group³ formulated the implementing guidelines for establishing the Local Disaster Risk Reduction and Management Offices in local government units via the Joint Memorandum Circular No. 2014-1. During the years after the law was implemented and before the implementing guidelines were issued, the tasks and responsibilities of the LDRRM officer had been designated to existing regular and permanent staff, without additional compensation, financial or otherwise. Thus, the disharmonies among rule-based national institutions and the inaction or the delayed actions of national organisational institutions were key sources of difficulties for the local government units on this matter.

The last conclusion advances the understanding of the institutional dimension of mainstreaming CCA. That is, each mainstreaming challenge exists and persists in its own institutional setting, with its own working or functioning institutions and institutional arrangements. These institutional settings are neither part of the land-use planning nor climate change adaptation systems (Figure 12 in Chapter 7). For example, in relation to the availability of information and access to information, the actions of producers, communicators, and translators of data were governed by sets of institutional arrangements not directly linked to the mainstreaming process. In particular, the Mines and Geosciences Bureau and the Philippine Atmospheric, Geophysical and Astronomical Services Administration, producers of climate change-related data, were governed by national rules such as Executive Order No. 292 that outlined the key structures, procedures, functions, and rules of governance in the Philippines. Also, the Mines and Geosciences Bureau followed the institutional arrangements specified in the Philippine Mining Act of 1995, and the Department of Environment and Natural Resources Administrative Orders No. 95-23, No. 96-40, and No. 97-11. Meanwhile, the Philippine Atmospheric, Geophysical and

³ Comprised of the Civil Service Commission, National Disaster Risk Reduction and Management Development Council, Department of Interior and Local Government, and the Department of Budget and Management.

Astronomical Services Administration's powers, duties and functions are shaped by Presidential Decrees No. 78 and No. 1149, and Executive Order 128. Likewise, the climate change champion, the governor of Albay, was working within the arrangements set by the Local Government Code—the rule that provided the governor position the authority and power to execute, formulate, and conduct local CCA policies, programs, and activities.

In essence, mainstreaming operationalisation involves a network of institutional settings, and the institutions governing the challenges in mainstreaming CCA interact within this network. These interactions generate a chain of institutional interplays, wherein one action can create a series of reactions that can make a challenge a serious barrier or an important opportunity for mainstreaming. A very apt example of this idea is the series of institutional interactions and interplays that were generated between and among rule-based institutions-Disaster Risk Reduction and Management Act, Local Government Code, Department of Budget and Management Memorandum (dated March 15, 2012)----and the institutional organisations-Department of Budget and Management, Civil Service Commission—as a result of one provision in the Disaster Risk Reduction and Management Act, specifically, the creation the local disaster risk reduction officer (refer Section 8.5 and Chapters 6 and 7). According to Kim (2013, p. 980) "institutions do not exist in isolation but as embedded in a maze-like structure". In mainstreaming operationalisation, this structure exists as a network of institutional settings; therefore, overcoming the challenges in mainstreaming CCA necessitates broad institutional reforms that should reach across the various institutional settings within this network.

8.6 Final conclusions

This research posed the questions "How can mainstreaming of climate change adaptation into local land-use planning be understood?" and "How can the challenges in the operationalisation of mainstreaming be overcome?" To answer these, the research examined the challenges in integrating or "mainstreaming" CCA into local land-use planning in Albay, Philippines, and determined how to identify, characterise, categorise, and assess these challenges in order to aid planners and decision-makers in overcoming them effectively. Hence, the thesis:

- (1) explored the process of mainstreaming CCA, from its theoretical foundations to its operationalisation, with special interest in local land-use planning;
- (2) formulated a methodology that examined effectively the challenges in mainstreaming CCA, and generated metrics that would address these challenges;

- (3) analysed the state-of-play and linkages between the different challenges in mainstreaming CCA into land-use planning in Albay, Philippines, and determined how these challenges can be overcome; and
- (4) generated a more refined understanding of the operationalisation of mainstreaming CCA at the local scale.

This research contributes to the expanding literature in mainstreaming CCA by addressing an important issue in the research field—examining the challenges in operationalising the approach at the local scale, and assessing how to overcome these challenges. In essence, the research provides empirical evidence of the practical application of the mainstreaming endeavour, thereby transcending analysis from theory to operationalisation through a case study in Albay, Philippines. Specifically, this research contributes to the knowledge in CCA mainstreaming through these three accomplishments:

- (1) It presents a systematic methodology that planners can use to examine the complexity of the mainstreaming process. This methodology:
 - (1.1) is comprised of an analytical framework that can investigate the institutional dimension (i.e., institutional changes and transformations) of CCA; and
 - (1.2) generates quantitative indicators to monitor and assess the state-of-play of the mainstreaming process, and thereby help transform the abstract nature of the climate change concept into concrete and measurable terms.

This methodology output is significant in planning. Although planners now acknowledge that there is a need to act on climate change, most are unsure about how to proceed with the initiative (Hamin et al. 2014). As mainstreaming operationalisation is not straightforward (Ayers et al. 2014), planners need a systematic methodology for examining the institutional settings into which CCA will be integrated. In general, the operational recommendations on mainstreaming CCA focus on climate-related issues and the suggested tools and techniques to address these issues include vulnerability assessments, climate risk screening, and climate change scenario building, among others (Olhoff and Schaer 2010; Lebel et al. 2012; SPREP & UNDP 2013). Thus, the analytical framework of this research, which can investigate the institutional dimension of CCA, provides a perspective that the current mainstreaming processes lack (Gigli & Agrawala 2007; Olhoff & Schaer 2010). Meanwhile, the quantitative indicators developed by this research can make the abstract concept of CCA more concrete (Persson & Klein 2008). The metrics,

therefore, can translate the state-of-play of the adaptation effort into a language that planners can understand, and thereby enable them to have a solid basis for action (Oates 2011).

- (2) This study advances CCA researchers' understanding of the nature, linkages, and interdependencies among the barriers to adaptation through the:
 - (2.1) quantitative mainstreaming indicators that measure the severity of the impacts of the challenges in mainstreaming CCA, and explain the nature of the barriers and opportunities for mainstreaming CCA;
 - (2.2) correlation analyses that quantitatively determine the degree by which mainstreaming challenges are linked to each other;
 - (2.3) assessments of the state-of-play of the local mainstreaming process and mainstreaming challenges;
 - (2.4) analyses that determine the institutional roots of the barriers in mainstreaming; and
 - (2.5) analyses of how the mainstreaming barriers potentially can be transformed into opportunities for mainstreaming.

This set of outputs is important in planning and decision-making because understanding the challenges in adaptation is key to the successful on-ground application of the mainstreaming approach (Amundsen et al. 2010; Biesbroek et al. 2013; Clar et al. 2013; Eisenack et al. 2014). These challenges that impede and obstruct the mainstreaming operationalization resulted to the overall progress of the endeavour to be slow and ineffective (Uittenbroek et al. 2013; Revi et al. 2014; Lehmann et al. 2015). Although the literature on the barriers or challenges to adaptation is expanding, with the works of Amundsen et al. (2010), Biesbroek et al. (2011, 2013, 2014), Dang et al. (2014), Matasci et al. (2014) and Lehmann et al. (2015), among many others, there are still unresolved issues concerning the nature of the barriers, linkages and interdependencies among the barriers, and how to overcome the barriers (Burch 2010; Eisenack et al. 2014; Hamin et al. 2014; Klein et al. 2014; Kuruppu & Willie 2015). The information generated by this research clarifies these issues, and thus provides a knowledge base that can assist CCA analysts, practitioners, and planners to formulate strategies that can overcome these barriers or challenges. Moreover, it demonstrates how quantitative (i.e., correlation analysis,

mainstreaming indicators) and qualitative (i.e., indicator score assessments) techniques, methods, and analyses can be used to accomplish this task of strategising.

- (3) This research advances and generates a more refined understanding of the institutional dimension of the mainstreaming process by:
 - (3.1) illustrating the need for institutional mechanisms that support mainstreaming actions for effective operationalisation;
 - (3.2) demonstrating how the concept of institutional nestedness is linked with the challenges in mainstreaming CCA;
 - (3.3) explaining how active participation of local governments can help transition the barriers into opportunities for mainstreaming CCA; and
 - (3.4) conceptualising and exploring the network of interacting institutions and institutional arrangements involved in operationalising mainstreaming of CCA.

The institutional dimension of CCA remains the least understood aspect of the climate change issue (Evans & Stevens 2009; Pradhan et al. 2012; Rodima-Taylor 2012; Rodima-Taylor et al. 2012). The outputs from this research clarify certain aspects of this dimension. First, the research established that operationalisation of mainstreaming is a matter of institutional change and transformation; and the approach creates changes that will affect realities that are already functioning within existing institutional systems and institutional arrangements. Thus, this research strengthens the significance of developing the institutional capacities of systems for successful long-term adaptation to climate change. It also highlights the importance of institutional mechanisms to support the institutional changes resulting from the mainstreaming endeavour. Second, the research demonstrates that the practical application of mainstreaming at the local scale is dependent on other scales; consequently, the actions of local governments on climate change concerns often are nested in institutional frameworks and processes at higher levels (i.e., institutional nestedness) (Corfee-Morlot et al. 2009; Measham et al. 2011). Third, this research provides empirical evidence that supports the notion that local governments are crucial in addressing the challenges in CCA (Schreurs 2008; Burch 2010; Sharma & Tomar 2010; Pasquini & Shearing 2014). Fundamentally, national and local scale actions and decisions are equally critical in mainstreaming operationalisation (Ostrom 2010; Mukheibir et al. 2013; Jordan & Huitema 2014). Finally, these inter-scalar interactions among institutions prove that mainstreaming operationalisation involves a network of interacting institutions and institutional arrangements; therefore, addressing these challenges needs extensive institutional transformations that reach across the various institutional settings within this network. According to Cuevas et al. (2014, p. 22), the "institutional dimension of climate change adaptation involves an intricate web of relationships between and among institutions". Results of this research support this idea and it proposes further that the network of interacting institutions in the case of mainstreaming may exist in what Cuevas et al. (2014, p. 2) refer to as the institutional environment—the "array of institutions that influence and affect climate change adaptation behaviours and decisions" (see Appendix A3 for more details).

The significance of the research extends beyond academic and scholarly fields. Its outputs help to improve understanding of the mainstreaming process at the local scale and the research provides practical contributions to planning and policy-making (Huq & Ayer 2008; Persson & Klein 2008; Tang et al. 2009; Measham et al. 2011; Ayers et al. 2014). First, knowing the magnitude of the impact of a mainstreaming challenge can assist planners and decision-makers to decide which challenges need to be prioritised. This is noteworthy because knowing the primary challenges that need intervention can help local governments use their limited resources efficiently and take advantage of the opportunities they have. Second, knowing the relationships among the mainstreaming challenges can help planners and decision-makers formulate strategies that can have maximum impacts on mainstreaming CCA. For example, in the Philippines, the national government can campaign and encourage local chief executives, both at the provincial and city/municipal scales, to champion CCA in their respective jurisdictions. Given the tripartite relationship among local leadership, local government prioritisation, and commitment to CCA, and knowing that this relationship is characterised by the local leadership's positive influence on the other two challenges, the national government can explore the possibility that a climate change champion can improve the status of the challenges relating to local government prioritisation and commitment to CCA. Third, intensive analysis of institutions can help policy-makers design the appropriate institutional reforms needed in mainstreaming CCA, including the institutional mechanisms that can support the mainstreaming endeavour. This aspect is important given that institutional changes and concerns are among the important factors that determine the success or failure of an adaptation measure, especially at the local scale. Finally, analysing the stateof-play and the conditions surrounding the challenges and understanding how the challenges

can transcend from becoming barriers to opportunities for mainstreaming CCA can help planners and decision-makers determine how to overcome them (challenges). A thorough understanding of these concerns will help scholars, practitioners, planners, and decisionmakers anticipate the types of challenges that can be encountered during the mainstreaming process and determine the severity of impacts of these challenges. Hence, they can develop strategies that will overcome the challenges.

8.7 Future research possibilities

This thesis is founded on the notion that mainstreaming operationalisation requires an institutional perspective and it proved that understanding the institutional dimension of mainstreaming CCA is critical in the mainstreaming process. The research generated empirical data which suggested that each mainstreaming challenge exists and persists in another institutional setting with its own working or functioning institutions and institutional arrangements, and this (setting) is independent of either the land use planning or CCA settings. Hence, the research concluded that mainstreaming involves a network of interacting institutions and institutional arrangements that transcend across governance scales, and that the challenges in mainstreaming CCA encompass a chain of interactions or interplays within this network of institutions. These findings link mainstreaming of CCA with network analysis and related concepts such as network theory and complex systems.

Network approaches and analyses had been applied to a variety of concerns such as politics and governance (Kahler 2009; Kim 2013; Orsini et al. 2013); policies (Jordan & Schubert 1992; Weible 2005); environment and natural resources (McAllister et al. 2015); social-ecological systems (Lubell et al. 2014); information systems (Seror 1998), and many more. Some studies have started to apply network-based approaches in global environment and climate governance (Kahler 2009; Pattberg 2010; Green 2013; Pattberg et al. 2014), and the recent works of Pahl-Wostl (2009), Juhola and Westerhoff (2011), Funfgeld (2015), and Ingold and Balsiger (2015) showed that network analysis is being applied particularly in CCA research. This research corroborates this line of inquiry by providing empirical evidence that institutional networks exist and are critical to the analysis of mainstreaming operationalisation. Therefore, it encourages academics, scholars, and practitioners to explore network analysis, models and approaches in conducting future research on mainstreaming CCA.

Also, the research offered a methodology—to address a wicked problem such as CCA—that can be replicated under different conditions and circumstances. Thus, to expand and improve the knowledge on the challenges in mainstreaming CCA into local land-use

planning, the methodology can be used to conduct case studies in other local areas. Similarly, future studies can apply the methodology in other sectors—mainstreaming of CCA into infrastructure, agriculture, water, or education. Finally, other researches can modify the methodology in terms of the quality or composition of the survey respondents. Specifically, these studies can apply the extended peer community facet of the post-normal science domain to its full extent. That is, aside from key actors in the institutional setting being investigated (e.g., local land-use planning system), and the representatives of the national government, non-government agencies, and academic organisations who had experience in implementing projects CCA mainstreaming, the private stakeholders (i.e., businesses, households), and community representatives affected by the mainstreaming endeavour can be surveyed and interviewed. The results of other studies can validate the hypotheses advocated by this research, and likewise, provide additional insights that can develop further understanding of the institutional dimension of mainstreaming, in particular, and CCA, in general.

REFERENCES

- Adger, W. N. (2000). Institutional Adaptation to Environmental Risk Under the Transition in Vietnam. *Annals of the Association of American Geographers*, 90(4), 738-758.
- Adger, W. N., Agrawala, S., Mirza, M. M. Q., Conde, C., O'Brien, K., Pulhin, J., Takahashi, K. (2007). Assessment of Adaptation Practices, Options, Constraints and Capacity. In M. L. Parry, O. F. Canziani, J. P. Palutikof, P. J. v. d. Linden & C. E. Hanson (Eds.), *Climate Change 2007: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change* (pp. 717-743). Cambridge, UK: Cambridge University Press.
- Adger, W. N., Arnell, N. W., & Tompkins, E. L. (2005). Successful Adaptation to Climate Change Across Scales. *Global Environmental Change*, 15(2), 77-86.
- Agrawal, A. (2008). The Role of Local Institutions in Adaptation to Climate Change Social Dimensions of Climate Change. Washington DC, USA: The World Bank.
- Agrawala, S. (2006). Putting Climate Change in the Development Mainstream. In S. Agrawala (Ed.), Bridge Over Troubled Waters: Linking Climate Change and Development (pp. 23-43). Paris, France: OECD Publishing.
- Agrawala, S., & van Aalst, M. (2006). Bridging the Gap between Climate Change and Development. In S. Agrawala (Ed.), *Bridge Over Troubled Waters: Linking Climate Change and Development* (pp. 133-146). Paris, France: OECD Publishing.
- Albright, E. A. (2011). Policy Change and Learning in Response to Extreme Flood Events in Hungary: An Advocacy Coalition Approach. *Policy Studies Journal*, 39(3), 485-511.
- Amaral, L. A. N., & Ottino, J. M. (2004). Complex networks: Augmenting the Framework for the Study of Complex Systems. *The European Physical Journal B*, 38, 147–162.
- Amundsen, H., Berglund, F., & Westskog, H. (2010). Overcoming Barriers to Climate Change Adaptation: A Question of Multilevel Governance? *Environment and Planning C: Government and Policy*, 28, 276-289.
- Asian Development Bank [ADB]. (2005). Climate Proofing, A Risk-based Approach to Adaptation: Summary for Policy and Decision Makers *Pacific Studies Series*. Manila, Philippines: Asian Development Bank.
- Asuero, A. G., Sayago, A., & González, A. G. (2006). The Correlation Coefficient: An Overview. *Critical Reviews in Analytical Chemistry*, 36(1), 41-59.
- Australian Agency for International Development [AusAID]. (2010). Integration in Practice Integrating Disaster Risk Reduction, Climate Change and Environmental

Considerations in Ausaid Programs. Canberra, Australia: Australian Agency for International Development.

- Ayers, J., & Dodman, D. (2010). Climate Change Adaptation and Development I: The State of the Debate. *Progress in Development Studies*, 10(2), 161–168.
- Ayers, J., Huq, S., Wright, H., Faisal, A. M., & Hussain, S. T. (2014). Mainstreaming Climate Change Adaptation into Development in Bangladesh. *Climate and Development*, 6(4), 293-305.
- Basurto, X., Kingsley, G., McQueen, K., Smith, M., & Weible, C. M. (2010). A Systematic Approach to Institutional Analysis Applying Crawford and Ostrom's Grammar. *Political Research Quarterly*, 63(3), 523-537.
- Baxter, P., & Jack, S. (2008). Qualitative case study methodology: Study Design and Implementation for Novice Researchers. *The Qualitative Report*, 13(4), 544-559.
- Bellamy, J. J., & Hill, K. (2010). Monitoring Guidelines of Capacity Development in Global Environment Facility Projects. New York, USA: Global Support Programme, Bureau for Development Policy, United Nations Development Programme.
- Benson, C. (2009). Mainstreaming Disaster Risk Reduction into Development: Challenges and Experience in the Philippines. Geneva, Switzerland: International Federation of Red Cross and Red Crescent Societies / the ProVention Consortium.
- Berrang-Ford, L., Ford, J. D., & Paterson, J. (2011). Are We Adapting to Climate Change? *Global Environmental Change*, 21, 25–33.
- Bettini, Y., Farrelly, M., de Haan, F. J., & Brown, R. R. (2012). Achieving Water Sensitive Cities: Institutions as Building Blocks of Collective Action. Paper presented at the 7th International Conference. WSUD 2012: 7th International Conference on Water Sensitive Urban Design, 21-23 February 2012, Melbourne, Victoria.
- Biagini, B., & Dobardzic, S. (2011). Accessing Resources Under the Least Developed Countries Fund. Washington, USA: Global Environment Facility.
- Biesbroek, G. R., Klostermann, J., Termeer, C., & Kabat, P. (2011). Barriers to Climate Change Adaptation in the Netherlands. *Climate Law*, 2, 181–199.
- Biesbroek, G. R., Klostermann, J. E. M., Termeer, C. J. A. M., & Kabat, P. (2013). On the Nature of Barriers to Climate Change Adaptation. *Regional Environmental Change*, 13, 1119–1129.
- Biesbroek, G. R., Termeer, C. J. A. M., Klostermann, J. E. M., & Kabat, P. (2014). Rethinking Barriers to Adaptation: Mechanism-Based Explanation of Impasses in the Governance of an Innovative Adaptation Measure. *Global Environmental Change*, 26, 108–118.

- Bjurström, A., & Polk, M. (2011). Physical and Economic Bias in Climate Change Research:A Scientometric Study of IPCC Third Assessment Report. *Climatic Change*, 108, 1–22.
- Brannen, J. (2005). Mixing Methods: The Entry of Qualitative and Quantitative Approaches into the Research Process. *International Journal of Social Research Methodology*, 8(3), 173-184.
- Bravo, G., & Potvin, L. (1991). Estimating the Reliability of Continuous Measures with Cronbach's Alpha or the Intraclass Correlation Coefficient: Toward the integration of Two Traditions. *Journal of Clinical Epidemiology*, 44(4/5), 381-390.
- Bridge, P. D., & Sawilowsky, S. S. (1999). Increasing Physicians' Awareness of the Impact of Statistics on Research Outcomes: Comparative Power of the t-test and Wilcoxon Rank-Sum Test in Small Samples Applied Research. *Journal of Clinical Epidemiology*, 52(3), 229–235.
- Brondizio, E. S., Ostrom, E., & Young, O. R. (2009). Connectivity and the Governance of Multilevel Social-Ecological Systems: The Role of Social Capital. Annual Review of Environment and Resources, 34, 253–278.
- Burby, R. J., Deyle, R. E., Godschalk, D. R., & Olshansky, R. B. (2000). Creating Hazard Resilient Communities through Lans-use Planning. *Natural Hazards Review*, 1, 99-106.
- Burch, S. (2010). Transforming Barriers into Enablers of Action on Climate Change: Insights from Three Municipal Case Studies in British Columbia, Canada. *Global Environmental Change*, 20, 287–297.
- Carlson, D. (2012). Preparing for Climate Change: An Implementation Guide for Local Governments in British Columbia. British Columbia, Canada: West Coast Environmental Law.
- Cash, D. W., Clark, W. C., Alcock, F., Dickson, N. M., Eckley, N., Guston, D. H., & Mitchell, R. B. (2003). Knowledge Systems for Sustainable Development. Proceedings of the National Academy of Sciences of the United States of America, 100(14), 8086– 8091.
- Chatterjee, N. (2013). Tsunami-Blocking Mangroves Lure Carbon Investors: Southeast Asia. News Retrieved 23 April 2015, from http://www.bloomberg.com/news/articles/2013-11-19/tsunami-blocking-mangroves-lure-carbon-investors-southeast-asia.
- Cheshire, P., & Sheppard, S. (2002). The Welfare Economics of Land Use Planning. *Journal of Urban Economics*, 52, 242–269.
- Chevallier, R. (2012). Political Barriers to Climate Change Adaptation Implementation in SADC. In L. Masters & L. Duff (Eds.), *Overcoming Barriers to Climate Change*

Adaptation Implementation in Southern Africa (pp. 1-20). South Africa: Africa Institute of South Africa.

- Chinvanno, S., & Kerdsuk, V. (2013). Mainstreaming Climate Change into Community Development Strategies and Plans: A Case Study in Thailand Adaptation Knowledge Platform, Partner Report Series No. 5. Bangkok, Thailand: Stockholm Environment Institute.
- Chuku, C. A. (2010). Pursuing an Integrated Development and Climate Policy Framework in Africa: Options for Mainstreaming. *Mitigation and Adaptation Strategies for Global Change*, 15, 41–52.
- Clar, C., Prutsch, A., & Steurer, R. (2013). Barriers and Guidelines for Public Policies on Climate Change Adaptation: A Missed Opportunity of Scientific Knowledge-Brokerage. *Natural Resources Forum*, 7, 1–18.
- Claudio, C. P. (Ed.). (2012). Climate Change Adaptation Best Practices in the Philippines. Manila, Philippines: Department of Environment and Natural Resources.
- Climate Change Commission [CCC]. (2011). National Climate Change Action Plan: 2011-2028. Manila, Philippines: Climate Change Commission.
- Collins, N., Smith, G., & Allen, J. B. (2005). A Guide for Incorporating Adaptation to Climate Change into Land-Use Planning: CEF Consultants Ltd and CBCL Limited.
- Collins Dictionaries. (Ed.) (2012) Collins English Dictionary Complete & Unabridged 2012 Digital Edition [Online]. William Collins Sons & Co. Ltd.
- Conway, D., & Schipper, E. L. F. (2011). Adaptation to Climate Change in Africa: Challenges and Opportunities Identified from Ethiopia. *Global Environmental Change*, 21, 227–237.
- Corfee-Morlot, J., Kamal-Chaoui, L., Donovan, M. G., Cochran, I., Robert, A., & Teasdale,
 P.-J. (2009). Cities, Climate Change and Multilevel Governance *OECD Environmental Working Papers No. 14.* Paris, France: OECD publishing.
- Corpuz, A. (2012). Land Use Policy Impacts on Human Development in the Philippines HDN Discussion Paper Series, PHDR Issue 2012/2013 No. 1. Quezon City, Philippines: Human Development Network.
- Cuevas, S. C. (2011). Climate Change, Vulnerability, and Risk Linkages. *International Journal of Climate Change Strategies and Management*, 3(1), 29-60.

______. (2012). Examining Climate Change Adaptation Measures: An Early Warning System in the Philippines. *International Journal of Climate Change Strategies and Management*, 4(4), 358-385.

- Cuevas, S. C., Peterson, A., & Morrison, T. (2014). An Analytical Framework for Investigating Complex Institutions in Climate Change Adaptation: The Institutional Environment Matrix. In W. Leal Filho (Ed.), *Handbook of Climate Change Adaptation* (pp. 1-22). Berlin, Germany: Springer Berlin Heidelberg.
- Cuevas, S. C., Peterson, A., Robinson, C. & Morrison, T. H. (2015). Challenges in Mainstreaming Climate Change Adaptation into Local Land-use Planning: Evidence from Albay, Philippines. *The International Journal of Climate Change: Impacts and Responses*, 7(3):45–65.
- Dai, F. C., Lee, C. F., & Zhang, X. H. (2001). GIS-based Geo-environmental Evaluation for Urban Land-use Planning: A Case Study. *Engineering Geology*, 61, 257-271.
- Dang, H. L., Li, E., Bruwer, J., & Nuberg, I. (2014). Farmers' Perceptions of Climate Variability and Barriers to Adaptation: Lessons Learned from an Exploratory Study in Vietnam. *Mitigation and Adaptation Strategies for Global Change*, 19, 531–548.
- David, C. P. C., Racoma, B. A. B., Gonzales, J., & Clutario, M. V. (2013). A Manifestation of Climate Change? A Look at Typhoon Yolanda in Relation to the Historical Tropical Cyclone Archive. *Science Diliman*, 25(2), 78-85.
- Dazé, A., Ambrose, K., & Ehrhart, C. (2009). Climate Vulnerability and Capacity Analysis Handbook: CARE International.
- de Winter, J. C. F. (2013). Using the Student's t-test with Extremely Small Sample Sizes. *Practical Assessment, Research & Evaluation*, 18(10), 1-12.
- Department of Environment and Natural Resources [DENR]. (2014). Geohazard Mapping and Assessment Program Retrieved 24 June 2015, from http://denr.gov.ph/priorityprograms/geo-hazard-mapping-and-assessment-program.html.
- Di Gregorio, M., Hagedorn, K., Kirk, M., Korf, B., Mccarthy, N., Meinzen-Dick, R, Swallow
 B., Mwangi E., & Markelova H. (2012). Property Rights and Collective Action for
 Poverty Reduction: A Framework for Analysis. In E. Mwangi, H. Markelova & R.
 Meinzen-Dick (Eds.), *Collective Action and Property rights for Poverty Reduction:*Insights from Africa and Asia. Philadelphia: University of Pennsylvania Press.
- Dick, L., & Meinzen-Dick, R. (2011). The Congregational Commons. Paper presented at the Paper presented at the Biennial Conference of International Association for Study of the Commons, 10-14 January 2011, Hyderabad, India.
- Dovers, S. R., & Hezri, A. A. (2010). Institutions and Policy Processes: The Means to the Ends of Adaptation. *WIREs Climate Change*, 1, 212-231.

- Duff, L. (2011). Introduction. In L. Masters & L. Duff (Eds.), Overcoming Barriers to Climate Change Adaptation Implementation in Southern Africa (pp. xxiii-xxx). Pretoria 0001, South Africa: Africa Institute of South Africa.
- Eisenack, K., Moser, S. C., Hoffmann, E., Klein, R. J. T., Oberlack, C., Pechan, A., Rotter, M. & Termeer, C. J. A. M. (2014). Explaining and Overcoming Barriers to Climate Change Adaptation. *Nature Climate Change*, 4, 867-872.
- Eisenack, K., & Stecker, R. (2012). A Framework for Analyzing Climate Change Adaptations as Actions. *Mitigation and Adaptation Strategies for Global Change*, 17(3), 243–260.
- Ekstrom, J. A., & Moser, S. C. (2014). Identifying and Overcoming Barriers in Urban Climate Adaptation: Case Study Findings from the San Francisco Bay Area, California, USA. Urban Climate, 9, 54–74.
- Ekstrom, J. A., Moser, S. C., & Torn, M. (2011). Barriers to Climate Adaptation: A Diagnostic Framework. USA: California Energy Commission.
- Enemark, S. (2012). *Sustainable Land Governance: Three Key Demands*. Paper presented at the FIG Working Week 2012: Knowing to Manage the Territory, Protect the Environment, Evaluate the Cultural Heritage, 6-10 May 2012, Rome, Italy.
- Engle, N. L. (2011). Adaptive Capacity and Its Assessment. *Global Environmental Change*, 21, 647–656.
- Espinas, A. (2012). Geography and Public Planning: Albay and Disaster Risk Management HDN Discussion Paper Series, PHDR Issue 2012/2013 NO. 4. Quezon City, Philippines: Human Development Network.
- Evans, A., & Steven, D. (2009). An Institutional Architecture for Climate Change: A Concept Paper. New York, USA: Center on International Cooperation.
- Fano, J. A., Alpasan, M. T., Mitsunaga, T., & Tokunaga, Y. (2007). The Mayon 2006 Debris
 Flow: The Destructive Path of Typhoon Reming *FCSEC Technical Report Volume 3*.
 Metro Manila, Philippines: Department of Public Works and Highways.
- Farber, D. A. (2009). Climate Adaptation and Federalism: Mapping the Issues. San Diego Journal of Climate and Energy Law, 1, 259-286.
- Farrell, L. A. (2010). Mainstreaming Climate Change Adaptation into Urban Development: Lessons from Two South African Cities. (Masters thesis), Massachusetts, USA: Massachusetts Institute of Technology.
- Federal Democratic Republic of Ethiopia. (2011). Ethiopia's Climate-Resilient Green Economy: Green Economy Strategy. Ethiopia: Federal Democratic Republic of Ethiopia.

- Field, C.B., Barros, V. R., Mach, K. J., Mastrandrea, M. D., van Aalst, M., Adger, W. N., Arent, D. J., Barnett, J., Betts, R., Bilir, T. E., Birkmann, J., Carmin, J., Chadee, D. D., Challinor, A. J., Chatterjee, M., Cramer, W., Davidson, D. J., Estrada, Y. O. Gattuso, J.-P., Hijioka, Y., Hoegh-Guldberg, O., Huang, H. Q., Insarov, G. E., Jones, R. N., Kovats, R. S., Romero-Lankao, P., Larsen, J. N., Losada, I. J., Marengo, J. A., McLean, R. F., Mearns, L. O., Mechler, R., Morton, J. F., Niang, I., Oki, T., Olwoch, J. M., Opondo, M., Poloczanska, E. S., Pörtner, H.-O., Redsteer, M. H., Reisinger, A., Revi, A., Schmidt, D. N., Shaw, M. R., Solecki, W., Stone, D. A., Stone, J. M. R., Strzepek, K. M., Suarez, A. G., Tschakert, P., Valentini, R., Vicuña, S., Villamizar, A., Vincent, K. E., Warren, R., White, L. L., Wilbanks, T. J., Wong, P. P. & Yohe, g. W. (2014). Technical Summary. In C. B. Field, V. R. Barros, D. J. Dokken, K. J. Mach, M. D. Mastrandrea, T. E. Bilir, M. Chatterjee, K. L. Ebi, Y. O. Estrada, R. C. Genova, B. Girma, E. S. Kissel, A. N. Levy, S. MacCracken, P. R. Mastrandrea & L. L. White (Eds.), Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 35-94). Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Fitts, D. A. (2010). Improved Stopping Rules for the Design of Efficient Small-Sample Experiments in Biomedical and Biobehavioral Research. *Behavior Research Methods*, 42(1), 3-22.
- FitzGibbon, J., & Mensah, K. O. (2012). Climate Change as a Wicked Problem: An Evaluation of the Institutional Context for Rural Water Management in Ghana. SAGE Open, 1-12.
- Florano, E. R. (2014). Mainstreaming Integrated Climate Change Adaptation and Disaster Risk Reduction in Local Development Plans in the Philippines. In W. Leal Filho (Ed.), *Handbook of Climate Change Adaptation* (pp. 433-456). Berlin, Germany: Springer Berlin Heidelberg.
- Flyvbjerg, B. (2006). Five Misunderstandings about Case-Study Research. *Qualitative Inquiry*, 12(2), 219-245.
- Flyvbjerg, B. (2011). Case Study. In N. K. Denzin & Y. S. Lincoln (Eds.), *The Sage Handbook of Qualitative Research* (4th ed., pp. 301-316). Thousand Oaks, CA: SAGE.
- Food and Agriculture Organization [FAO]. (1993). Guidelines for Land-use Planning FAO Development Series 1. Rome: Food and Agriculture Organization of the United Nations.
- Forsyth, A. (2012). Defining Suburbs. Journal of Planning Literature, 27(3), 270-281.

- Frankel-Reed, J., Fröde-Thierfelder, B., & Porsché, I. (2011). Integrating Climate Change Adaptation into Development Planning: A Practice-Oriented Training Based on an OECD Policy Guidance. Eschborn, Germany: Deutsche Gesellschaft für Internationale Zusammenarbeit (GIZ) GmbH.
- Freudenberger, M., & Miller, D. (2010). Climate Change, Property Rights, & Resource Governance: Emerging Implications for USG Policies and Programming USAID Issue Brief. Washington DC, USA: US Agency for International Development.
- Fritz, C. O., Morris, P. E., & Richler, J. J. (2012). Effect Size Estimates: Current Use, Calculations, and Interpretation. *Journal of Experimental Psychology: General*, 141(1), 2–18.
- Fröhlich, J., & Knieling, J. (2013). Conceptualising Climate Change Governance. In J. Knieling & W. L. Filho (Eds.), *Climate Change Governance* (pp. 9-26). New York, USA: Springer-Verlag Berlin Heidelberg.
- Frost, B. (2007). Measuring Performance: Seven Good Reasons to Use a Scorecard *Performance Perspectives Series*. Texas, USA: Measurement International.
- Fry, P.-J., & Williams, S. (2013). Reforming Planning Processes: Rockhampton 2050 Pilot. Local Government Climate Hazard Risk Management Toolkit Study (pp. 196). Gold Coast, Australia: National Climate Change Adaptation Research Facility.
- Funfgeld, H. (2015). Facilitating local climate change adaptation through transnational municipal networks. *Current Opinion in Environmental Sustainability*, 12, 67–73.
- Funtowicz, S.O. & Ravetz, J.R. (1993). Science for the Post-Normal Age. *Futures*, 25, 739–755.
- Gardner, J., Parsons, R., & Paxton, G. (2010). Adaptation Benchmarking Survey: Initial Report CSIRO Climate Adaptation Flagship Working paper No. 4. Queensland, Australia: Commonwealth Scientific and Industrial Research Organisation.
- Getchell, M. (2013). *Investigating the Barriers to the Adoption of Solar Water Heating in Oregon*. (Master of Public Policy Essay), Oregon, USA: Oregon State University.
- Gigli, S., & Agrawala, S. (2007). Stocktaking of Progress on Integrating Adaptation to Climate Change into Development Co-operation Activities OM/ENV/EPOC/DCD/DAC(2007)1/FINAL. Paris, France: Organisation for Economic Co-Operation and Development.
- Giupponi, C. (2014). Decision Support for Mainstreaming Climate Change Adaptation in Water Resources Management. *Water Resource Management*, 28, 4795–4808.

- Gliem, J. A., & Gliem, R. R. (2003). Calculating, Interpreting, and Reporting Cronbach's Alpha Reliability Coefficient for Likert-Type Scales. Ohio, USA: Midwest Research to Practice Conference in Adult, Continuing, and Community Education.
- Godschalk, D. R. (2004). Land Use Planning Challenges: Coping with Conflicts in Visions of Sustainable Development and Livable Communities. *Journal of the American Planning Association*, 70(1), 5-13.
- Gonzales, R. P. (1997). Case Studies of Decentralization: Philippines *Decentralization and Rural Development (FAO), no. 13.* Rome, Italy: Food and Agriculture Organization.
- Goosen, H., Groot-Reichwein, M. A. M. d., Masselink, L., Koekoek, A., Swart, R., Bessembinder, J., Witte, J. M. P., Stuyt, L., Blom-Zandstra, G. & Immerzeel, W. (2014). Climate Adaptation Services for the Netherlands: An Operational Approach to Support Spatial Adaptation Planning. *Regional Environmental Change*, 14, 1035–1048.
- Gotis, M. Q. (2008). Introduction to the Rationalized Planning System. Paper presented at the Presentation at the 5th National CBMS Conference, 28 January 01 February 2008, Manila, Philippines.
- Gottlieb, C. (2014). Regional Land Use Planning: A Collaborative Solution for the Conservation of Natural Resources. *Journal of Environmental Law and Litigation*, 29(1), 35-86.
- Govind, P. (2011). Managing the Relationship Between Adaptation and Coastal Land Use Development through the Use Of s 149 Certificates. *Macquarie Journal of International and Comparative Environmental Law*, 7(1), 94-107.
- Gupta, J., Termeer, C., Klostermann, J., Meijerink, S., van den Brink, M., Jong, P., Sibout, N.
 & Bergsma, E. (2010). The Adaptive Capacity Wheel: A Method to Assess the Inherent Characteristics of Institutions to Enable the Adaptive Capacity of Society. *Environmental Science & Policy*, 13(6), 459-471.
- Gupta, J., & van der Brink, M. (Eds.). (2010). Mainstreaming Climate Change in Development Cooperation: Theory, Practice and Implications for the European Union. 1st ed. Cambridge: Cambridge University Press.
- Haanpaa, S., & Peltonen, L. (2007). Institutional Vulnerability of Spatial Planning Systems Against Climate Change in the BSR. Espoo, Finland: Helsinki University of Technology.
- Hamdani, Y., Setyawan, D., Setiawan, B., & Affandi, A. K. (2014). Mainstreaming Adaptation Climate Change into Strategic Environmental Assessment Case Study Banyuasin District, South Sumatra Province. *Journal of Sustainable Development*, 7(6), 8-17.

- Hammill, A., & Tanner, T. (2011). Harmonising Climate Risk Management: Adaptation Screening and Assessment Tools for Development Co-operation OECD Environment Working Papers, No. 36. Paris: OECD Publishing.
- Hamin, E. M., Gurran, N., & Emlinger, A. M. (2014). Barriers to Municipal Climate Adaptation: Examples from Coastal Massachusetts' Smaller Cities and Towns. *Journal* of the American Planning Association, 80(2), 110-122.
- Harley, M., & van Minnen, J. (2009). Development of Adaptation Indicators ETC/ACC Technical Paper 2009/6. The Netherlands: European Topic Centre on Air and Climate Change.
- Head, B.W. (2014). Evidence, Uncertainty, and Wicked Problems in Climate Change Decision Making in Australia. *Environment and Planning C: Government and Policy*, 32, 663–679.
- Head, B.W. & Alford, J. (2015). Wicked Problems: Implications for Public Policy and Management, *Administration & Society*, 47(6), 711–739.
- Healy, S. (2011). Post-normal Science in Postnormal Times. Futures, 43, 202–208.
- Heikkila, T., Schlager, E., & Davis, M. W. (2011). The Role of Cross-Scale Institutional Linkages in Common Pool Resource Management: Assessing Interstate River Compacts. *The Policy Studies Journal*, 39(1), 25.
- Hedger, M. M., Mitchell, T., Leavy, J., Greeley, M., & Downie, A. (2008). Desk review: evaluation of adaptation to climate change from a development perspective. East Sussex, United Kingdom: Institute of Development Studies.
- Henry, A. D., & Dietz, T. (2011). Information, Networks, and the Complexity of Trust in Commons Governance. *International Journal of the Commons*, 5(2), 188–212.
- Hesse-Biber, S., & Johnson, R. B. (2013). Coming at Things Differently: Future Directions of Possible Engagement with Mixed Methods Research. *Journal of Mixed Methods Research*, 7(2), 103–109.
- Holtz, M. (2013). Saved by the Mangroves? A Philippine Town Dodges Haiyan's Storm Surge. Science, Tech & Environment Retrieved 23 April 2015, from http://www.pri.org/stories/2013-11-29/saved-mangroves-philippine-town-dodgeshaiyans-storm-surge.
- Horrocks, L., Pringle, P., Cornu, E. L., & Winne, S. (2012). Review of International Experience in Adaptation Indicators, Report for Adaptation Sub-Committee, AEA/R/ED57591 Issue Number 3. Oxfordshire, UK: AEA Technology.
- Housing and Land Use Regulatory Board [HLURB]. (2001). Planning Strategically: Guidelines for the Application of the Strategic Planning Process in the Preparation of

the Comprehensive Land Use Plan (CLUP) and to Important Urban Area Issues and Problems. Manila, Philippines: Housing and Land Use Regulatory Board.

- ______. (2006). CLUP Guidebook: A Guide for Comprehensive Land-Use Plan Preparation, Volume 1 Manila, Philippines: Housing and Land Use Regulatory Board.
- . (2014). Supplemental Guidelines on Mainstreaming Climate and Disaster Risks in the Comprehensive Land Use Plan. Manila, Philippines: Housing and Land-Use Regulatory Board.
- ______. (2015). About Us Retrieved 24 June 2015, from http://hlurb.gov.ph/about-us/?tabgarb=tab4.
- Huq, S., & Ayers, J. (2008). Taking Steps: Mainstreaming National Adaptation IIED Briefing. London, UK: International Institute for Environment and Development.
- Huq, S., Reid, H., Konate, M., Rahman, A., Sokona, Y., & Crick, F. (2004). Mainstreaming Adaptation to Climate Change in Least Developed Countries (LDCs). *Climate Policy*, 4(1), 25-43.
- Huxtable, J., & Yen, N. T. (2009). Mainstreaming Climate Change Adaptation: A Practioner's Handbook. Vietnam: CARE International.
- Ike, R. M. (2009). Advocacy Coalition Framework: An Approach to Critical Theory and Belief Systems in Policy Making. Paper presented at the Annual Convention of the British Political Studies Association, 7 – 9 April 2009, Manchester, U.K.
- Inderberg, T. H., & Eikeland, P. O. (2009). Limits to Adaptation: Analysing Institutional Constraints. In W. N. Adger, I. Lorenzoni & K. L. O'Brien (Eds.), *Adapting to Climate Change: Thresholds, Values, Governance* (pp. 433-447). UK: Cambridge University Press.
- Ingold, K., & Balsiger, J. (2015). Sustainability principles put into practice: case studies of network analysis in Swiss climate change adaptation. *Regional Environmental Change*, 15, 529–538.
- International Institute for Sustainable Development [IISD]. (2012). CRiSTAL User's Manual
 Version 5: Community-based Risk Screening Tool Adaptation and Livelihoods.
 Manitoba, Canada: International Institute for Sustainable Development.
- Intergovernmental Panel on Climate Change [IPCC]. (2001). *Climate Change 2001: Impacts, Adaptation and Vulnerability*. Contribution of Working Group II to the Third Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, UK: Cambridge University Press.

______. (2007). Climate Change 2007: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, UK: Cambridge University Press.

. (2014a). Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.

- . (2014b). Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Janu'sonis, S. (2009). Comparing Two Small Samples with an Unstable, Treatment-Independent Baseline. *Journal of Neuroscience Methods*, 179, 173–178.
- Jenkins-Smith, H. C., & Sabatier, P. A. (1994). Evaluating the Advocacy Coalition Framework. *Journal of Public Policy*, 14(2), 175-203.
- Jentoft, S. & Chuenpagdee, R. (2009). Fisheries and Coastal Governance as a Wicked Problem. *Marine Policy*, 33, 553–560.
- Jones, R., Young, C., Handmer, J., Keating, A., Mekala, G., & Sheehan, P. (2013). Valuing Adaptation under Rapid Change NCCARF Publication 56/13. Gold Coast: National Climate Change Adaptation Research Facility.
- Jordan, A., & Huitema, D. (2014). Policy Innovation in a Changing Climate: Sources, Patterns and Effects. *Global Environmental Change*, 29, 387–394.
- Jordan, A., & O'Riordan, T. (1997). Social Institutions and Climate Change: Applying Cultural Theory to Practice CSERGE Working paper GEC 97- 15. Norwich, UK: Centre for Social and Economic Research on the Global Environment.
- Jordahl Jr., H. C. (1984). Land Use Planning in the 1980's. In F. P. Miller (Ed.), Land Use Planning Techniques and Policies (pp. 13-36). Madison, WI: Soil Science Society of America, and American Society of Agronomy.
- Juhola, S., & Westerhoff, L. (2011). Challenges of adaptation to climate change across multiple scales: a case study of network governance in two European countries. *Environmental Science & Policy*, 14, 239–247.
- Kaiser, E. J., & Godschalk, D. R. (1995). Twentieth Century Land Use Planning: A Stalwart Family Tree. *Journal of the American Planning Association*, 61(3), 365-385.

- Kaswamila, A. L., & Songorwa, A. N. (2009). Participatory Land-use Planning and Conservation in Northern Tanzania Rangelands. *African Journal of Ecology*, 47, 128– 134.
- Kato, T., Ellis, J., Pauw, P., & Caruso, R. (2014). Scaling Up and Replicating Effective Climate Finance Interventions *Climate Change Expert Group Paper No. 2014(1)*. Paris, France: Organisation for Economic Co-operation and Development.
- Kim, R. E. (2013). The Emergent Network Structure of the Multilateral Environmental Agreement System. *Global Environmental Change*, 23, 980–991.
- Kissinger, G., & Namgyel, T. (2014). NAPAs and NAPs in Least Developed Countries. In A. Sharma (Ed.), *LDC Briefing*. UK: UK Department for International Development.
- Klein, P. G. (2000). New Institutional Economics. In B. Bouckeart & G. de Geest (Eds.), *Encyclopedia of Law and Economics* (pp. 456-489). Cheltenham, UK: Edward Elgar.
- Klein, R. J. T., Eriksen, S. E. H., Næss, L. O., Hammill, A., Tanner, T. M., Robledo, C., & O'Brien, K. L. (2007). Portfolio Screening to Support the Mainstreaming of Adaptation to Climate Change into Development Assistance. *Climatic Change*, 84, 23–44.
- Klein, R. J. T., Midgley, G. F., Preston, B. L., Alam, M., Berkhout, F. G. H., Dow, K., & Shaw, M. R. (2014). Adaptation Opportunities, Constraints, and Limits. In C. B. Field, V. R. Barros, D. J. Dokken, K. J. Mach, M. D. Mastrandrea, T. E. Bilir, M. Chatterjee, K. L. Ebi, Y. O. Estrada, R. C. Genova, B. Girma, E. S. Kissel, A. N. Levy, S. MacCracken, P. R. Mastrandrea & L. L. White (Eds.), *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects*. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 899-943.). Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Klein, R. J. T., Schipper, E. L. F., & Dessai, S. (2005). Integrating Mitigation and Adaptation into Climate and Development Policy: Three Research Questions. *Environmental Science & Policy*, 8, 579–588.
- Kok, M. T. J., & Coninck, H. C. d. (2007). Widening the scope of policies to address climate change: directions for mainstreaming. *Environmental Science & Policy*, 10(7-8), 587– 599.
- Koomen, E., Loonen, W., & Hilferink, M. (2008). Climate-change Adaptations in Land-Use Planning; A Scenario-Based Approach. In L. Bernard, A. Friis-Christensen & H. Pundt (Eds.), *The European Information Society Taking Geoinformation Science One Step Further: Lecture Notes in Geoinformation and Cartography Series*. Berlin, Germany: Springer-Verlag Berlin Heidelberg.

- Koontz, T. M. (2006). Collaboration for Sustainability? A Framework for Analyzing Government Impacts in Collaborative-Environmental Management. Sustainability: Science, Practice, & Policy, 2(1), 15-24.
- Kreft, S., & Eckstein, D. (2013). Global Climate Risk Index 2014: Who Suffers Most from Extreme Weather Events? Weather-Related Loss Events in 2012 and 1993 to 2012. Berlin, Germany: Germanwatch.
- Kreft, S., Eckstein, D., Junghans, L., Kerestan, C., & Hagen, U. (2014). Global Climate Risk Index 2015: Who Suffers Most From Extreme Weather Events? Weather-related Loss Events in 2013 and 1994 to 2013. Berlin, Germany: Germanwatch.
- Kuruppu, N., & Willie, R. (2015). Barriers to Reducing Climate Enhanced Disaster Risks in Least Developed Country-Small Islands Through Anticipatory Adaptation. Weather and Climate Extremes, 7, 72–83.
- La Trobe, S. & Davis, P. I. (2005). Mainstreaming Disaster Risk Reduction: A Tool for Development Organisations. Middlesex, UK: Tearfund.
- Lagmay, A. M. F. (2014). Devastating Storm Surges of Typhoon Yolanda. *Project NOAH Open-File Reports*, 3(6), 45-56.
- Lal, P. N., Mitchell, T., Aldunce, P., Auld, H., Mechler, R., Miyan, A., Romano, L. E., & Zakaria, S. (2012). National Systems for Managing the Risks from Climate Extremes and Disasters. In C. B. Field, V. Barros, T. F. Stocker, D. Qin, D. J. Dokken, K. L. Ebi, M. D. Mastrandrea, K. J. Mach, G.-K. Plattner, S. K. Allen, M. Tignor & P. M. Midgley (Eds.), *Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation: A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change (IPCC)* (pp. 339-392). Cambridge, UK, and New York, NY, USA: Cambridge University Press.
- Lasco, R. D., Delfino, R. J., Pulhin, F. B., & Rangasa, M. (2008). The Role of Local Government Units in Mainstreaming Climate Change Adaptation in the Philippines *AdaptNet Policy Forum 08-09-P-Ad*. Melbourne, Australia: RMIT University Global Cities Institute's Climate Change Adaptation Working Group.
- Lasco, R. D., Pulhin, F. B., Jaranilla-Sanchez, P. A., Delfino, R. J. P., Gerpacio, R., & Garcia, K. (2009). Mainstreaming Adaptation in Developing Countries: The Case of the Philippines. *Climate and Development*, 1, 130-146.
- Lasco, R. D., Delfino, R. J. & Sanchez, P. A. (n.d.). Local Government Initiatives to Climate Change Adaptation: A Case Study of Albay, Philippines Retrieved 24 June 2015, from http://www.worldagroforestry.org/downloads/publications/PDFS/PO08105.PDF.
- Lazarus, R. J. (2009). Super Wicked Problems and Climate Change: Restraining the Present to Liberate the Future. *Cornell Law Review*, 94(1153), 1189–1190.
- LDC Experts Group. (2012). Least Developed Countries National Adaptation Plans: Technical Guidelines for the National Adaptation Plan Process. Bonn, Germany: United Nations Climate Change Secretariat.
- Lebel, L., Anderies, J. M., Campbell, B., Folke, C., Hatfield-Dodds, S., Hughes, T. P., & Wilson, J. (2006). Governance and the Capacity to Manage Resilience in Regional Social-Ecological Systems. *Ecology and Society*, 11(1), 9.
- Lebel, L., Li, L., Krittasudthacheewa, C., Juntopas, M., Vijitpan, T., Uchiyama, T., & Krawanchid, D. (2012). Mainstreaming Climate Change Adaptation into Development Planning. Bangkok, Thailand: Adaptation Knowledge Platform and Stockholm Environment Institute.
- Lebel, L., Nikitina, E., Pahl-Wostl, C., & Knieper, C. (2013). Institutional Fit and River Basin Governance: a New Approach Using Multiple Composite Measures. *Ecology and Society* 18(1), 1-7.
- Lehmann, P., Brenck, M., Gebhardt, O., Schaller, S., & Süßbauer, E. (2015). Barriers and Opportunities for Urban Adaptation Planning: Analytical Framework and Evidence from Cities in Latin America and Germany. *Mitigation and Adaptation Strategies for Global Change*, 20, 75–97.
- Levin, K., Cashore, B., Bernstein, S., & Auld, G. (2012). Overcoming the Tragedy of Super Wicked Problems: Constraining Our Future Selves to Ameliorate Global Climate Change. *Policy Science*, 45, 123–152.
- Lhendup, P. (2012). Integration of Climate Adaptation into Development and Conservation Planning in Bhutan: Issue Identification and Recommendations Adaptation Knowledge Platform, Partner Report Series No. 1. Bangkok, Thailand: Stockholm Environment Institute.
- Lindley, S. J., Handley, J. F., Theuray, N., Peet, E., & Mcevo, D. (2006). Adaptation Strategies for Climate Change in the Urban Environment: Assessing Climate, Change Related Risk in UK Urban Areas. *Journal of Risk Research*, 9(5), 543-568.
- Linnér, B.-O. (2006). Authority through Synergism: The Roles of Climate Change Linkages. *European Environment*, 16(5), 278-289.
- Ludwig, D. (2001). The Era of Management is Over. *Ecosystems*, 4, 758–764.
- Lyle-Gonga, M. (2013). A Critical Analysis of Gender Mainstreaming. *Politics & Gender*, 9(2), 209-213.

- Maheady, L., Gradel, K., & Magiera, K. (2005). Mainstreaming. Encyclopedia of Behavior Modification and Cognitive Behavior Therapy: Volume 1: Adult Clinical Applications Volume 2: Child Clinical Applications Volume 3: Educational Applications. SAGE Publications, Inc. Thousand Oaks, CA: SAGE Publications, Inc.
- Mangoyana, R. B., Thomsen, D. C., Smith, T. F., Preston, B. L., Heinz, S., Maloney, M., Withycombe G., & Armstrong, I. (2012). Prioritising Coastal Adaptation Development Options for Local Government: Literature Review of Adaptation to Climate Change in the Coastal Zone. Canberra, Australia: Australian Department of Climate Change and Energy Efficiency.
- Manila Observatory. (2005). Climate- and Weather-Related Risk Maps Retrieved 11 June 2015, from http://vm.observatory.ph/cw_maps.html.
- Maplecroft. (2012). Climate Change Vulnerability Index 2013 Most at Risk CitiesRetrieved14August2015,fromhttp://www.preventionweb.net/files/29649_maplecroftccvisubnationalmap.pdf.
- ______. (2013). Climate Change Vulnerability Index 2014 Retrieved 14 August 2015, from http://reliefweb.int/sites/reliefweb.int/files/resources/Climate_Change_Vulnerability_In
 - dex_%202014_Map.pdf.

. (2014). Climate Change and Environmental Risk Atlas 2015 Retrieved 14 August 2015, from http://maplecroft.com/portfolio/new-analysis/2014/10/29/climatechange-and-lack-food-security-multiply-risks-conflict-and-civil-unrest-32-countriesmaplecroft/.

- Marshall, S., & Banister, D. (2008). Land Use and Transport European Research towards Integrated Policies Retrieved 29 April 2015, from http://app.knovel.com/hotlink/toc/id:kpLUTERIP4/land-use-transport-european/landuse-transport-european.
- Matasci, C., Kruse, S., Barawid, N., & Thalmann, P. (2014). Exploring Barriers to Climate Change Adaptation in the Swiss Tourism Sector. *Mitigation and Adaptation Strategies* for Global Change, 19, 1239–1254.
- McFadden, L., Priest, S., & Green, C. (2010). Introducing Institutional Mapping: A Guide for SPICOSA Scientists, SPICOSA Project Report. London, UK: Flood Hazard Research Centre, Middlesex University.
- McGinnis, M. (2011). An Introduction to IAD and the Language of the Ostrom Workshop: A Simple Guide to a Complex Framework. *The Policy Studies Journal*, 39(1), 169-183.

- Measham, T. G., Preston, B. L., Smith, T. F., Brooke, C., Gorddard, R., Withycombe, G., & Morrison, C. (2011). Adapting to Climate Change through Local Municipal Planning: Barriers and Challenges. *Mitigation and Adaptation Strategies for Global Change*, 16, 889–909.
- Mercado, E. S. (2011). Integration of CCA and DRMM into Local Land Use Plans: Issues and Challenges and Way Forward: Adaptation to Climate Change and Conservation of Biodiversity (ACCBio) in cooperation with the LGU Planning Synergy Product Environment and Rural Development Program (EnRD) / Decentralization Program (DP) and Inventory of Methods for CCA.
- Merriam-Webster [Web] (n.d.). Mainstream. *Merriam-Webster.com.*, from http://www.merriam-webster.com/dictionary/mainstream.
- Mertz, O., Halsnæs, K., Olesen, J. E., & Rasmussen, K. (2009). Adaptation to Climate Change in Developing Countries. *Environmental Management*, 43, 743–752.
- Miller, K., Harley, M., Kent, N., & Beckmann, K. (2012). Climate Change Adaptationrelated Indicators *ER23 Final Report*. Scotland, UK: Sniffer.
- Mimura, N., Pulwarty, R. S., Duc, D. M., Elshinnawy, I., Redsteer, M. H., Huang, H. Q., Nkem, J. N. & Rodriguez, R. A. S. (2014). Adaptation Planning and Implementation. In C. B. Field, V. R. Barros, D. J. Dokken, K. J. Mach, M. D. Mastrandrea, T. E. Bilir, M. Chatterjee, K. L. Ebi, Y. O. Estrada, R. C. Genova, B. Girma, E. S. Kissel, A. N. Levy, S. MacCracken, P. R. Mastrandrea & L. L. White (Eds.), *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects.* Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 869-898). Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Mines and Geosciences Bureau [MGB] Region 5. (2014). RD Pestaño renews call on LGU on geohazards Retrieved 26 June 2015, from http://www.mgb.gov.ph/MGBV/PR_2014_10JAN.htm.
- Ministry of Environment and Forests [MoEF]. (2009). Bangladesh Climate Change Strategy and Action Plan 2009 Dhaka, Bangladesh: Ministry of Environment and Forests, Government of the People's Republic of Bangladesh.
- Mitchell, T., Tanner, T., & Wilkinson, E. (2006). Overcoming the Barriers: Mainstreaming Climate Change Adaptation in Developing Countries *Tearfund Climate Change Briefing Paper 1*. UK: Tearfund.

Morrison, T. H. (2006). Pursuing Rural Sustainability at the Regional Level: Key Lessons from the Literature on Institutions, Integration, and the Environment. *Journal of Planning Literature*, 21(2), 143-152.

______. (2014). Developing a Regional Governance Index: The Institutional Potential of Rural Regions. *Journal of Rural Studies*, 35, 101-111.

- Moser, S. C., & Ekstrom, J. A. (2010). A Framework to Diagnose Barriers to Climate Change Adaptation. *PNAS*, 107(21), 22026–22031.
- Moser, S. C., Kasperson, R. E., Yohe, G., & Agyeman, J. (2008). Adaptation to Climate Change in the Northeast United States: Opportunities, Processes, Constraints. *Mitigation and Adaptation Strategies for Global Change*, 13, 643–659.
- Moutinho, L. (2011). Correlation Analysis. In L. Moutinho & G. Hutcheson (Eds.), *The SAGE Dictionary of Quantitative Management Research* (pp. 57-61). London, UK: SAGE Publications Ltd.
- Mukheibir, P., Kuruppu, N., Gero, A., & Herriman, J. (2013). Overcoming Cross-scale Challenges to Climate Change Adaptation for Local Government: a Focus on Australia. *Climatic Change*, 121, 271–283.
- Mukheibir, P., & Ziervogel, G. (2007). Developing a Municipal Adaptation Plan (MAP) for Climate Change: The City of Cape Town. *Environment & Urbanization*, 19(1), 143-158.
- Nambi, A. A., & Prabhakar, S. V. R. K. (2011). Mainstreaming Adaptation to Climate Change in Agriculture and Water Sectors in India: Current Status, Issues and Barriers. *Asian Journal of Environment and Disaster Management*, 2(4), 443–451.
- National Aeronautics and Space Administration [NASA]. (2013). Super Typhoon Haiyan Surges Across the Philippines Retrieved 04 June 2014, from http://earthobservatory.nasa.gov/NaturalHazards/view.php?id=82348.
- National Disaster Risk Reduction and Management Council [NDRRMC]. (2011). National Disaster Risk Reduction and Management Plan (NDRRMP) 2011-2028. Manila, Philippines: National Disaster Risk Reduction and Management Council.
 - . (2012). Memorandum for the SND and Chairperson, NDRRMC: Subject: SUBJECT: I. Final Report on the Effects and Emergency Management re Tropical Storm "SENDONG" (Washi); II. Status of Early Recovery Programs in Region X (Cagayan de Oro and Iligan Cities). Metro Manila, Philippines: National Disaster Risk Reduction and Management Council.

. (2014). NDRRMC update: Updates re the Effects of Typhoon "Yolanda" Haiyan. National Disaster Risk Reduction and Management Council Metro Manila, Philippines.

National Statistical Coordination Board [NSCB]. (2013). Table 1. Annual Per Capita Poverty Threshold, Poverty Incidence and Magnitude of Poor Families, by Region and Province: 1991, 2006, 2009 and 2012. Retrieved 04 August 2015, from: http://www.nscb.gov.ph/poverty/dataCharts.asp.

. (2014). Concepts and Definitions: Local Government Units Retrieved 24 June, 2015, from http://www.nscb.gov.ph/activestats/psgc/articles/con_lgu.asp.

- . (2015). Press Release: Total Number of Barangays Still Stands at 42,029 Retrieved 26 May 2015, from http://www.nscb.gov.ph/pressreleases/2015/PR-2015_18May_psgc.asp.
- National Statistics Office [NSO]. 2010. 2010 Census of Population and Housing [Online]. Retrieved 16 January 2015, from

http://census.gov.ph/sites/default/files/attachments/hsd/pressrelease/Bicol.pdf.

- Newman, M. E. J. (2011). Complex systems. American Journal of Physics, 79(800-810).
- Niang, I., Ruppel, O. C., Abdrabo, M. A., Essel, A., Lennard, C., Padgham, J., & Urquhart, P. (2014). Africa. In V. R. Barros, C. B. Field, D. J. Dokken, M. D. Mastrandrea, K. J. Mach, T. E. Bilir, M. Chatterjee, K. L. Ebi, Y. O. Estrada, R. C. Genova, B. Girma, E. S. Kissel, A. N. Levy, S. MacCracken, P. R. Mastrandrea & L. L. White (Eds.), *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change* (pp. 1199-1265). Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Nicholson-Cole, S., & O'Riordan, T. (2009). Adaptive governance for a changing coastline:
 Science, policy and publics in search of a sustainable future. In W. N. Adger, I.
 Lorenzoni & K. L. O'Brien (Eds.), *Adapting to climate change: thresholds, values, governance* (pp. 368-383). Cambridge, UK: Cambridge University Press.
- Nielsen, J. O., & D'haen, S. A. L. (2014). Asking about Climate Change: Reflections on Methodology in Qualitative Climate Change Research Published in Global Environmental Change since 2000. *Global Environmental Change*, 24, 402–409.
- Noble, I. R., Huq, S., Anokhin, Y. A., Carmin, J., Goudou, D., Lansigan, F. P., Osman-Elasha, B. & Villamizar, A. (2014). Adaptation needs and options. In C. B. Field, V. R. Barros, D. J. Dokken, K. J. Mach, M. D. Mastrandrea, T. E. Bilir, M. Chatterjee, K. L. Ebi, Y. O. Estrada, R. C. Genova, B. Girma, E. S. Kissel, A. N. Levy, S. MacCracken,

P. R. Mastrandrea & L. L. White (Eds.), *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects.* Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 833-868). Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.

- Oates, N., Conway, D., & Calow, R. (2011) The 'Mainstreaming' Approach to Climate Change Adaptation: Insights from Ethiopia's Water Sector. ODI Background Notes. London, UK: Overseas Development Institute.
- O'Brien, K., M. Pelling, A. Patwardhan, S. Hallegatte, A. Maskrey, T. Oki, U. Oswald-Spring, T. Wilbanks, & P.Z. Yanda (2012). Toward a Sustainable and Resilient Future. In C. B. Field, V. Barros, T. F. Stocker, D. Qin, D. J. Dokken, K. L. Ebi, M. D. Mastrandrea, K. J. Mach, G.-K. Plattner, S. K. Allen, M. Tignor & P. M. Midgley (Eds.), *Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change* (pp. 437-486). Cambridge, UK, and New York, NY, USA: Cambridge University Press.
- O'Riordan, T., & Jordan, A. (1999). Institutions, Climate Change and Cultural Theory: Towards a Common Analytical Framework. *Global Environmental Change*, 9, 81-93.
- Oberlack, C., & Eisenack, K. (2014). Alleviating Barriers to Urban Climate Change Adaptation through International Cooperation. *Global Environmental Change* 24, 349– 362.
- Oberlack, C., & Neumärker, B. (2011). Economics, Institutions and Adaptation to Climate Change *Constitutional Economics Network Working Paper Series CEN Paper No. 04-*2011. Breisgau, Germany: University of Freiburg.
- Office of the Presidential Assistant for Rehabilitation and Recovery [OPARR]. (2014). Yolanda Rehabilitation and Recovery Efforts. Manila, Philippines: Office of the Presidential Assistant for Rehabilitation and Recovery.
- Ogato, G. S. (2013). The Quest for Mainstreaming Climate Change Adaptation into Regional Planning of Least Developed Countries: Strategy Implications for Regions in Ethiopia. *Herald Journal of Geography and Regional Planning*, 2(2), 71-81.
- Olhoff, A., & Schaer, C. (2010). Screening Tools and Guidelines to Support the Mainstreaming of Climate Change Adaptation into Development Assistance – A Stocktaking Report. New York, USA: United Nations Development Programme.

- Onwuegbuzie, A. J., Johnson, R. B., & Collins, K. M. (2009). Call for Mixed Analysis: A Philosophical Framework for Combining Qualitative and Quantitative Approaches. *International Journal of Multiple Research Approaches*, 3(2), 114-139.
- Organisation for Economic Co-operation and Development [OECD]. (2006). Executive Summary. In S. Agrawala (Ed.), *Bridge Over Troubled Waters: Linking Climate Change and Development* (pp. 15-21). Paris, France: OECD Publishing.
 - ______. (2009). Integrating Climate Change Adaptation into Development Cooperation: Policy Guidance. Paris, France: Organisation for Economic Co-operation and Development.
- Orindi, V. A., & Eriksen, S. (2005). Mainstreaming Adaptation to Climate Change in the Development Process in Uganda. In J. W. Wakhungu (Ed.), *Ecopolicy Series*, no. 15. Nairobi, Kenya: African Centre for Technology Studies.
- Ostrom, E. (1990). Governing the Commons: The Evolution of Institutions for Collective Action. New York, NY: Cambridge University Press.
 - . (2007). Institutional Rational Choice: An Assessment of the Institutional Analysis and Development Framework. In P. A. Sabatier (Ed.), *Theories of the Policy Process, Second edition* (pp. 21-64). Colorado, USA: Westview Press.
 - ______. (2010). Polycentric Systems for Coping with Collective Action and Global Environmental Change. *Global Environmental Change*, 20, 550-557.
- ______. (2011). Background on the Institutional Analysis and Development Framework. *The Policy Studies Journal*, 39(1), 7-27.
- Pahl-Wostl, C. (2009). A Conceptual Framework for Analysing Adaptive Capacity and Multi-Level Learning Processes in Resource Governance Regimes. *Global Environmental Change*, 19, 354–365.
- Parry, M. L., Canziani, O. F., Palutikof, J. P., Linden, P. J. v. d., & Hanson, C. E. (Eds.). (2007). Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Pasquini, L., Cowling, R. M., & Ziervogel, G. (2013). Facing the Heat: Barriers to Mainstreaming Climate Change Adaptation in Local Government in the Western Cape Province, South Africa. *Habitat International*, 40, 225-232.
- Pasquini, L., & Shearing, C. (2014). Municipalities, Politics, and Climate Change: An Example of the Process of Institutionalizing an Environmental Agenda within Local Government. *The Journal of Environment and Development*, 23(2), 271–296.

- Persson, A., & Klein, R. J. T. (2008). Mainstreaming Adaptation to Climate Change into Official Development Assistance: Integration of Long-Term Climate Concerns and Short-Term Development Needs. Stockholm, Sweden: Stockholm Environment Institute.
- Pervin, M., Sultana, S., Phirum, A., Camara, I. F., Nzau, V. M., Phonnasane, V., Khounsy,
 P., Kaur, N., & Anderson, S. (2013). A Framework for Mainstreaming Climate
 Resilience into Development Planning *IIED Working Paper*, November 2013. London,
 UK: International Institute for Environment and Development.
- Petersen, A.C., Cath, A. Hage, M., Kunseler, E., & van der Sluijs, J.P. (2011). Post-normal Science in Practice at the Netherlands Environmental Assessment Agency. *Science*, *Technology, & Human Values*, 36(3), 362-388.
- Pettengell, C. (2010). Climate Change Adaptation: Enabling People Living in Poverty to Adapt *Oxfam International Research Report*. Oxford, UK: Oxfam International.
- Philippine Atmospheric Geophysical and Astronomical Services Administration [PAGASA].
 (2014). Tracks of Tropical Cyclones which Crossed the Province of Albay and 50 Kilometers from Boundaries from 1948-2013. Quezon City, Philippines: Philippine Atmospheric Geophysical and Astronomical Services Administration.
 - ______. (2015). About Us Retrieved 17 August 2015, from http://www.pagasa.dost.gov.ph/index.php/transparency/about-pagasa#brief-history.
- Philippine Statistics Authority [PSA]. (2014). Table 1. Projected Population, by Age Group, Sex, and by Five-Calendar Year Interval, Philippines: 2010 - 2045 (Medium Assumption) Retrieved 06 May 2015, from http://web0.psa.gov.ph/sites/default/files/attachments/hsd/pressrelease/Table1_8.pdf.
- Picketts, I. M., Déry, S. J., & Curry, J. A. (2014). Incorporating Climate Change Adaptation into Local Plans. *Journal of Environmental Planning and Management*, 57(7), 984-1002.
- Pini, B., River, S. W., & McKenzie, F. M. H. (2007). Factors Inhibiting Local Government Engagement in Environmental Sustainability: case studies from rural Australia. *Australian Geographer*, 38(2), 161-175.
- Plan International. (2010). Province of Albay, Philippines: Responding to the Challenge of Disaster Risk Reduction and Climate Change Adaptation. Provincial Government of Albay and the Center for Initiatives and Research on Climate Adaptation *Strengthening Climate Resilience-Case Study Summary Philippines*. Bangkok, Thailand: Plan International.

- Planning Commission. (2010). National Sustainable Development Strategy 2010-21. Dhaka, Bangladesh: Planning Commission, Government of the People's Republic of Bangladesh.
- Planning Commission. (2013). Twelfth Five Year Plan (2012–2017): Faster, More Inclusive and Sustainable Growth, Volume I. New Delhi, India: Planning Commission, Government of India.
- Pradhan, N. S., Khadgi, V. R., Schipper, L., Kaur, N., & Geoghegan, T. (2012). Role of Policy and Institutions in Local Adaptation to Climate Change: Case Studies on Responses to Too Much and too Little Water in the Hindu Kush Himalayas. Kathmandu, Nepal: ICIMOD.
- Preston, B. L., Westaway, R., Dessai, S., & Smith, T. F. (2009). Are we adapting to climate change? Research and methods for evaluating progress. Paper presented at the Greenhouse 2009: Climate Change and Resources, 23-26 March 2009, Perth, Australia.
- PreventionWeb. (2010). Philippines: Salceda 'UN senior global champion of DRR' Retrieved 24 June 2015, from http://www.preventionweb.net/english/professional/news/v.php?id=16235.
- Provincial Government of Albay [PGA]. (2007). Provincial Resolution No. 2007-24: Proclaiming the Province of Albay as One of the First and Pioneering Prototype Province that Will Adopt Climate Change Adaptation. Albay, Philippines: Provincial Government of Albay.
- Provincial Government of Albay [PGA] & Center for Initiatives and Research for Climate Adaptation [CIRCA]. (2010). Province of Albay, Philippines: Responding to the Challenge of Disaster Risk Reduction and Climate Change Adaptation SCR Regional Consultation. Bangkok, Thailand: Strengthening Climate Resilience.
- Pütz, M., Kruse, S., Casanova, E., & Butterling, M. (2011). Climate Change Fitness of Spatial Planning: WP5 Synthesis Report. Switzerland: European Territorial Cooperation Alpine Space Project CLISP.
- Quan, J., & Dyer, N. (2008). Climate Change and Land Tenure: The Implications for Land Tenure and Land Policy *Land Tenure Working Paper 2*. Rome: Food and Agriculture Organization.
- Ratner, B. D., Meinzen-Dick, R., May, C., & Haglund, E. (2013). Resource Conflict, Collective Action, and Resilience: An Analytical Framework. *International Journal of the Commons*, 7(1), 183–208.
- Rauken, T., Mydske, P. K., & Winsvold, M. (2015). Mainstreaming Climate Change Adaptation at the Local Level. *Local Environment*, 20(4), 408–423.

Ravetz, J.R. (1999). What is Post-Normal Science. Futures, 31, 647–653.

- Regional Development Council [RDC] XII (2012). RDC XII Resolution No. 69, series of 2012: Requesting NEDA Central Office to Fast Track the Completion of the Reference Manual on Mainstreaming Disaster Risk Reduction/Climate Change Adaptation in the Comprehensive Land Use Plans. General Santos City, Philippines: Regional Development Council (RDC) XII.
- ReliefWeb. (2014). CCC, HLURB rise to the challenge of climate change, adopt supplemental guidelines to integrate climate and disaster risks into local land use plans Retrieved 02 May 2014, from http://reliefweb.int/report/philippines/ccc-hlurb-risechallenge-climate-change-adopt-supplemental-guidelines-integrate.
- Republic of the Philippines. (1987). *Executive Order No. 292: The Administrative Code of 1987.* Manila, Philippines: Republic of the Philippines.
- . (1991). *The Local Government Code of the Philippines*. Manila, Philippines.
- _____. (2009). *Climate Change Act*. Manila, Philippines.
- ______. (2010). *Philippine Disaster Risk Reduction and Management Act*. Manila, Philippines.
 - _____. (2011). *People's Survival Fund*. Manila, Philippines.
- Republic of Zambia (2011). Sixth National Development Plan 2011-2015: Sustained Economic Growth and Poverty Reduction. Zambia: Ministry of Finance and National Planning, Republic of Zambia.
- Resurreccion, B. P., Sajor, E. E., & Fajber, E. (2008). Climate Adaptation in Asia: Knowledge Gaps and Research Issues in South East Asia: Full Report of the South East Asia Team: ISET-International and ISET-Nepal.
- Revi, A., Satterthwaite, D. E., Aragón-Durand, F., Corfee-Morlot, J., Kiunsi, R. B. R., Pelling, M., Roberts, D. C. & Solecki, W. (2014). Urban Areas. In C. B. Field, V. R. Barros, D. J. Dokken, K. J. Mach, M. D. Mastrandrea, T. E. Bilir, M. Chatterjee, K. L. Ebi, Y. O. Estrada, R. C. Genova, B. Girma, E. S. Kissel, A. N. Levy, S. MacCracken, P. R. Mastrandrea & L. L. White (Eds.), *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change* (pp. 535-612). Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Risbey, J., Baethgen, W., Martino, D., Agrawala, S., Gigli, S., Raksakulthai, V., Hemp, A., Moehner, A., Conway, D., El Raey, M. & Ahmed, A. U. (2006). Climate Change and Natural Resource Management. In S. Agrawala (Ed.), *Bridge Over Troubled Waters:*

Linking Climate Change and Development (pp. 85-132). Paris, France: OECD Publishing.

- Rittel, H. & Webber, M.M. (1973). Dilemmas in a General Theory of Planning. *Policy Sciences*, 4(2), 155–169.
- Roberts, D. (2008). Thinking Globally, Acting Locally Institutionalizing Climate Change at the Local Government Level in Durban, South Africa. *Environment & Urbanization*, 20(2), 521–537.
- Roberts, D. (2010). Prioritizing Climate Change Adaptation and Local Level Resilience in Durban, South Africa. *Environment & Urbanization*, 22(2), 397–413.
- Robichaud, I., & Wade, T. (2011). Planning and Policy in Atlantic Canada. Canada: Atlantic Climate Solutions Association.
- Rodima-Taylor, D. (2012). Social Innovation and Climate Adaptation: Local Collective Action in Diversifying Tanzania. *Applied Geography*, 33, 128-134.
- Rodima-Taylor, D., Olwig, M. F., & Chhetri, N. (2012). Adaptation as Innovation, Innovation as Adaptation: An Institutional Approach to Climate Change. *Applied Geography*, 33, 107-111.
- Rudd, M. A. (2004). An Institutional Framework for Designing and Monitoring Ecosystem-Based Fisheries Management Policy Experiments. *Ecological Economics*, 48, 109– 124.
- Sabatier, P. A., & Weible, C. M. (2007). The Advocacy Coalition Framework: Innovations and Clarifications. In P. A. Sabatier (Ed.), *Theories of the Policy Process, Second edition* (pp. 189-220). Colorado, USA: Westview Press.
- Salceda, J. S. (2012a). Adaptation Practices of Albay. Paper presented at the Presentation at the Training Workshop Climate Change for National & Local Governments in Southeast Asia and South Asia, 11-12 February 2012, Metro Manila, Philippines.

. (2012b) Adapting to Climate Change: Strategies of Albay, Philippines. *Agriculture and Development Notes on Climate Change Adaptation: Vol. 2.* Laguna, Philippines: Southeast Asian Regional Center for Graduate Study and Research in Agriculture.

- Salceda, J. S., & Rangasa, N. C. (2011). A Model Philippine Climate Change Adaptation Initiatives 2011. Albay, Philippines: Local Climate Change Adaptation for Development.
- Saloranta, T.M. (2001). Post-normal Science and the Global Climate Change Issue. *Climatic Change*, 50, 395–404.

- Santos, R. R. C. (1999). Cronbach's Alpha: A Tool for Assessing the Reliability of Scales. Journal of Extension, 37(2), 34-36.
- Schipper, L., Liu, W., Krawanchid, D., & Chanthy, S. (2010). Review of Climate Change Adaptation Methods and Tools *MRC Technical Paper No. 34*. Vientiane, Lao PDR: Mekong River Commission.
- Schmitz, O. J., Lawler, J. J., Beier, P., Groves, C., Knight, G., Boyce, Jr., D. A., Bulluck, J., Johnston, K. M., Klein, M. L. Muller, K., Pierce, D. J., Singleton, W. R., Strittholt, J. R., Theobald, D. M., Trombulak, S. C., & Trainor, A. (2015). Conserving Biodiversity: Practical Guidance about Climate Change Adaptation Approaches in Support of Landuse Planning. *Natural Areas Journal*, 35(1), 190-203.
- Scholz, R. W., & Tietje, O. (2002). Types of Case Studies. In R. W. Scholz & O. Tietje (Eds.), *Embedded Case Study Methods*. USA: SAGE Publications, Inc.
- Schreurs, M. A. (2008). From the Bottom Up: Local and Subnational Climate Change Politics. *The Journal of Environment and Development*, 17(4), 343-355.
- Secretariat of the Pacific Regional Environment Programme [SPREP] & United Nations Development Programme [UNDP] (2013). Mainstreaming Climate Change Adaptation in the Pacific: A Practical Guide. Apia, Samoa: Secretariat of the Pacific Regional Environment Programme and United Nations Development Programme.
- Serote, E. M. (2004). Property, Patrimony & Territory: Foundations of Land-Use Planning in the Philippines. Quezon City, Philippines: School of Urban and Regional Planning, University of the Philippines.
- Sharma, D., & Tomar, S. (2010). Mainstreaming Climate Change Adaptation in Indian cities. *Environment and Urbanization*, 22(2), 451–465.
- Siddiqui, N., & Fitzgerald, J. A. (2014). Elaborated Integration of Qualitative and Quantitative Perspectives in Mixed Methods Research: A Profound Enquiry into the Nursing Practice Environment. *International Journal of Multiple Research Approaches*, 8(2), 137-147.
- Sietz, D., Boschutz, M., & Klein, R. J. T. (2011). Mainstreaming Climate Adaptation into Development Assistance: Rationale, Institutional Barriers and Opportunities in Mozambique Environmental Science & Policy, 14(4), 493–502.
- Sinh, B. T., & Toan, V. C. (2012). Mainstreaming Adaptation into Local Development Plans in Vietnam Adaptation Knowledge Platform, Partner Report Series No. 3. Bangkok: Stockholm Environment Institute.
- Silberstein, M. A. J., & Maser, C. (2013). Land-Use Planning for Sustainable Development, Second Edition (Second ed.). Florida, USA: CRC Press.

- Sims, M. J., & Voltz, D. L. (2010). *Mainstreaming. Encyclopedia of African American Education. SAGE Publications, Inc.* Thousand Oaks, CA: SAGE Publications, Inc.
- Smajgl, A., Leitch, A., & Lynam, T. (2009). Outback institutions: An Application of the Institutional Analysis Aand Development (IAD) Framework to Four Case Studies in Australia's Outback DKCRC Report 31. Alice Springs, Australia: Desert Knowledge Cooperative Research Centre.
- Smit, B., & Wandel, J. (2006). Adaptation, Adaptive Capacity and Vulnerability. *Global Environmental Change*, 16, 282–292.
- Stevenson, A. (Ed.) (2010) Oxford Dictionary of English (3 ed.) Oxford University Press.
- Stewart, T. J., Janssen, R., & Herwijnen, M. v. (2004). A Genetic Algorithm Approach to Multi Objective Land Use Planning. *Computers & Operations Research*, 31, 2293– 2313.
- Tang, Z., Hussey, C. M., & Wei, T. (2009). Assessing Local Land-Use Planning's Awareness, Analysis, and Actions for Climate Change. International Journal of Climate Change Strategies and Management, 1(4), 368-381.
- Tavakol, M., & Dennick, R. (2011). Making Sense of Cronbach's Alpha. International Journal of Medical Education, 2, 53-55.
- Theesfeld, I., Schleyer, C., Hagedorn, K., Callois, J.-M., Aznar, O., & Olsson, J. A. (2010).
 The Institutional Dimension in Policy Assessment. In F. M. Brouwer & M. v. Ittersum (Eds.), *Environmental and Agricultural Modelling: Integrated Approaches for Policy Impact Assessment* (pp. 37-59). Dordrecht, The Netherlands: Springer Netherlands.
- Termeer, C., Dewulf, A., & Breeman, G. (2013). Governance of Wicked Climate Adaptation
 Problems. In J. Knieling & W. L. Filho (Eds.), *Climate Change Governance* (Vol. 27-39). New York, USA: Springer-Verlag Berlin Heidelberg.
- Uittenbroek, C. J., Janssen-Jansen, L. B., & Runhaar, H. A. C. (2013). Mainstreaming Climate Adaptation into Urban Planning: Overcoming Barriers, Seizing Opportunities and Evaluating the Results in Two Dutch case studies. *Regional Environmental Change*, 13, 399–411.
- United Nations Development Group [UNDG]. (2008). UNDG Capacity Assessment Methodology: User Guide for National Capacity Development. New York, USA: United Nations Development Group.
- United Nations Human Settlements Programme [UN-HABITAT]. (2010). Land and Natural Disasters: Guidance for Practitioners. Kenya: United Nations Human Settlements Programme.

- United Nations Development Programme [UNDP]. (2005). Measuring Capacities: An Illustrative Catalogue to Benchmarks and Indicators. New York, USA: United Nations Development Programme.
- United Nations Development Programme [UNDP]/Global Environment Facility [GEF]. (2003). Capacity Development Indicators: UNDP/GEF Resource Kit (No. 4) UNDP/GEF Resource Kit series. New York, USA: United Nations Development Programme.
- United Nations Development Programme-United Nations Environment Programme [UNDP-UNEP]. (2011). Mainstreaming Climate Change Adaptation into Development Planning: A Guide for Practitioners UNDP-UNEP Poverty-Environment Facility. New York, USA: United Nations Development Programme.
- United Nations International Strategy for Disaster Reduction [UNISDR]. (2012a). Champion Profile: Jose Clemente Salceda, Governor of Albay Province Retrieved 04 February 2015, from http://www.unisdr.org/campaign/resilientcities/champions/view/47682031.
 - ______. (2012b). Making Cities Resilient Report 2012: My City is Getting Ready! A Global Snapshot of How Local Governments Reduce Disaster Risk. Geneva, Switzerland: United Nations International Strategy for Disaster Reduction.
- United States Agency for International Development [USAID]. (2009). Adapting to Coastal Climate Change: A Guidebook for Development Planners. Washington DC, USA: United States Agency for International Development.
- Uy, N., Takeuchi, Y., & Shaw, R. (2011). Local Adaptation for Livelihood Resilience in Albay, Philippines. *Environmental Hazards* 10(2), 139-153.
- van Lier, H. N. (1998). The Role of Land Use Planning in Sustainable Rural Systems. Landscape and Urban Planning, 41, 83–91.
- Verburg, P. H., Schot, P. P., & Veldkamp, M. J. D. A. (2004). Land Use Change Modelling: Current Practice and Research Priorities. *GeoJournal*, 61, 309–324.
- Wang, J.-J. (2012). Integrated Model Combined Land-use Planning and Disaster Management: The Structure, Context and Contents. *Disaster Prevention and Management: An International Journal*, 21(1), 110 - 123.
- Ward, M. M. (2011). Analysis of global sea level rise impact and adaptation risk assessments. (Masters thesis), North Carolina, USA: Duke University.
- Waters, E., Barnett, J., & Puleston, A. (2014). Contrasting Perspectives on Barriers to Adaptation in Australian Climate Change Policy. *Climatic Change*, 124, 691–702.
- Weaver-Hightower, M. B. (2014). A Mixed Methods Approach for Identifying Influence on Public Policy. *Journal of Mixed Methods Research*, 8(2), 115–138.

- Wheeler, D. (2011). Quantifying Vulnerability to Climate Change: Implications for Adaptation Assistance *Center for Global Development Working Paper 240*.
 Washington DC, USA: Center for Global Development.
- Wong, P. P., Losada, I. J., Gattuso, J.-P., Hinkel, J., Khattabi, A., McInnes, K. L., Saito, Y. & Sallenger, A. (2014). Coastal systems and low-lying areas. In C. B. Field, V. R. Barros, D. J. Dokken, K. J. Mach, M. D. Mastrandrea, T. E. Bilir, M. Chatterjee, K. L. Ebi, Y. O. Estrada, R. C. Genova, B. Girma, E. S. Kissel, A. N. Levy, S. MacCracken, P. R. Mastrandrea & L. L. White (Eds.), *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 361-409). Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.*
- World Bank. (2006). Natural Disaster Hotspots: Case Studies. In M. Arnold, R. S. Chen, U.Deichmann, M. Dilley, A. L. Lerner-Lam, R. E. Pullen & Z. Trohanis (Eds.), DisasterRisk Management Series. Washington DC, USA: The World Bank.

_____. (2009). Convenient Solutions to an Inconvenient Truth: Ecosystem-based Approaches to Climate Change. Washington, DC, USA: The World Bank.

_____. (2013a). Getting a Grip... on Climate Change in the Philippines: Executive Report. Washington DC, USA: The World Bank.

. (2013b). Turn Down the Heat: Climate Extremes, Regional Impacts, and the Case of Resilience, A Report for the World Bank by the Potsdam Institute for Climate Impact Research and Climate Analytics. Washington DC, USA: The World Bank.

______. (2013c). World Development Report 2014: Risk and Opportunity- Managing Risk for Development. Washington DC, USA: The World Bank.

- Yin, R. K. (2014). Case Study Research: Design And Methods (5th ed.). Los Angeles, USA: Sage Publications.
- Young, O. R. (2002). *The Institutional Dimensions of Environmental Change: Fit, Interplay, and Scale.* London, England: MIT Press.
- Yumul, G. P. J., Cruz, N. A., Servando, N. T., & Dimalanta, C. B. (2011). Extreme Weather Events and Related Disasters in the Philippines, 2004–08: A Sign of What Climate Change Will Mean? *Disasters*, 35(2), 362–382.

APPENDIX

A1: Table A Mainstreaming indicator scores by scale

Mainstreaming indicator	National	Provincial	City/Municipal
Information capacity			
Availability of information	2.06	2.08	2.14
Access to information	2.11	2.15	2.17
Credibility and reliability of information	2.88	2.63	3.00
Communication of information	1.75	2.00	2.57
Translation of information	2.56	2.67	2.43
Knowledge and awareness	2.29	2.70	2.69
Institutional capacity			
Autonomy of local governments	2.67	2.62	2.00
Leadership	2.38	2.67	2.57
Commitment to climate change adaptation	2.22	2.71	2.43
Community support	2.22	2.58	2.70
Organisational cohesion	1.33	2.30	2.29
Organisational cooperation and collaboration arrangements	2.00	2.38	2.14
Local government prioritisation	2.56	2.77	2.57
Institutional issues	1.00	1.46	1.36
Institutional incentive	2.61	2.81	2.86
Resource capacity			
Availability of funds	2.33	2.69	2.43
Access to funds	2.11	2.50	2.43
Stability of funds	2.61	2.62	2.71
Availability of experts	2.63	2.50	2.71
Availability of human resources	2.11	2.38	2.57

Note: Indicator levels = $1^{st} - 1.0 \le n < 2$; $2^{nd} - 2 \le n < 2.25$; $3^{rd} - 2.25 \le n < 2.5$; 4th: $n \ge 2.5$

A2: Survey on the challenges of mainstreaming climate change adaptation into land-use planning

Background information

(1) How long have you been involved in local land-use planning? _____ months _____ years

Please put an X mark on the choice that best describes your answer. Unless specified in the question, a single answer is expected in each item.

- (2) In what capacity are you involved in local land-use planning? Please select all items that best describe your answer.
- ____ I draft the local land-use plan
- ____ I review the local land-use plan
- ____ I provide technical/sectoral information used in the land-use plan
- ____ I implement the land-use plan
- ____ I am involved in mapping
- ____ Others, please specify ____

Information capacity: Questions 3 to 11 pertain to information capacity for mainstreaming climate change adaptation into land-use planning.

- (3) What kinds of climate change related information, with specific focus on typhoons, are needed in mainstreaming climate change adaptation into land-use plans? Please select all items that best describe your answer.
- ____ Projected changes in frequency of typhoons
- ____ Projected changes in intensity of typhoons
- ____ Projected areas in danger or at risk from flooding due to typhoons
- ____ Projected extent of flooding caused by typhoons
- ____ Projected level/height of storm surges caused by typhoons
- ____ Projected impacts to marine and coastal ecosystems (e.g. coastal erosion)
- ___ Others __
- ____ I do not know

Other comments:

- (4) What is/are the main data source/s for the information identified in Question Item #3? Please select all items that best describe your answer.
- ____ PAGASA
- _____ www.Typhoon2000.ph
- ____ Manila Observatory
- _____ Local climate models
- _____ Marine Science Institute
- _____ National Institute of Geological Sciences
- _____ Climate Change Data Portal by the World Bank
- ____ Others, please specify _____

Other comments:

(5) Information identified in Question Item #3 are

____ not available.

- _____ available but limited or inadequate (e.g. not downscaled or localised).
- _____ available and comprehensive.
- ____ I do not know

Other comments:

(6) Information identified in Question Item #3 are available but/and

____ inaccessible.

____ partially accessible.

____ completely accessible.

____ I do not know.

Please cite reason/s for your answer:

- (7) The available information identified in Question Item #3 and are provided by agencies in Question Item #4 are
- _____ ignored because data and data source are not credible.
- _____ accepted with doubts, hence, used but not fully optimised in the land-use planning process.
- _____ accepted with confidence and are shared for effective use in relevant land-use decision-making and planning process.
- ____ I do not know.

Other comments:

(8) Climate experts, i.e., people producing climate related data identified in Question Item #3, and land-use planners

____ do not communicate with one another.

- ____ have one way communication i.e., experts provide information to planners without planners' input.
- ____ have two way communication or dialogue.
- ____ I do not know.

Other comments:

(9) Please rate the type of communication between the climate experts and the land-use planners

- ____ poor, i.e., planners are excluded from relevant dialogues regarding knowledge mobilisation/sharing
- _____ fair, i.e., there is irregular dialogue between experts and planners resulting to the information shared to be outdated
- ____ good, i.e., communication is inclusive and regular.

____ I do not know.

Other comments:

- (10) Available relevant climate information i.e., information identified in Question Item #3, as explained by climate experts (e.g., storm surge, climate change vulnerability, climate change sensitivity) are
- ____ not useful to planners.
- ____ partially useful to planners.
- ____ completely useful to planners.
- ____ I do not know.

Other comments:

(11) Please assess the knowledge (understanding of the implications of climate change) and awareness (familiarity or recognition of the existence of climate change) of the following regarding the climate risks related to land-use. Please put an X mark on the space provided for each category. Specifically, please rate whether the level of awareness is poor, fair, or good for each grouping (provincial development council, provincial land-use committee, etc.). Kindly do the same for the knowledge category.

	Awareness				Knowledge		
	Poor (not aware)	Fair (partially aware)	Good (completely aware)	Poor (not knowledge- able)	Fair (partially knowledge- able)	Good (completely knowledge- able)	I do not know
a) Provincial							
development council							
b) Provincial							
land-use committee							
c) Provincial							
land-use							
planners				<u> </u>			
d) City/							
development							
council							
e) City/							
municipal							
planning							
office							
f) Community							
members							

Institutional capacity: Questions 12 to 25 pertain to institutional issues related to mainstreaming climate change adaptation into local land-use planning.

- (12) Local government follows the climate change adaptation policies and frameworks provided by higher levels of government
- ____ but cannot modify/change them to suit local conditions.
- _____ and can modify/change them with approval from higher levels.
- _____ and can completely change them to suit local conditions without approval from higher levels.
- ____ I do not know.

nts:
nts:

(13) In devising local climate change adaptation policies, local governments have

____ no autonomy i.e., need higher level approval.

____ partial autonomy.

____ complete autonomy.

____ I do not know.

Other comments:

(14) A climate change adaptation champion i.e., staunch advocate, promoter, implementer, of climate change adaptation initiatives, in the locality

____ does not exist.

_____ exists but has limited influence on the behavior of the local community.

_____ exists and has significant influence on the behavior of the local community.

____ I do not know.

Who is/are the champion/s? Please provide name and contact information (e.g. agency and designation, address, or contact number)

(15) Local policies/regulations for local climate change adaptation

____ do not exist.

- _____ exist but are not sufficient to support the adaptation planning process (e.g. just provide framework/designate bodies).
- _____ exist and provide an environment that supports adaptation initiatives (e.g. provide general guidance and good linkage to investment plans).
 _____ I do not know.

Other comments:

(16) Please assess the following with regard to the support of community members on climate change adaptation initiatives of local governments. Please put an X mark on the space provided.

	Ignore them	Partially support them	Completely support them	I do not know
a) Provincial level				
c) City level				
d) Barangay level				

(17) There are a number of local organisations engaged in climate change adaptation, and their activities related to land-use and land-use planning are

_____ uncoordinated or lacking coordination.

- ____ coordinated but lacking consistency.
- _____ coordinated, consistent, and coherent with one another.
- ____ I do not know.

Other comments:

(18) Formal cooperation and collaboration mechanisms, i.e., Memorandum of Understanding (MOU), Memorandum of Agreement (MOA), between organisations engaged in climate change adaptation and land-use planning

____ do not exist.

- ____ exist but not always operational.
- _____ exist and are effectively operating.
- ____ I do not know.

Please provide examples:

(19) Informal cooperation and collaboration mechanisms between organisations engaged in climate change adaptation and land-use planning

____ do not exist.

- _____ exist but not always operational.
- _____ exist and are effectively operating.

____ I do not know.

Please provide examples:

(20) Climate change adaptation is

____ not in the local government agenda because there are more important issues.

_____ in the local government agenda but underrepresented because there are more important issues.

_____ a priority local government agenda.

____ I do not know.

Other comments:

(21) Unresolved institutional issues concerning land-use and land-use planning

_____ exist, affect, and have an impact on the adaptation approach.

_____ exist but do not affect the adaptation approach.

____ do not exist.

____ I do not know.

Other comments:

(22) The existing unresolved institutional issues (described in Question item #21) concerning landuse and land-use planning are

____ not being addressed.

- ____ partially being addressed.
- ____ completely (all are) being addressed.

____ I do not know.

Please provide examples:

(23) Other institutional issues i.e., political in nature

_____ exist and are affecting the adaptation approach.

_____ exist but are not affecting the adaptation approach.

____ do not exist.

____ I do not know.

Other comments:

(24) Local governments perceive the economic, social, and environmental advantages from integrating climate change adaptation into the land-use plans as

____ not beneficial.

____ somewhat beneficial.

____ most beneficial.

____ I do not know.

Other comments:

(25) The benefits from mainstreaming or integrating climate change adaptation into land-use plans

- _____ do not encourage local government units to implement the approach.
- _____ somewhat encourage local government units to implement the approach.
- _____ definitely encourage local government units to implement the approach.

____ I do not know.

Other comments:

Resource capacity: Questions 26 to 30 pertain to financial and human resources related to mainstreaming climate change adaptation into local land-use planning.

- (26) Funds to support activities in mainstreaming climate change adaptation into land-use plans are
- ____ not available.
- _____ sometimes available, i.e., on project basis; depends on foreign or private funding.
- _____ always available i.e., part of the national/local government budget.
- ____ I do not know.

Other comments:

(27) The available funds for mainstreaming climate change adaptation into development efforts are

- ____ not accessible.
- _____ partially accessible (e.g. available but access to it is not well-explained).
- ____ completely accessible.
- ____ I do not know.

Please cite reason/s for your answer.

- (28) Funds to support initiatives for mainstreaming climate change adaptation into land-use plans is/are
- _____ a one-time occurrence.
- _____ irregular (e.g. available when events related to climate change impacts take place).
- ____ part of the local/national budget.
- ____ I do not know.

Other comments:

(29) Experts on local climate change risks and impacts on land-use that can train local land-use planners to integrate climate change into local land-use plans are

____ not available.

- _____ available but lack skills to fully train land-use planners.
- _____ available with skills to fully train local land-use planners.
- ____ I do not know.

Other comments:

(30) The local staff (i.e., administrative personnel, research staff, etc.) to undertake tasks and responsibilities of current and future climate change adaptation and land-use planning activities are

____ not available.

- _____ available on an irregular basis.
- ____ available on a regular basis.
- ____ I do not know.

Other comments:

Additional information

- (31) Please include any other relevant information concerning mainstreaming climate change adaptation into local land-use planning that you would like to share.
- (32) Can you recommend the name of anyone else who should participate in this survey? Please provide name and contact details (e.g. designation and agency, address or phone number).
- (33) Please write your name, designation, and agency/office. This set of information is confidential and will only be used as reference for the data analysis.

Name: _____

Designation:

Agency/Office: _____

A3: Final interview schedule as revised in the field

EXAMINING THE CHALLENGES OF MAINSTREAMING CLIMATE CHANGE ADAPTATION INTO LOCAL LAND-USE PLANNING: THE CASE OF ALBAY, PHILIPPINES

Section 1: Climate change adaptation in the Philippines

- (1) Compared to the other adaptation initiatives being implemented in the Philippines, how important is mainstreaming or integrating climate change adaptation into the comprehensive land-use planning process? Please rate it on a scale of 1 to 5, with 1 being the lowest priority or importance and 5 the highest. Please explain why it should be given a high, medium, or low priority.
- (2) Please describe the mind-set of the local government unit planners in terms of integrating or mainstreaming climate change adaptation into land-use planning. In your opinion, how to the planners perceive the importance of the mainstreaming endeavour? Please rate their perception on a scale of 1 to 5, with 1 being the lowest priority or importance and 5 the highest. Please explain why you gave this rating.

Section 2: Institutions in the land-use planning process

- (3) In your experience, what institutional issues or challenges affect the mainstreaming endeavour? How do these issues affect the mainstreaming process?
- (4) Do you have any other additional thoughts regarding institutional issues that you would like to share?
- (5) Please identify the institutions i.e., government and civil society, that are important in the formulation of the comprehensive land-use plans and describe the role or responsibilities of these institutions in the planning process.
- (6) How do they affect the mainstreaming of climate change adaptation into the comprehensive landuse plans?

Section 3: Climate change champion

- (7) Is there a climate change champion in Albay (at the national scale)?
- (8) How does the champion influence the mainstreaming of climate change adaptation into local land-use planning? Can you cite specific examples?

Section 4: Availability of human resources to undertake climate change adaptation tasks

(9) Please describe the conditions in the local government units in terms of the available personnel that undertake the responsibilities relating to climate change adaptation, specifically that of integrating adaptation into the comprehensive land-use plans.

Section 5: Summary

- (10) How do you rate the overall progress of mainstreaming climate change adaptation into the comprehensive land-use plan? Please rate at a scale of 1 to 5 with 1 being the lowest and 5 the highest and please explain your rating.
- (11) In your opinion, what is the key factor needed to generate a comprehensive land-use plans with climate change components?

Section 6: Closing

Thank you very much for participating in this interview. Before we conclude it:

- (12) Do you have any other thoughts regarding the mainstreaming approach or land-use planning processes that you would like to share that were not covered by the interview?
- (13) Can you think of any other person/organisations which would be significant to this research that I could interview?

Thank you very much. You have been a very valuable part of my research.

A4: An Analytical Framework for Investigating Complex Institutions in Climate Change Adaptation: The Institutional Environment Matrix

Cuevas, S., Peterson, A., & Morrison, T. (2014). An Analytical Framework for Investigating Complex Institutions in Climate Change Adaptation: The Institutional Environment Matrix. In W. Leal Filho (Ed.), Handbook of Climate Change Adaptation (pp. 1-22). Berlin: Springer Berlin Heidelberg.

Abstract

This Chapter introduces the Institutional Environment Matrix (IEM), a diagnostic and planning framework designed to analyze complex institutional environments and determine the institutional fit of climate adaptation responses. The framework argues that the institutional environment is comprised of rules, social structures, and organizations. It establishes the vital role of institutional arrangements in characterizing the functions and functional interdependencies of institutions. The IEM framework has a dual layer design that allows complex institutional relationships to be examined across scales. The institutional environment layer is a comprehensive inventory of institutions that outlines institutional complexities. The institutional matrix layer is the system of institutional arrangements that determines the functional interdependencies of institutions. The matrix explores institutional interplay in relation to several general institutional functions: reducing uncertainty, connecting individuals to society, fostering adaptive capacity, and mobilizing resource utilization. By providing a structure to examine complex institutional relationships, the IEM is a significant innovation for assessing the institutional fit of and interplay between existing and planned climate change adaptation responses. This framework may also be used as an analytical tool in adaptation planning and evaluation.

Keywords

Institutions; Institutional environment; Climate change adaptation; Analytical framework

Introduction

Urban and rural systems are increasingly subject to complex and uncertain problems such as climate change, biodiversity loss, land-use conflict, pandemic disease, and rapid market fluctuations. These problems challenge the abilities of societies to manage change in traditional ways. Subjective perceptions, cross-sectoral misunderstandings, and technological contingencies only increase this challenge (Gunderson and Holling 2002, Crowder et al. 2006). Institutions play a critical role in ensuring successful adaptation to rapid and unpredictable change, yet are one of the least examined and ambiguous aspects of climate adaptation (O'Riordan and Jordan 1999; Adger 2000a, Adger et al. 2005a, 2005b).

Following the works of North (1990) and Ostrom (1990), several facets of institutions have been examined in the literature (Young 2002, Sabatier 2007, Oberthür and Stokke 2011). Recently, of particular interest to scholars is the linkages between institutions, climate change, and adaptation with studies addressing the effects of institutional barriers and constraints on adaptation (Inderberg and Eikeland 2009), the fundamental functions of institutions in facilitating climate adaptation (Rodima-Taylor 2012), and the institutional requirements for adaptation (Adger et al. 2005). Yet, despite these works, the research area is still in its infancy, as evidenced by a number of competing frameworks.

To analyze these institutional concerns more effectively, scholars have developed a variety of frameworks to examine the role of institutions in the context of climate change adaptation. These analytical frameworks are differentiated by how they define institutions—

in essence, whether they consider institutions as rules, social patterns of behaviors, or organizations. In other words, the types of analysis that can be performed using these frameworks are bound by the frameworks' institutional perceptions.

For example, one framework that focuses on organizational institutions examines the institutional linkages between and among public, private, and civil society institutions and the significance of institutional partnerships in enabling adaptation. It provides a tool to analyze organizational partnerships and the impacts of these associations on the access of vulnerable social groups to resources (Agrawal 2008). Another framework that defines institutions in terms of rules, customs, and norms concentrates on the intrinsic characteristics of institutions in influencing the behaviors of individuals and in fostering collective action. Essentially, this framework deals with understanding and assessing the ability of institutions to raise the adaptive capacity of society (Gupta et al. 2010).

This raises two significant issues. First is the disharmony in defining institutions in the context of climate change. Research has focused on institutions as rules and social-structures (O'Riordan and Jordan 1999, Eriksen and Selboe 2012), and also as organizations (Agrawal et al. 2008, Vallejo 2011). Therefore, as climate change research advances, there is a discrepancy on how the concept of institutions is founded. Second is the inability of the frameworks to simultaneously analyze the various facets of institutions. If these gaps are not addressed, it can lead to a divergence in institutional concepts and the direction of research.

To address the first concern a conceptual framework was developed that defines institutions as a triad of rules, social structures, and organizations in the context of climate change adaptation. Thus, institutions are the commonly known and acknowledged rules, social structures, and organizations founded on common belief systems that transform individual acts and expectations into collective actions; convert personal values into social norms and shared beliefs; and define the formal and informal behavioral systems of human existence. As an extension of this endeavor, this Chapter develops an analytical framework that helps to examine the complexity of the triad institutions in climate change adaptation responses.

Institutional interventions formulated and implemented to adapt to climate change bring either conflict or harmony into existing institutions and arrangements (Young 2002, Nilsson et al. 2012). A framework that can be utilized to examine the relationships between and among rule-based, social structure-based, and organizational institutions is useful in planning for and evaluating the effects of these institutional changes. Moreover, the efficiency and success of adaptation responses rest on how they fit in the institutional environment and institutional arrangements that are in place (Theesfeld et al. 2010). Every case has a unique institutional environment or *array of institutions that influence and affect climate change adaptation behaviors and decisions*. Therefore, the institutional fit of the adaptation measures, i.e., *whether adaptive institutional interventions are synchronized or in harmony with the existing institutions*, is vital to effectively implement an adaptation response. The more fitting the adaptive institutions are with the other institutions in the system, the better each institution performs, and thus, the more relevant each one becomes.

This Chapter is divided into two major parts. The first establishes the theoretical foundations of the framework by presenting the concept of institutions as rules, social structures, and organizations in the context of climate change adaptation (Section 2); examining the concepts of institutional arrangements (Section 3); and classifying institutional functions (Section 4). The second part synthesizes these concepts (Section 5) to form a framework termed the Institutional Environment Matrix (IEM). In developing the IEM, this

paper adopts the definitions established by other authors and develops some new definitions befitting the context in which they are used (Table 1).

Term	Definition
Rule-based	Constraints that structure political, economic and social interactions,
institutions	and determine decisions, actions, information, pay-offs and actors in
	various conditions and situations (North 1990, Ostrom 1990)
Social structure-	Self-sustaining, salient patterns of social interactions (Aoki 2007) that
based institutions	form individual and social expectations, relations, conduct,
	interactions, and behavior (Agrawal 2008)
Organizational	Structures of power that form the social, economic, legal, and political
institutions	organizations of a society (O'Riordan and Jordan 1999, Acemoglu and
	Johnson 2005)
Institutional	Structure of the rules that govern human decisions (Tang 1991) or the
arrangements	specific guidelines designed to facilitate social interactions (Klein
	2000)
Institutional fit	State where the adaptive institutional interventions are synchronized
	or in harmony with the existing triad of institutions – rules, social
	structures, and organizations
Institutional	Interactions and reactions between and among institutions that build
interplay	institutional linkages, relationships, and interdependencies
Functional	Relationships built resulting from the interactions among the
interdependencies	arrangements that allow institutions to perform their functions
Institutional	Comprehensive inventory of the differing institutions—rules, social-
Environment (IE)	structures, and organizations—that may influence adaptation
	responses
Institutional Matrix	System of institutional arrangements that determines the functional
(IM)	interdependencies of institutions
Institutional	A planning and diagnostic framework designed to analyze institutional
Environment Matrix	environments and determine the institutional fit of climate adaptation
(IEM) Framework	responses

Table 1 Definition of key terms

Sources: North (1990), Ostrom (1990), Tang (1991), O'Riordan and Jordan (1999), Klein (2000), Acemoglu and Johnson (2005), Aoki (2007), Agrawal (2008)

Institutions in Climate Change Adaptation Planning

There are varying notions of what constitutes an institution. Institutions are rules, procedures, conventions, and protocols in rational choice, economics, and game theory; moral templates, cognitive scripts, and frames of meaning in sociology and anthropology; and organizations in comparative politics and state theory (North 1990, Jordan and O'Riordan 1997, O'Riordan and Jordan 1999, Markvart 2009).

In climate change adaptation, institutions should have a synthesis of definition that has cross-disciplinary relevance. Therefore, institutions are the commonly known and acknowledged rules, social structures, and organizations founded on common belief systems that transform individual acts and expectations into collective actions; convert personal values into social norms and shared beliefs; and define the formal and informal behavioral systems of human existence. Hence, rules, social structures, and organizations are all institutions. The core relationships among the three forms of institutions are illustrated in the Venn diagram (Figure 1). Rules, social structures, and organizations are linked through a system of beliefs that allow them to exist and continue to persist as institutions. The beliefs associated with rules and organizations are significant components shaping the self-enforcing expectations, which consequently affect behavior and motivate individual actions (Greif and Kinston 2011). The belief system itself is a part of what constitutes social structure-based institutions (Nelson and Sampat 2001, Nelson and Nelson 2002).





Rule-based and social structure-based institutions are further linked through informal rules. This linkage is shown in the overlap between the spheres representing rule-based and social structure-based institutions (Figure 1). Together with formal rules, informal rules such as practices, norms, and traditions comprise part of the rule-based institutions. Formal rules define the hierarchical structure, decision-making powers, contracts and property rights allocation in the political and economic systems (Pejovich 1995). Meanwhile, informal rules determine individual interactions and are engraved in society's culture and heritage (North 1990, Hasan 2000, Hodgson 2006). These same social patterns of behaviors are the elements that form social structure-based institutions (Nelson and Sampat 2001, Aoki 2007).

The conceptual framework considers organizations as an assembly of rules and contracts that operate through some type of relationship among individuals. This perception interlaces organizations—perceived as a "collection of rules" (March et al. 2011: 239)—with the rule-based institutions. Organizations rely on the norms and patterns of behaviors to be implemented (Hodgson 2006), which ties organizations to the social structure-based institutions.

The formal and informal structures of organizational institutions are rationalized by the formal rules and social customs, values, and beliefs, respectively (Meyer and Rowan 1977, Shafritz et al. 2005). Formal rules are the written and legally sanctioned rules, whereas informal rules are represented by the unwritten social patterns of interactions and behaviors (Nabli and Nugent 1989). Formal rules are usually applied, managed, observed, and monitored by formal political, legal, and government institutions. Conversely, informal institutions fall under the private realm (Williamson 2009).

Formal structures are particularly associated with informal rules (illustrated by the broken arrow linking the two factors) (Figure 1). Formal organizations are created and legitimized by formal rules, whereas the relationships among the members of formal organizations are typically governed by informal rules. Hence, "in every formal organization, there arise informal organizations" (Shafritz et al. 2005: 205), thereby forming an additional linkage between formal and informal organizations (indicated by the broken arrow between the two entities). This relationship denotes that informal organizations do not directly affect the structure, composition, or creation of formal organizations. However, informal organizations are defined as collective behaviors (in the form of organized groups of people) that influence the choices and decisions of the formal organizations' members. Meanwhile, formal organizations directly affect informal organizations through the formal rules they implement or the actions they perform (solid arrow in Figure 1). In essence, "the root of informal systems are imbedded in the formal organization itself and nurtured by the very formality of its arrangements" (Shafritz et al. 2005: 205). For example, a formal organization that funds the activities of an informal organization may influence the latter to become formal itself, especially if formality is a requirement to gain further financial assistance. Alternatively, the funds provided may have allowed the informal organization to expand its operations, with the transition to a formal structure being essential to continue these new activities.

This unified definition of institutions is vital in analyzing linkages among climate adaptive institutions. Climate change impacts include extensive aspects of human existence (i.e., social, economic, political, ecological, and environmental). Hence an amalgam of ideas from various disciplines befits the concept of institution in the climate change adaptation context. This synthesized definition should be further investigated, particularly in relation to how rules, social structures, and organizational institutions function together in systems where adaptation responses are applied.

Institutional Analysis and Institutional Arrangements

Institutional arrangements are critical to address the climate change challenge as adaptation "never occurs in an institutional vacuum" (Agrawal et al. 2008: 2). The success of adaptation practices rests on specific institutional arrangements, such as well-defined property rights that address resource access and risk exposure (Agrawal 2008). For example, building a seawall would not depend only on the physical construction of the structure itself, the costs associated with it, or the science that projects the rate of sea level rise. It also would be affected by the rules governing property (Caldwell and Segall 2007), including agreements on the allowable height, thickness, and length of the structure; the social norms of the communities affected by the predicted sea level rise and storm surge; and the rights of private property owners. Therefore, developing suitable adaptation responses entails institutional arrangements that enable these measures to be implemented (Rodima-Taylor et al. 2012).

Arrangements in rule-based and social structure-based institutions

Institutional analysis assumes that institutional change will affect some areas of reality that already are exposed to existing institutions. Therefore, the environment where the institutional changes (e.g., the creation of new policies or amendments in prevailing regulations) are to be implemented must be understood first before the possible consequences of such changes can be determined (Theesfeld et al. 2010). More importantly, intensive institutional analysis involves understanding the detailed working rules⁴ and norms that influence people's decisions (Ostrom 2011).

In **rule-based** and **social structure-based** institutions, these rules exist at three levels, namely, operational rules, collective-choice rules, and constitutional-choice rules (Ostrom 1990). Among the three, constitutional-choice rules are the most extensive. They are the basis of all rules—the set which determines who and what (specific rules) are authorized to create the other levels of rules. Next in the hierarchy are the collective-choice rules—the rules created to resolve conflicts, impose decisions, and formulate or transform operational rules (Ostrom 1990). Essentially, they are the rules which underpin operational rules (Tang 1991). Lastly, operational rules are those that directly influence the daily decisions about who oversees the actions of others and how or who takes part in which situation, what information must be given, what are the participants allowed to do, and what rewards or penalties will be designated to various sets of acts and consequences. Operational rules are typically known as policies (Ostrom 1990).

Ostrom's (1990) classic text, "Governing the Commons", identified specific processes at each level of rules. Operational rules cover appropriation, provision, monitoring, and enforcement; the collective-choice level encompasses policy-making, management, and mediation of policy decisions; and the constitutional level includes formulation, governance, adjudication, and modification of constitutional decisions. Through these rules, institutions are able to affect and influence individual and collective actions. For instance, operational rules, such as those that specify fishing technologies permitted at a particular fishing ground, constrain and predict operational actions. Similarly, collective-choice rules are translated into collective-choice actions, and constitutional rules into constitutional-choice actions (Schlager and Ostrom 1992). These levels of rules form the categories of institutions (Feder and Feeny 1991). The constitutional-choice rules comprise the constitutional order, whereas the *collective-choice rules and operational rules constitute institutional arrangements* (Figure 2) (Feder and Feeny 1991, Tang 1991).

Institutional arrangements are the structure of rules governing human decisions (Tang 1991) or the specific guidelines which facilitate social interactions (Klein 2000). Institutional arrangements are also sets of rules or agreements with a common objective (e.g., contract) that preside over the activities of people. For instance, a group of farmers may enter into an agreement to jointly purchase agricultural inputs or supply products to buyers, thus forming a producer's organization (Eaton et al. 2008). Institutional arrangements likewise identify an individual in relation to others within the group that s/he belongs to, as well as with those outside the group. For example, in property regimes, the property relation between individuals is defined by the interest of one that is protected by virtue of the right and the duty of others to follow the arrangement (Bromley and Cernea 1989). Institutional arrangements, therefore, guide individual behaviors towards collective actions.

⁴ Working rules are the set of rules that rationalize and justify the decisions and actions of people.



Figure 2 Institutions and institutional arrangements (Source: Authors)

Arrangements in organizational institutions

In terms of **institutional organizations**, institutional arrangements involve the system of (organizational) units that plan, support, and/or implement programs, practices, and actions. These arrangements include the linkages between and among organizations at different administrative scales (national, regional, state, provincial and local) or sectors (economic, political, legal, social). They also represent relationships between government and non-government units such as households, communities, and civic organizations (Mattingly 2002). As institutions, organizations are governing structures that motivate collective behaviors and actions (Nelson and Sampat 2001, Williamson 2009), while institutional arrangements are the governance arrangements (Klein 2000, Kooiman 2008). Institutional arrangements oversee the relationships and interactions between, among, and within groups of individuals, and thus, influence the variability of commitments of institutions to governance (Klein 2000, Andersson and Ostrom 2008). These ideas are significant because they link organizational arrangements to the rule-based and social structure-based arrangements⁵ (Fig. 2). To illustrate, policy goals depend on the set of dominant actors and ideas in the area, and when these policy debates and decision-making occur. Governance arrangements then determine the aims and the general implementation preferences of policies, regulations, and state-society interactions (Howlett 2009).

Institutional arrangements allow multiple types of linkages between and among institutions (Heikkila et al. 2011). As guidelines, they are the means by which institutional interplay (i.e., in the form of functional interdependencies or consequences of institutional design and management) is implemented (Young 2002). Accordingly, institutional interplay refers to the interactions among institutions that build institutional relationships (Young 2002). Institutional interaction is determined by the impact of one institution on another,

⁵ This relationship is represented by the broken arrow in Fig. 2, linking governance arrangements with the collective-choice actions.

thereby exhibiting causation (Gehring and Oberthur 2009, Oberthur and Stokke 2011). Thus, the effect or interaction will not be observed without a stimulus and a receiver. The stimulus is the source institution (independent variable) and the receiver is the target institution or system (dependent variable) (Gehring and Oberthur 2009). For example, an introduced institutional adaptive measure (the independent variable) will interact with the existing institutions in the system (the dependent variable). Institutional interplay, however, is not one-directional. Interplay involves functional interdependencies (Young 2002, Linner 2006), and thus include mutual influences or effects. Although the interaction may be triggered by a stimulus, the outcomes or institutional linkages are the result of the institutional integration. Therefore, the interplay exists in the institutional environment comprised of the adaptive institution and the other existing institutions.

Institutional interactions may be complementary, neutral or co-existing, counterproductive, conflicting, or overlapping (Gunningham and Grabosky1998, Young 2002, Nilsson et al. 2012). In relation to the triad institutional concept introduced in this chapter, institutional linkages encompass relationships within, between, and among laws, policies, regulations, traditions, norms, practices, government units, civic organizations, and community groups, among others. Institutional linkages must be understood to determine how institutions influence adaptation practices and responses (Agrawal 2008). Consequently, institutional functions in the context of this institutional definition should be determined to understand the institutional fit of adaptation responses.

Institutional Functions

Institutions are crucial in promoting successful adaptation to climate change. They influence key decisions in the system, shape the direction of adaptation efforts, frame the adaptive capacities of systems, and enable collective action toward attaining the adaptation goals (Naess et al. 2005, Agrawal et al. 2008, Eriksen and Selboe 2012). Institutions accomplish these tasks through their innate characteristics and the functions they perform.

Institutions, as rules, social structures, and organizations serve the same functions, thus they are interdependent. These institutional functions can be classified into four main types, namely, reducing uncertainty (by forming individual and social expectations), connecting individuals to society, fostering adaptive capacity, and mobilizing resource utilization. These functions are performed by all institutional types, thereby, strengthening the interconnections among these institutions (Figure 3).

The different characteristics of rules, social structures, and organizations limit their respective capabilities to perform some of these functions. For example, organizations (as a collection of rules and bundles of relationships and interactions) can establish systems of power and authority, and identify the people included and excluded from the organization. However, the entirety of organizations, including their characteristic as "actors" (Gupta et al. 2010), constrains the organizations' ability to create rights and entitlements, but promotes their capacity to deliver external resources into the system (Table 1).



Figure 3 Functions of institutions (Source: Authors)

Reduce uncertainty

Institutions **reduce uncertainty** by forming individual and collective expectations and by developing a constant structure of social interactions (North 1990, Ostrom 1990, Kirsten et al. 2009, Brousseau et al. 2011). They also provide stability and predictability by *establishing the power and authority systems* (O'Riordan and Jordan 1999, Acemoglu and Johnson 2005, Berman et al., 2012), and *creating rights and entitlements* (Ostrom 1990). Institutions also *identify inclusions and exclusions* by determining which actions are permissible and the conditions by which to undertake certain activities (North 1990, Ostrom 1990, and Klein 2000). By outlining constraints, institutions set up the boundaries for each individual and society as a whole (Ostrom, 1990). For example, as an institution, property rights form expectations that the claims to the property would be respected and be abided by all, thereby reducing the uncertainty associated to these claims (Bromley and Cernea 1989).

	Types of Institutions					_	
Functions of an	Rules		Social	Social Organizat			
Institution	Formal	Informal	Structures	Formal	Informal	Features/ Descriptions	
Reduces uncertainty						-	
Establishes systems of power and authority	✓	✓	✓	~	✓	Actors and actions involved in decision- making; who has the authority and what kind of authority	
Identifies inclusions and exclusions	\checkmark	√	V	~	\checkmark	Scope and jurisdiction of actors (who) and actions (what) allowed and constrained	
Creates rights and entitlements	✓	√	\checkmark	×	×	Claims, privileges, etc., to resources, i.e., access rights, management rights	
Connects individuals to so	ciety						
Converts personal values into social norms and shared beliefs	✓	✓	✓	~	✓	Social principles, beliefs, and philosophies	
Influences and transforms individual acts and expectations into collective actions	✓	~	✓	~	~	Plans and programs for collective efforts and actions	
Creates (dis)incentives for individual and collective actions	✓	\checkmark	\checkmark	✓	~	Rewards and penalties; pay-offs on actions	
Coordinates individual or collective behaviors	\checkmark	V	√	~	\checkmark	Management of multiple efforts	
Foster adaptive capacity							
Defines information systems	\checkmark	\checkmark	\checkmark	~	\checkmark	Provision of information (who, what, when, where, and for whom)	
Mediates external interventions	✓	~	✓	~	✓	Access to and management of outside resources (how, what, when, where)	
Mobilizes resource utilizat	ion						
Means of delivery of external resources	×	×	×	\checkmark	\checkmark	Actors facilitating access to outside resources (who and for whom)	
Determines transaction costs of activities and decisions	\checkmark	\checkmark	\checkmark	~	\checkmark	Integration of multiple efforts; internal costs (financial or otherwise)	

Table 1 Institutional functions by type of institution

Sources: North (1990, 1994), Ostrom (1990), O'Riordan and Jordan (1999), Jentoft (2004), Acemoglu and Johnson (2005), Pfahl (2005), Agrawal (2008), Adkisson (2009), Dorward and Omamo (2009), Kirsten et al. (2009), Gupta et al. (2010), Greif and Kingston (2011), and Berman et al. (2012)

Connects individuals to society

Institutions **connect individuals to society** by giving everyone a shared identity (Jentoft 2004). They *convert personal values into social norms and shared beliefs* as individuals get emotionally attached and identify with the institutions. This function is specifically attributed to social structures, informal rules, and informal organizations. Thus, institutions become the social standard for understanding and reacting to circumstances, which accordingly become the source of people's compliance and submission to institutions. As such, institutions *influence and transform individual acts and expectations into collective actions* (Kirsten et al. 2009).

Institutions also *create the incentive structure* that determines the actions of people as individuals and as a society (Agrawal 2008). Incentive structures incorporate behavioral patterns that encourage actors to change norms and practices, implement the changes, uphold the changes, and stand by the decisions to change (Biermann 2007, Gupta et al. 2010). Thus, incentive mechanisms enable individuals to choose how to respond efficiently to goals and objectives such as those of climate change adaptation (Young 2002). Conversely, institutions also influence behaviors by providing disincentives or penalties to various actions and consequences (Ostrom 1990). Thus, choosing to conform to institutional arrangements becomes attractive.

Institutions are also the *means by which people coordinate their beliefs, interactions,* and activities, thereby affecting how individuals and society make decisions (Pfahl 2005, Adkisson 2009, Greif and Kingston 2011). For instance, the local governing body in the Carteret Island in Papua New Guinea (i.e., the Council of Elders) organized the voluntary relocation of community households to the main island of Bougainville (Rakova 2009). Labeled as one of the first climate change refugees, the Carteret people were forced to leave their homes due to the accelerated sea level rise and the worsening extreme coastal events in the area (Boano et al. 2008). The Council formed a non-government organization, named Tulele Peisa, which designed and administered the Carterets Integrated Relocation Programme (CIRP) (Rakova 2009, Boege 2011). In this case, the organizational institutions such as the Council of Elders and the Tulele Peisa were vital in planning and mobilizing the relocation efforts. The community's norms and traditions authorized the organizations (specifically the Council of Elders) to make decisions for the whole community. Meanwhile, the CIRP guided the people on how to follow through with the community resettlement. Thus, the social and cultural norms, organizations, and formal policies all affect how an individual, a household, and/or a community responds to climatic and other stressors (Young 2002).

Fosters adaptive capacity

Institutions are critical in **building the adaptive capacities of systems** (Berman et al. 2012, Pradhan et al. 2012). They *affect information systems* (Dorward and Omamo 2009) and strengthen the ability of vulnerable communities to prepare for the impacts of climate change. They influence the flow of information, the types of studies undertaken, and the interpretations made from the research results (March and Olsen 1996). Moreover, they influence the kind of information to be disseminated (Ostrom 1990) and how this knowledge is disseminated (Agrawal 2008).

Institutions are *mediators of external interventions* that affect how individuals, communities, and social groups utilize assets and resources (Agrawal 2008). Institutions provide leadership, facilitate negotiations, and create networks with other institutions such
that external interventions can be systematically filtered, effectively absorbed, accepted, or refused (Agrawal 2008, Rodima-Taylor 2012). For example, a culture of solid community ties suggests an accommodating attitude for external interventions promoting community-based management; but the reverse can be expected if individualism is the norm.

Mobilizes resource utilization

In mediating external interventions, organizational institutions are the *means by which the external resources that facilitate adaptation are delivered*, and they accordingly administer access to such resources. These resources may be information, technical inputs, and/or financial support. Institutions "mediate the extent to which climate change affects communities" (Pradhan et al. 2012: 9); therefore, they are crucial to the successful implementation of externally facilitated adaptation strategies (Agrawal 2008).

Institutions matter because they *determine the cost of transacting* activities (North 1994) and are comprised of "transaction-cost–reducing arrangements" (Kirsten et al. 2009: 43). Transaction costs pertain to the costs incurred from the activities that lessen the risk of transaction failure such as planning, negotiating, creating, monitoring, and enforcement of an agreement. These also include the costs of maladaption, bargaining and other operations related to governance, and securing the commitment of actors to the contracts (Kirsten et al. 2009). Institutions will fail to reduce transaction costs if the institutional context where the transactions take place are in disarray (Theesfeld et al. 2010).

Similarly, a weak institutional environment, specifically in terms of legal frameworks, makes it difficult to enforce contracts and agreements that exchange goods and services, and to coordinate activities (Eaton et al. 2008). The norms and practices existing in the system also affect the cost of transactions. For example, if bribery is the custom, then people may bribe corrupt law enforcers to accomplish their goals. Costs would include resources (e.g., money, time, and people) in bribing transactions plus the regular expenses incurred in undertaking such tasks. If the rule of law is upheld, then bribing will be useless and the associated costs will not exist. Likewise, if cheating is the norm, then there will be additional costs to prevent other parties from cheating. The effectiveness and efficiency of actions and activities depend on the institutional environment and arrangements in place.

Interdependencies and linkages among institutions occur through these functions. These associations are the product of the interactions between and among institutional arrangements (Young 2002). In this sense, functional interdependencies can be defined *as the relationships between institutions resulting from the interactions among arrangements that allow institutions to perform their functions*. These linkages are explored in the IEM framework that analyzes the institutional environment in adaptation responses.

Institutional Environment Matrix Framework

The proposed framework incorporates two layers, the **institutional environment (IE)** and **institutional matrix (IM)** (Figure 4). The institutional environment focuses on a specific type of system, a particular adaptation goal or a type of adaptation strategy. It is a *comprehensive inventory of the differing institutions—rules, social-structures, and organizations—that may influence adaptation responses*. This layer assumes that examining the institutional environment in assessing and planning for climate change adaptation responses is a significant feat. For instance, the coastal management and governance arrangements in East of England showed that there exist:

three central government departments, four regional bodies, five statutory agencies, four ad-hoc groupings, seventeen local authorities, and four forums with an interest in coastal planning, but not necessarily working together.... five sets of overlapping plans, fourteen designations of coastal sites and landscapes, a mix of management bodies, many organizational cultures, un-coordinated organizational activity at different scales, and overlapping jurisdictions, responsibilities and functions (Nicolson-Cole and O'Riordan 2009: 373)

Analyzing or planning for adaptation responses incorporates an exhaustive assessment of the institutional environment in which these responses have been or will be applied. This is important in institutional analysis because the number of institutions in a given system is directly related to the rate and complexity of the institutional linkages (Young 2002).

The institutional matrix (IM) can analyze the interactions among the arrangements. It is defined as the system of institutional arrangements—including operational and collectivechoice rules and governance arrangements—that determines the functional interdependencies of institutions. This layer examines the various relationships among the institutions, and assumes that institutions affect an adaptation response via the institutional arrangements that enable institutions to perform their functions.

The IE and the IM stages are closely linked, such that the IM is dependent on the information provided by the IE. This relationship, however, is one directional. Significant IE analyses can be done even without proceeding to the IM level, but the reverse is not possible.

Various institutional interactions, like complementary, neutral, and counterproductive relationships, are realized from the IM layer. Complementary interaction indicates beneficial associations such that institutions perform better because of the creation or existence of the other (Gunningham and Grabosky 1998). Conversely, counterproductive interactions result when institutional arrangements either destabilize or weaken one another, thus impeding the ability of institutions to perform their functions effectively. Neutral interaction suggests that institutions just simultaneously exist in the institutional environment without interacting with each other. The institutions neither improve nor worsen each other.

Contradicting relationships arise when institutions are mismatched, thereby creating situations in which institutional arrangements are not attuned with each other. This institutional linkage forms tensions and conflicts among institutions and the corresponding elements that function within these institutions (Nicholson-Cole and O'Riordan 2009). Overlapping associations involve disputes in jurisdictions (Davis 2006), especially when institutions have similar mandates (Aggarwal 2005). Institutional overlaps are common and more significant in a high frequency institutional environment where there is a high density of institutional arrangements operating in a single system (Young 2002). Lastly, redundancy signifies complete duplication of all institutional functions (Fig. 3).

All the relationships, except redundancy, may exist in a single or multiple types of institutional functions. Policy 1 may be counterproductive with Policy 2 in establishing systems of power and authority, but may be complementary in creating incentives for individual and collective actions. Likewise, there might not be any connection (neutral) between the two on defining information systems on climate change and adaptation. These linkages can be thoroughly examined using the institutional matrix analysis, which is further explained in the succeeding sections.



Figure 4 Institutional Environment Matrix (IEM) Framework (Source: Authors)

Institutional environment

The IE layer (Table 2) is comprised of formal rules (FR), social structures (SS), formal organization (FO), and informal organization (IO). Formal rules represent the written laws, policies, and regulations, whereas social structures are the traditions, norms, and practices affecting social collective behaviors. This framework incorporates informal rules in the social structure-based institutions, following the notion that they are linked. Formal organizations are the groups legitimized by the formal rules, whereas the informal organizations are those sanctioned by informal rules. The institutions comprising the IE may have been created to address a variety of issues, some of which may not be related to climate change. These rules, social structures, and organizations have particular arrangements that can affect climate change adaptation decisions, hence their inclusion in the IE. Take the case

of Carteret climate change refugees⁶. The adaptation response—community relocation—involves property institutions and property rights arrangements, both of which have been existing and working in the systems long before climate change concerns emerged.

Table 2 Institutional environment frame work fay-out											
Institutional Systems:	Institutional Environment										
Complex System/	Formal	Social	Formal	Informal							
Adaptation Response	Rules (FR)	Structures	Organizations	Organizations (IO)							
		(SS)	(FO)								
	FR1	SS1	FO1	IO1							
	FR2	SS2	FO2	IO2							
	FR3	SS3	FO3								
	FR4										

Table 2 Institutional environment framework lay-out

In this hypothetical case (Table 2), the institutional framework identifies four formal rules (FR1, FR2, FR3, FR4); three social structures (SS1, SS2, SS3); three formal organizations (FO1, FO2, FO3); and two informal organizations (IO1 IO2) that affect the adaptation response. All types of institutions are included in this layer regardless of scale. For example, FR1 may be a national program; FR2, a regional regulation; and FR3 and FR4, local policies. This is possible because the IE layer assumes that the institutions existing in various scales may simultaneously affect (or be affected) by the same adaptation response(s) through their arrangements. These institutional arrangements cut across differing scales, and they structure the relationships and functional interdependencies of institutions. The arrangements associated with each institution are critical in determining the extent of the institution's influence in the decision-making process (Figure 5).

For instance, the Environmental Protection Act 1994 (EP Act)—an act primarily concerned with environmental pollution in the state of Queensland, Australia—can influence local authorities' responsibilities and decisions. Though a state law, the EP Act specifies the responsibilities of local governments in notifying administering authorities of violations at the local level (EP Act, Part 8 [2]).

Thus, the IE layer is composed of all institutions that may affect climate change adaptation, regardless of the scale at which the institution primarily operates. In contrast, the institutional matrix layer is limited to a single scale analysis—only those institutional arrangements that cover the scale (federal/national, state/regional/territory, provincial/local) being analyzed will be examined. In the previous hypothetical case (Table 1), while FR1 is a national program and FR2 is a regional regulation, only those arrangements affecting the local scale will be included in the matrix if the scale of analysis is local. These notions are further elaborated below.

⁶ People were forced to leave their homes and resettle elsewhere because of climate change related events.



Figure 5 Multi-layered institutional linkages (Source: Authors)

Institutional Matrix (IM)

From the IE, the analysis progresses to the individual institutional arrangements in the institutional matrix (IM). The IM is dependent on the set of information provided by the IE stage, and those institutions identified in the environment are incorporated in the matrix (Table 3). In this hypothetical case, the IM analysis focuses on the local scale.

The functions are the source of interdependencies among institutions, which are vital elements of the IM layer. Institutional interplay is observed by analyzing the institutional arrangements that shape these functions. With this, the institutional functions compose the row headings, and they are the categories by which the institutional arrangements are organized in the IM cells.

Framework analyses: vertical and horizontal

The framework offers two types of analyses—vertical and horizontal. Vertical analysis (Table 4) shows the influence of individual institutions on the adaptation response by examining each institution's function. As the hypothetical case has a local scale, only local arrangements will be included in the matrix. The vertical analysis of the formal rules indicates that the national program (FR1) incorporates local arrangements that establish systems of power and authority, identifies inclusions and exclusions, influence and transforms individual acts and expectations into collective actions, coordinates individual or collective behaviors, and defines information systems. The regional regulation (FR2) performs the same tasks in addition to determining transaction costs of activities and decisions. The IM vertical analysis also outlines the dominant institution in the institutional environment. In the example, the local policy FR3 is the most influential among all four formal rules (Table 4).

FUNCTIONS OF AN INSTITUTION		TYPES OF INSTITUTIONS										
		Formal Rules (FR)			Social		Formal Organization s (FO)			Informal		
					Structures					Organizations (IO)		
					(SS)							
	FR1	FR2	FR3	FR4	SS1	SS2	SS3	FO1	FO2	FO3	IO1	IO2
Reduces uncertainty												
Establishes systems of												
power and authority												
Identifies inclusions and												
exclusions												
Creates rights and												
entitlements												
Connects individuals to society	/											
Converts personal values												
into social norms and												
shared beliefs												
Influence and transforms												
individual acts and												
expectations into collective												
actions												ļ
Creates (dis)incentives for												
individual and collective												
actions												ļ
Coordinates individual or												
collective behaviors												L
Fosters adaptive capacity		1	1	1	1	1		1				
Defines information												
systems												ļ
Mediates influence of												
external interventions												
Mobilizes resource utilization												
Means of delivery of												
external resources												ļ
Determines transaction												
costs of activities and												
decisions	1											1

Table 3 Institutional Matrix (IM): An institutional framework for adaptation analysis

Notes: The function "creates rights and entitlements" does not apply to organizations, while the function "means of delivery of external resources" does not relate to formal rules and social structures. The cells are shaded accordingly.

¥	TYPE OF					
FUNCTIONS OF AN INSTITUTION	Formal Rules					
	FR1	FR2	FR3	FR 4		
Reduces uncertainty						
Establishes systems of power and authority	✓	✓	\checkmark	\checkmark		
Identifies inclusions and exclusions	\checkmark	\checkmark	\checkmark			
Creates rights and entitlements			\checkmark			
Connects individuals to society						
Converts personal values into social norms and shared beliefs						
Influence and transforms individual acts and expectations into collective actions	\checkmark	\checkmark	\checkmark	\checkmark		
Creates (dis)incentives for individual and collective actions			\checkmark	\checkmark		
Coordinates individual or collective behaviors	\checkmark	\checkmark	\checkmark	\checkmark		
Fosters adaptive capacity						
Defines information systems	\checkmark	\checkmark	\checkmark	\checkmark		
Mediates external interventions			\checkmark			
Mobilizes resource utilization						
Means of delivery of external resources						
Determines transaction costs of activities and decisions		\checkmark	\checkmark			

 Table 4 Vertical analysis for formal rules

An empty cell signifies that the institution does not perform the associated function at the specific scale in question. However, this does not imply that the institution does not implement the function at all. For example, FR1 may not create rights and entitlements at the local scale, but may have such arrangements in either the national or regional scales. This aspect is the major difference between the IE and IM layers. Although cross-scale investigation is possible in the IE, this cannot be done in the IM. Overall, the vertical analysis shows the extent of the institution's influence on the adaptation response. It also compares its functions across institutions in a specific scale.

The horizontal analysis is more complicated as it studies the functional interdependencies of institutions and assesses the relationships across various institutions based on their functions. In the hypothetical case (Table 5) the institutional linkages of all 12 institutions are illustrated in relation to the function "establishes systems of power and authority". The cells in red are negative relationships, specifically counterproductive and contradicting. Conversely, the green cells are positive associations, particularly the complementary type. Neutral and overlapping relationships are white and yellow, respectively With regard to structuring power and authority systems, some of the possible interpretations of the matrix are as follows:

- 1. Formal rules generally have negative relationships with informal organizations.
- 2. Informal organizations are in harmony with the social structures.
- 3. Formal organizations typically have overlapping jurisdictions with one another.
- 4. Informal organizations typically have overlapping jurisdictions with one another.
- 5. Formal rules are generally counterproductive or contradictory to social structures.
- 6. The national program, FR1, has relationships only with social structures and other formal rules. It does not affect organizations, whether formal or informal.
- 7. The national program, FR1, contradicts with the regional regulation, FR2.

- 8. The national program, FR1, complements the local policy, FR3. As the national program and the regional regulation have a negative relationship, it is consistent that FR2 and FR3 are also contradicting each other. Thus, the regional regulation has a negative linkage with all other formal rules.
- 9. Local policies FR3 and FR4 overlap.
- 10. Both local policies are not attuned with the existing local norms, SS2, but are neutral to the local practices, SS1 and SS3.

Other assessments can be gleaned from this example. This type of analysis can be duplicated to the other functional classifications, thereby creating the overall assessment of the linkages between and among institutions. The matrix, thus, enables a planner or analyst to structurally examine complex institutional relationships, thus, possibly effectively evaluate and plan adaptation responses to climate change.

Establishes Systems of			Formal Rules				Social Structures			Formal Organizations			Informal Organizations	
Power and Authority		FR1	FR2	FR3	FR4	SS1	SS2	SS3	FO1	FO2	FO3	IO1	IO2	
Formal Rules	FR1		Contradicting	Comple- mentary	Neutral	Comple- mentary	Counter- productive	Counter- productive	Neutral	Neutral	Neutral	Neutral	Neutral	
	FR2	Contradicting		Counter- productive	Contradicting	Comple- mentary	Contradicting	Neutral	Comple- mentary	Contradicting	Comple- mentary	Counter- productive	Counter- productive	
	FR3	Comple- mentary	Counter- productive		Overlapping	Neutral	Counter- productive	Neutral	Comple- mentary	Comple- mentary	Comple- mentary	Counter- productive	Contradicting	
	FR4	Neutral	Contradicting	Overlapping		Neutral	Contradicting	Neutral	Contradicting	Contradicting	Neutral	Contradicting	Contradicting	
Social Structures	SS1	Comple- mentary	Comple- mentary	Neutral	Neutral		Comple- mentary	Comple- mentary	Comple- mentary	Neutral	Contradicting	Comple- mentary	Comple- mentary	
	SS2	Counter- productive	Contradicting	Counter- productive	Contradicting	Comple- mentary		Comple- mentary	Contradicting	Contradicting	Contradicting	Comple- mentary	Comple- mentary	
	SS3	Counter- productive	Neutral	Neutral	Neutral	Comple- mentary	Comple- mentary		Neutral	Neutral	Neutral	Comple- mentary	Comple- mentary	
Formal Organizations	FO1	Neutral	Comple- mentary	Comple- mentary	Contradicting	Comple- mentary	Contradicting	Neutral		Overlapping	Overlapping	Counter- productive	Comple- mentary	
	FO2	Neutral	Contradicting	Comple- mentary	Contradicting	Neutral	Contradicting	Neutral	Overlapping		Overlapping	Comple- mentary	Neutral	
	FO3	Neutral	Comple- mentary	Comple- mentary	Neutral	Contradicting	Contradicting	Neutral	Overlapping	Overlapping		Counter- productive	Counter- productive	
Informal Organizations	IO1	Neutral	Counter- productive	Counter- productive	Contradicting	Comple- mentary	Comple- mentary	Comple- mentary	Counter- productive	Comple- mentary	Counter- productive		Overlapping	
	IO2	Neutral	Counter- productive	Contradicting	Contradicting	Comple- mentary	Comple- mentary	Comple- mentary	Comple- mentary	Neutral	Counter- productive	Overlapping		

Table 5 Horizontal analysis for structure power and authority systems function

Conclusion

Institutions in climate change encompass rules, social structures, and organizations. The institutional dimension of climate change adaptation involves an intricate web of relationships between and among these institutions. In analyzing the complexity of institutions, a number of factors need to be considered such as institutional functions and interplay, as well as issues of scale (national, state, regional and local) and jurisdiction. Thus, a purpose built framework that can perform these kinds of analysis, such as the Institutional Environment Matrix (IEM) framework, is needed.

The IEM adopts a dual-layered approach in examining the various institutions that directly or indirectly influence adaptation decisions and responses in a particular system. Institutions are intrinsically complex, hence a single layer analysis cannot cover the intricacies involved in an institutional analysis. Furthermore, this design allows institutions to be examined across scales and provides an easy transition toward a single scale analysis.

The dual layer design of the IEM allows the institutional environment and arrangements to be extensively studied. The IE layer includes all kinds of institutions in the environment regardless of scale, identifies the dominant institutions in the system affecting adaptation responses, and outlines the complexity of institutions in the institutional environment. Meanwhile, the IM layer enables a scale-focused analysis by dealing with particular institutional arrangements. The matrix allows for complex analysis of institutional linkages and interactions through the vertical and horizontal analytical approaches. By using these techniques, the functional interdependencies of institutions can be identified and institutional interplay can be explored.

The institutional dimension of climate change adaptation is motivated by the need to design or re-design arrangements to address the risks and impacts of climate change (Young 2002). Accordingly, the IEM framework helps identify whether the existing institutions hinder effective adaptation, especially when there are negative relationships among the institutional arrangements. This condition may warrant modifying or replacing the arrangements such that institutions will fit more effectively in the institutional environment. When new institutions need to be introduced into the system, the IEM framework may be useful in developing arrangements that will be compatible with the existing ones. This will help avoid mismatches among institutions, and thus minimize conflicts.

This Chapter has outlined a theoretical tool that can be used in adaptation planning and evaluation. However, the real value of the framework lies in its applicability in empirical cases. The need for further research in this area is vital.